20030121007

Best Available Copy



,¥**		7	۴	44		*	5		

	READ INSTRUCTIONS
1. REPOPT NUMBER	ACCESSION NO. 3. AECIPIENT'S CATALOG HUMBER
4. TITLE (and Subtilie) Archaeological Investigations at Site 4 Chief Joseph Dam Project, Washington	5-D0-282, Aug 1978Oct 1984 6. PERFORMING ORG. REPORT NUMBER
7. AUTHOR(*) E.S. Lohse with S.K. Campbell, S.N. Croz L. Hause, S. Livingston, and D. Sammons.	E. CONTRACT OR GRANT NUMBER(*) zier, DACW67-78-C-0106 -Lohse
9. PERFORMING ORGANIZATION NAME AND ADDRESS Office of Public Archaeology.	10. PROGRAM ELEMENT, PROJECT, TASK AREA & WORK UNIT HUMBERS
Institute for Environmental Studies University of Washington, Seattle WA 98	BF285 18 08E U 0000
1. CONTROLLING OFFICE NAME AND ADDRESS	12. REPORT DATE
Seattle District, Corps of Engineers	13. NUMBER OF PAGES
P.O. Box C-3755, Seattle, WA 98124 14. MONITORING AGENCY NAME & ADDRESS(If different from Cont	troiling Office) 15. SECURITY CLASS. (of this report)
•	UNCLASSIFIED
is. DISTRIBUTION STATEMENT (of this Report) Approved for fullic re	lease, distribution unlimited
16. DISTRIBUTION STATEMENT (of this Report) Approved for pulling re 17. DISTRIBUTION STATT'SENT (of the obstract entered in Black 20	1ease, distribution unlimited
Approved for fullic re Approved for fullic re 7. Distribution Stattigent (of the obstract entered in Black 20 18. Supplementary NOTES	15. DECLASSIFICATION/DOWNGRADING SCHEDULE lease, distribution unlimited
Approved for pullic re Approved for pullic re 7. DISTRIBUTION STATT Col the obstract entered in Black 20 8. SUPPLEMENTARY NOTES 9. KEY WORDS (Continue on reverse side if necessary and identify to Cultural Resources, Washington	<pre> 15. DECLASSIFICATION/DOWNGRADING SCHEDULE 1ease, distribution unlimited 0, 11 different from Report) by block number) Microblades </pre>
Approved for public re Approved for public re 7. DISTRIBUTION STATT-GENT (of the obstract entered in Black 20 9. KEY WORDS (Continue on reverse side if necessary and identify to Cultural Resources, Washington Columbia River Frehistory Chief Joseph Dam Project Archaeology Settlement and Subsistence Patterns	by black number) Microblades Cascade Phase Frenchman Springs Phase Southern Okanogan Indians
Approved for public re Approved for public re 7. DISTRIBUTION STATT-GENT (of the obstract entered in Black 20 9. KEY WORDS (Continue on reverse side if necessary and identify to Cultural Resources, Washington Columbia River Frehistory Chief Joseph Dam Project Archaeology Settlement and Subsistence Patterns 9. ABSTRACT (Continue on reverse side if measures and identify by	<pre>ise. DECLASSIFICATION/DOWNGRADING SCHEDULE lease, distribution unlimited o, if different from Report; by block number; Microblades Cascade Phase Frenchman Springs Phase Southern Okanogan Indians y block number;</pre>

auto an turn and an and and and and

- .

SECURITY CLASSIFICATION OF THIS PAGE/MAR Unie Aniard,

BLOCK 20 (Continued)

Site 45-D0-282 is on the south bank of the Columbia River (River Mile 556) near the Okanogan Highland - Columbia Plateau boundary in an Upper Sonoran life zone. The University of Washington excavated 186.1 m³ of site volume in 1979 for the U.S. Army Corps of Engineers, Seattle District, as part of a mitigation program associated with adding 10 ft to the pool level behind Chief Joseph Dam. Systematic aligned random sampling with 1 x 1 x 0.2-m units of record in 1 x 2 or 2 x 2-m cells disclosed one historic and four prehistoric occupations on an alluvial fan built onto an early river terrace, interbedded with later overbank and aeolian sediments. There are no radiocarbon dates, but projectile points indicate the earliest occupation is early to mid-Kaitar Phase. The second, more intensive occupation probably occurred 6,000 to 5,000 years ago. The third and fourth occupations in the late Kartar Phase took place about 5,000 to 4,000 years ago. Sccupation character shows no change in 2,500 years; all occupations are lithic scatters with blade and microblade technology and chipping stations. Shelters, earth ovens, hearths, and bone concentrations are absent. Environmental stability is indicated by soil formation after 4,000 years ago. The historic occupation is an early 20th century homestead.

and the second second

ARCHAEOLOGICAL INVESTIGATIONS AT SITE 45-DO-282, CHIEF JOSEPH DAM PROJECT, WASHINGTON

by

Ernest S. Lohse

with

Sarah K. Campbell, S. Neal Crozier Larry Hause, Stephanie Livingston, and Dorothy Sammons-Lohse

Principal Investigators

R.C. Dunnell 1978-1984 D.K. Grayson 1978-1981 M.E.W. Jaehnig 1981-1984 J.V.Jermann 1978-1981

Final report submitted to the U.S. Army Corps of Engineers, Seattle District, in partial fulfillment of the conditions and specifications of Contract No. DACW67-78-C-0106.

The technical findings and conclusions in this report do not necessarily reflect the views or concurrence of the sponsoring agancy.

> Office of Public Archaeology Institute for Environmental Studies University of Washington

ABSTRACT

Site 45-D0-282 is on the south bank of the Columbia River (River Mile 556) near the Okanogan Highland - Columbia Plateau boundary in an Upper Sonoran life zone. The University of Washington excavated 186.1 m³ of site volume in 1979 for the U.S. Army Corps of Engineers, Seattle District, as part of a mitigation program associated with adding 10 ft to the pool level behind Chief Joseph Dam. Systematic cligned random sampling with $1 \times 1 \times 0.2$ -m units of record in 1×2 or 2×2 -m cells disclosed one historic and four prehistoric occupations on an alluvial fan built onto an early river terrace, interbedded with later overbank and aeolian sediments. There are no radiocarbon dates, but projectile points indicate the earliest occupation is early to mid-Kartar Phase. The second, more intensive occupation probably occurred 6,000 to 5,000 years ago. The third and fourth occupations in the late Kartar Phase took place about 5,000 to 4,000 years ago. Occupation character shows no change in 2,500 years; all occupations are lithic scatters with blade and microblade technology and chipping stations. Shelters, earth ovens, hearths, and bone concentrations are absent. Environmental stability is indicated by soil formation after 4,000 years ago. The historic occupation is an early 20th century homestead,

1		
Acce	sion For	1
NTIS DTIC	CRA&I TAB	vi -
Justif	ication	
By Ditit	Nition 1	
A	vailability C	odes
Dist	Avail and Special	or
A-1		

iii

TABLE OF CONTENTS

ABSTRACT
TABLE OF CONTENTS
LIST OF FIGURES
LIST OF TABLES ix
LIST OF PLATES
PREFACE
ACKNOWLEDGEMENTS
1. INTRODUCTION
INVESTIGATIONS AT 45-D0-282
2. NATURAL AND CULTURAL STRATIGRAPHY
GEOLOGIC SETTING 11 PROCEDURES 13 DEPOSITIONAL HISTORY 13 ANALYTIC ZONES 20 ZONE 4 20 ZONE 3 20 ZONE 2 20 ZONE 1 22 THE BEACH COLLECTION (ZONE 25) 22
GEOLOGIC SETTING 11 PROCEDURES 13 DEPOSITIONAL HISTORY 13 ANALYTIC ZONES 20 ZONE 4 20 ZONE 3 20 ZONE 2 20 ZONE 1 22 THE BEACH COLLECTION (ZONE 25) 23 Ernest S. Lohse 23
GEOLOGIC SETTING 11 PROCEDURES 13 DEPOSITIONAL HISTORY 13 ANALYTIC ZONES 20 ZONE 4 20 ZONE 3 20 ZONE 2 20 ZONE 1 22 ZONE 1 22 THE BEACH COLLECTION (ZONE 25) 22 3. ARTIFACT ANALYSIS 23 Ernest S. Lohse 27 OBJECT TYPES 27 MANUF ACTURE 30 INDUSTRIES 39 TEMPORAL AND SPATIAL DISTRIBUTION 51 FUNCTIONAL ANALYSIS 52

STYLI	UGGESTED USE	71 73 75 76 76 82
4. FAUNAL ANAL	'SIS	93
FAUNA SPECI DISCU	ASSEMBLAGE	93 93 97
5. FEATURES AN	ILYSIS	99
ZONE ZONE ZONE CONCL	JSIONS	99 105 105 107
6. SYNTHESIS.	rnest S. Lohse	109
ZONE ZONE ZONE ZONE THE B DISCU	ACH COLLECTION (ZONE 25)	110 111 111 111 111 111 112
REFERENCES	• • • • • • • • • • • • • • • • • • • •	117
APPENDIX A: Re	ults of Scil Analyses	125
APPENDIX B: Ar	ifact Assemblage	131
APPENDIX C: Fa	ral Assemplage	1 <i>3</i> 9 143

vi

·

NTREASERS AND AND A CONTREMENTATION AND A CONTREASERS AND A CONTREASE AND A CONTREASE AND A CONTREASERS AND A CONTREASER

LIST OF FIGURES

Figure 1-1.	Map of project area showing location of 45-D0-282 2
Figure 1-2.	Map of site vicinity
Figure 1-3.	Topographic map showing location of excavation areas 6
Figure 1-4.	Units excavated and surface collected, Area A 8
Figure 1-5.	Units excavated, Areas B and C 9
Figure 2-1.	Geologic map of site vicinity
Figure 2-2.	Location of profiled walls in Area A
Figure 2-3.	Location of profiled walls in Areas B and C 15
Figure 2-4.	Profile of 83N200E, Area A
Figure 2-5.	Profile of 16N6W, Area B
Figure 3-1.	Microblade core and microblade terms
Figure 3-2.	Length-width relationship of microblades
Figure 3-3.	Type and location of wear by functional object type 66
Figure 3-5.	Morphological projectile point types 74
Figure 3-5.	Morphological classification of projectile points 78
Figure 3-6.	Definition of projecti's point outline
Figure 3-7.	Historical projectile point type classification 83
Figure 3-8.	Projectile point type distribution across cultural phases
Figure 3-9.	Projectile point type distribution within cultural phases
Figure 5-1.	Location of features in Area A
Figure 5-2.	Location of features in Areas B and C
Flaure B-1.	Digitized projectile point outlines

vii

ļ

1

LIST OF TABLES

3

たいというときがある

Ś

Table 2-1	. Summary of depositional units ,
Tatle 2-2	The analytic zones: their stratigraphic definition and contents
Table 3-1	. Technological dimensions
Table 3-2	. Count of ilthic material by zone
Tatle 3-3	. Count of non-lithic material by zone
Table 3-4	. Material by object type by zone
Table 3-5	. Type of manufacture by zone
Table 3-6	. Count of treatment
Table 3-7	. Heat treatment by material by zone
Table 3-8	. Count of dorsal topography
Table 3-9	. Amount of cortex by material by zone
Table 3-1	0. Cortex on cryptocrystalline and other material by object type and zone
Table 3-1	1. Count of flake size by material and zone
Table 3-1	2. Average weight of conchoidally flaked material by zone
Table 3-1	3. Average length of conchoidally flaked material by zone
Table 3-1	 Average width of conchoidally flaked material by zone 38
Table 3-1	5. Average thickness of conchoidally flaked material by zone
Table 3-1	6. Material by object type, functional type, dorsal topography, and zone
Table 3-1	7. Attributes of microblade cores 49
	ix
6 	LANGER FRANKSKINGER STOLEN STOLEN STOLEN. -

Table 3-18.	Microblade measurements
Table 3-19.	Microblade attributes by zone
Table 3-20.	Functional dimensions
Table 3-21.	Utilization/modification and type of manufacture of formed lithic objects by zone
Table 3-22.	Functional object type and wear area/object ratio by zone
Table 3-23.	Type and location of wear by zone 67
Table 3-24.	Ranking of functional tool types by wear type 68
Table 3-25.	Distribution of functional types by zone
Table 3-26.	Dimensions of morphological projectile pcint classification
Table 3-27.	Line segment definitions of morphological projectile point types
Table 3 - 28.	Classification of projectile points and projectile point fragments
Table 4-1.	Taxonomic composition and distribution of vertebrate remains
Table 5-1.	Provenience of features
Table 5-2.	Provenience of artifacts by feature
Table 5-3.	Lithic artifacts and debitage by material types 104
Table 5-4.	Dorsal typography of lithic materials by feature 106
Table A-1.	Results of physical and chemical soil analysis, Column 1 . 125
Table A-2.	Results of physical and chemical soil analysis, Column 2 . 127
Table A-3.	Results of physical and chemical soil analysis, Column 3 . 128
Table A-4.	Results of physical and chemical soli analysis, Column 4 . 128
Table A-5.	Results of physical and chemical soil analysis, Column 5 . 129
Table 8-1.	Object edge angle by artifact type and zone
Table B-2.	Distribution of tool types by zone

というないであるというでは、そのからいのではないいったのではないできたが、そのことでもないです。そのできたです。そのできたが、そのできたので、これできたとうないないないです。

4

x

.

LIST OF PLATES

.

Plate 1-1.	Downstream view showing area of beach collection 1	
Plate 3-1.	Microblade cores	
Plate 3-2.	Microblades	
Plate 3-3.	Flake cores and bifaces	
Plate 3-4.	Projectile points	
Plate 3-5.	Drills, burins, biades, and gravers 60	
Plate 3-6.	Large cobble tools	
Plate 3-7.	Pestles	
Plate 5-1.	Feature 4, Zone 4, Area A	
Plate 5-2.	Feature 5, Zone 4, Area A	
Plate 5-3.	Feature 7, Zone 3, Area B	

хi

The Chief Joseph Dam Cultural Resources Project (CJDCRP) has been sponsored by the Seattle District, U.S. Army Corps of Engineers (the Corps) in order to salvage and preserve cultural resources imperiled by a 10-foot pool raise resulting from modifications to Chief Joseph Dam.

From Fall 1977 to Summer 1978, under contract to the Corps, the University of Washington, Office of Public Archaeology (OPA) undertook detailed reconnaissance and testing along the banks of Rufus Woods lake in the Chief Joseph Dam project area (Contract No. DACW67-77-C-0099). The project area extends from Chief Joseph Dam at Columbia River Mile (RM) 545 upstream to RM 590, above seven miles below Grand Coulee Dam and includes 2,015 hectares (4,979 acres) of land within the guide-taking lines for the expected pool raise. Twenty nine cultural resource sites were identified during reconnaissance, bringing the total number of recorded prehistoric sites in the area to 279. Test excavations at 79 of these provided information about prehistoric cultural variability in this region upon which to base further resource management recommendations (Jermann et al. 1978; Leeds et al. 1981).

Only a short time was available for testing and mitigation before the planned pool raise. Therefore, in mid-December 1977, the Corps asked the OPA to review the 27 sites tested to date and identify those worthy of immediate investigation. A priority list of six sites was complied. The Corps, in consultation with the Washington State Historic Preservation Officer and the Advisory Council on Historic Preservation, established an interim Memorandum of Agreement under which full-scale excavations at those six sites. could proceed. In August 1978, data recovery (Contract No. DACW67-78-C-0106) began at five of the six sites.

٠.

Concurrently, data from the 1977 and 1978 testing, as well as those from previous testing efforts (Osborne et al. 1952; Lyman 1976), were synthesized into a management plan recommending ways to minimize loss of significant resources. This document calls for excavations at 34 prehistoric habitation sites, including the six already selected (Jermann et al. 1978). The final Memorandum of Agreement includes 20 of these. Data recovery began in May 1979 and continued until late August 1980.

Full-scale excavation could be undertaken at only a limited number of sites. The testing program data allowed identification of sites in good condition that were directly threatened with inundation or severe erosion by the projected pool raise. To aid in selecting a representative sample of prehistoric habitation sites for excavation, site "components" defined during testing were characterized according to (1) probable age, (2) probable type of occupation, (3) general site topography, and (4) geographic location along the

PREFACE

river (Jermann et al. 1978:Table 18). Sites were selected to attain as wide a diversity as possible while keeping the total number of sites as low as possible.

The Project's investigations are documented in four report series. Reports describing archaeological reconnaissance and testing include (1) a management plan for cultural resources in the project area (Jermann et al. 1978), (2) a report of testing at 79 prehistoric habitation sites (Leeds et al. 1981), and (3) an inventory of data derived from testing. Series I of the mitigation reports includes (1) the project's research design (Campbell 1984d) and (2) a preliminary report (Jaehnig 1983b). Series II consists of 14 descriptive reports on prehistoric habitation sites excavated as part of the project (Campbell 1984b; Jaehnig 1983a, 1984a,b; Lohse 1984a-f; Miss 1984a-d), reports on prehistoric nonhabitation sites (Campbell 1984a) and burial relocation (Campbell 1984c), and a report on the survey and excavation of historic sites (Thomas et al. 1964). A summary of results is presented in Jaehnig and Campbell (1984).

This report is one of the Series II mitigation reports. Mitigation reports document the assumptions and contingencies under which data were collected, describe data collection and analysis, and organize and summarize data in a form useful to the widest possible archaeological audience.

ACKNOWLEDGEMENTS

This report is the result of the collaboration of many individuals and agencies. During the excavation and early reporting stages, Co-principal investigators were Drs. Robert C. Dunnell and Donald K. Grayson, both of the Department of Anthropology, University of Washington, and Dr. Jerry V. Jermann, Director of the Office of Public Archaeology, University of Washington. Dr. Manfred E.W. Jaehnig served as Project Supervisor during this stage of the work. Since the autumn of 1981, Dr. Jaehnig has served as Co-principal investigator with Dr. Dunnell.

Three Corps of Engineers staff members have made major contributions to the project. They are Dr. Steven F. Dice, Contracting Officer's Representative, and Corps archaeologists Lawr V. Salo and David A. Munsell. Both Mr. Munsell and Mr. Salo have worked to assure the success of the project from its initial organization through site selection, sampling, analysis, and report writing. Mr. Munsell provided guidance in the initial stages of the project and developed the strong ties with the Colville Confederated Tribes essential for the undertaking. Mr. Salo gave generously of his time to guide the project through data collection and analysis. In his review of each report, he exercises that rare skill, an ability to criticize constructively.

We have been fortunate in having the generous support and cooperation of the Colville Confederated Tribes throughout the entire length of project. The Tribes' Business Council and its History and Archaeology Office have been invaluable. We owe special thanks to Andy Joseph, former representative from the Nespelem District on the Business Council, and to Adeline Fredin, Tribal Historian and Director of the History and Archaeology Office. Mr. Joseph and the Business Council, and Mrs. Fredin, who acted as ilaison between the Tribe and the project, did much to convince appropriate federal and state agencies of the necessity of the investigation. They helped secure land and services for the project's field facilities as well as helping establish a program which trained local people (including many tribal members) as field excavators and laboratory technicians. Beyond this, their hospitality has made our stay in the project area a most pleasant one. In return, conscious of how much gratitude we wish to convey in a few brief words, we extend our sincere thanks to all the members of the Colville Confederated Tribes who have supported our efforts, and to Mrs. Fredin and Mr. Joseph, in particular.

Excavations at 45-D0-282 were carried out by a joint Western Washington University and University of Washington field school, under the direction of Garland Grabert. Site 45-D0-282 is located on land owned by Howard F. Brandt of Bridgeport, whom we thank for granting us permission to excavate. As authors of this report, we take responsibility for its contents. What we have written here is only the final stage of a collaborative process which is analogous to the integrated community of people whose physical traces we have studied. Some, by dint of hard labor and archaeological training, salvaged those traces from the earth; others processed and analyzed those traces; some manipulated the data and some wrote, edited and produced this report. Each is a member of the community essential to the life of the work we have done.

A CONTRACTOR OF THE OWNER OF THE

Jerry V. Jermann, Co-principal investigator during the field excavation and artifact analysis phase of the project, developed site excavation sampling designs that were used to select data from each site. The designs provided a uniform context for studying prehistoric subsistence-settlement patterns in the project area.

S. Neal Crozier did the initial data summary for the stratigraphic analysis; he also performed the chemical and mechanical sort analyses. The laboratory staff did the technological and functional artifact analysis. Janice Jaehnig did keypunching and John Chapman and Duncan Mitchell manipulated the computerized data.

The writing of the report itself is a cooperative effort. Ernest S. Lohse wrote Chapters 1, 3, and 6. As senior author, he also coordinated and integrated the contributions of the other authors. S. Neal Crozier, Sarah K. Campbell and Larry Hause wrote Chapter 2. Stephanie Livingston analyzed the faunal assemblage and wrote Chapter 4. Dorothy Sammons-Lohse analyzed the cultural features and wrote Chapter 5.

Marc Hudson edited the text; Dawn Brislawn typed it, and coordinated production. Melodie Tune and Bob Radek drafted the final versions of the graphics except Figure 3-2, drafted by the senior author, and Figures 3-3 and 3-4, drafted by Bob Thomas. Larry Bullis photographed the artifacts and took the cover photograph. Production of the final camera-ready copy was accomplished by Natalle Cadoret, Charlotte Beck, and Karen Weed under the direction of Sarah Campbell.

xvi

1. INTRODUCTION

Site 45-D0-282 is on the left bank of the Columbia River about 125 m downstream from River Mile (RM) 556 in the NW1/4 of the NW1/4 of Section 29, T30N, R27E, Boot Mtn. Quadrangle (U.T.M. Zone 11, N. 5327650, E. 315900). Figure 1-1 shows the location of site 45-D0-282 in relation to the other salvaged sites in the Rufus Woods Lake project area. Lying within Box Canyon, the site is situated on the south side of a bend in the river, about 300 m downstream from an unnamed series of rapids and about .5 km upstream from Eagle Rapids. It lies on a broad, sloping terrace at an elevation of 289-295 m (947-968 ft) m.s.l. and about 40-50 m above the original level of the river (calculated from annual low water level, 1931). The river forms the northern site boundary, marked by a wide, shallow beach studded with large basalt erratics (Plate 1-1). On the south, the site is bounded by a steep talus slope which rises to another higher terrace.



Place 1-1. Downstream view showing area of beach collection, 45-D0-282. (View to west.)



Figure 1-1. Map of project area showing location of 45-00-282.

2

The site terrace rises gradually to the south, away from the river, with a gain in elevation across the site area of zbout 9 m (Figure 1-2). Two ephemeral stream channels, running roughly north-south, cut across the site area. An unimproved dirt road crosses the site from east to west, parallelling the river margin. An historic homestead was built in the approximate middle of the site area, with buildings and other structures concentrated near the westernmost stream channel. Apple trees are widely scattered over the site near channels of natural runoff. The entire terrace has been subject to numerous channel cutting episodes, during which heavy rain or snowmelt has sluiced down from the higher terrace into the river below. The attendant erosion, and deposition of sediments, coupled with historic Euroamerican activities, have greatly modified the character of cultural remains. A plowzone, about 20-30 cm in depth, was evident in excavation units. Deeper site deposits showed little or no clustering of artifacts or recognizable coundaries defining cultural features, suggesting that runoff has moved the cultural materials down slope toward the river. Thus, most of the site's artifact associations are probable products of secondary depositions.

A sagebrush-grass association (<u>Artamisia tridentata-Agropyron</u>) (Daubenmire 1970), which is typical of the Jpper Sonoran life zone (Piper 1906), dominates the vegetation in the site area. Scattered sagebrush and rabbitbrush (<u>Chrysothamnus nausecsus</u>), spring flowers, and a dense understory of grasses grow on the site (cf. Franklin and Dyrness 1973). Introduced elements include cheatgrass (<u>Bromus tectorum</u>), and thisties (<u>Saisola kali</u> and <u>Cirsium sp.</u>), among others. A more mesic association, including rose (<u>Rosa</u> sp.), serviceberry (<u>Amelanchier sp.</u>), horsetail (<u>Equisetum spp.</u>), rushes (<u>Equisetum hymale</u>), tule (<u>Scirpus acutus</u>), and sedges (<u>Carex spp.</u>), is found in nearby drainages.

On the upper terraces above the river, <u>Artemisia rigida</u> replaces big sagebrush in areas of thinner, rocky soils. Bitterbrush (<u>Purshia tridentata</u>) and isolated pines (<u>Pinus ponderosa</u>), with an understory of grasses, grow along the steep draws draining the slopes and terraces. To the south, scattered pines give way to sagebrush covered uplands dotted with small lakes and springs. To the north, across the river, mixed Douglas fir (<u>Pseudotsuga menziesii</u>) and pine are dominant in moister bottomiands and along streams, where they grow with broadleaf trees and shrubs. At the highest elevations, the fir forest gives way to pine forest, except on north-facing slopes and valley floors, where the dominant species is still Douglas fir with larch (<u>Larix occidentalis</u>), some spruce (<u>Picea engelmannii</u>) and an associated understory of woody shrubs.

All of these environments, from river bottom to mountain zones, may be found within an eight km radius centered on site 45-D0-282. The Tumwater and Achimin Basins across the river to the north represent moderately well-drained zones with a variety of associated plant communities on surrounding slopes and along stream channels. Small patches of pine forest grow along Tumwater Creek and atop highland areas to the north of the basin. On the south side of the river, numerous small lakes occur in the uplands. Several of these are drained by stream channels which empty into the Columbia and provide natural



Figure 1-2. Map of site vicinity, 45-D0-282.

ሚሆንት ጊዜ የጋር የጋር ጊዜ ጊዜ ጊዜ ጊዜ ጊዜ ጊዜ ጊዜ ጊዜ ግን ብር ብር እና አግሪ ላይ በኒክስር ነዋሪ እና አግሪ ትር እና አግሪ ነዋሪ እና አግሪ ነዋሪ እና አግሪ ነዋሪ

4

いうう 高齢的ない いいじょう とうかせい たん イイブ 御護者 ビング・シック いたけけ ど さんしょく オーキン

routes down to the river bottom. Goose Lake and the Omak Trench lie less than 20 km to the northeast, offering a variety of microenvironments, including riparian habitats along the lake shore. On both sides of the Omak Trench are high uplands with pine and fir forest. To the north is Omak Lake, and to the east, Whitmore Mountain, which rises to an elevation of over 1,050 m m.s.l.

A variety of large and small game was present. Deer and elk ranged widely between upland and river bottom vegetation zones. Smaller species had more restricted ranges tied to specific vegetation communities or water sources. Migratory waterfowl were plentiful along lake, river, and stream margins during the spring and fail migrations. The river offered salmon, suckers, and freshwater mussels, with salmon available during spring and fail runs.

The location of 45-D0-282 on the river, near dependable sources of fresh water, with easy access to upland environments, and near the historically recorded fishing site of Kalitsin, would have allowed systematic exploitation of a number of these floral and faunal resource zones. The occupants may have been drawn there by the seasonal abundance of salmon, or perhaps the pleasant aspect of the site--the river with its sandy beach, the nearby springs, the readily accessible riverine resources. The nearby rapids could certainly have been a major focus of activity during the salmon runs in the spring and fall. And the shoreline with its springs and streams would have attracted wildlife, particularly migratory waterfowl in the spring and fall and big game in the winter months.

たいいい

INVESTIGATIONS AT 45-DO-282

Site 45-D0-282 was excavated during the summer 1979 field season, and was the subject of a combined University of Washington and Western Washington University field school under the direction of Dr. Garland F. Grabert. It was selected for excavation because survey and testing had disclosed a large scatter of debris over 90,000 m² in extent and two meters in depth. Further, the general site area had included numerous springs, ephemeral streams, and large rapids, factors that would seem to have been ideal for prehistoric occupation and exploitation of a variety of permanent and seasonal resources. Testing confirmed the potential of 45-D0-282: three separate cultural components, spanning a period as great as 7,000 years, were identified. The site was thus considered significant because of its great size, its physical setting favorable for occupation, and its indicated time depth.

The designation 45-D0-282 actually encompasses three prior site numbers: 45-D0-188, 45-D0-282, and 45-D0-187H. 45-D0-282 was chosen to designate the site because testing began there and was continued upstream until general site boundaries were established. Three separate areas were designated during excavation (Figure 1-3). They were defined on the basis of relative concentrations of cultural material, and do not correspond to prior site identifications. Area B includes the major portion of the site. Area A, encompassing site 45-D0-188, was near the beach, immediately upstream from historic site 45-D0-187H. Area C was inland, near the boulder terrace





Protocology Internetion Internetion

bordering the site on the south.

Excavation began 19 June 1979, and continued until 17 August 1979. A systematic sampling design was employed in all three areas. Eighteen sampling units were placed in Area A (Figure 1-4), 17 units in Area B, and five in Area C (Figure 1-5). A surface collection was also made on the beach, an intermittently inundated area just north of Area A (Figure 1-4). All 1 \times 1-m units within the grid coordinates defining this collection were examined. Collection was confined to artifacts found on the surface. No excavation of soil matrix or screening was done.

7

Investigation was done by 26 field school students under the direction of three teaching assistants and the instructor. Excavation was done within either 1 x 2-m or 2 x 2-m grid squares. Matrix was removed in arbitrary 10 cm levels keyed to unit and site datums and screened through i/8 in wire mesh. Stratigraphic profiles were created for units whenever these were deemed appropriate. A transit and alidade were used to record unit excavation levels and for mapping the general site area.

The sheer size of 45-D0-282 (over 40 hectares) dictated the approach to excavation. The paramount sampling concern was coverage of the defined area. The number of sampling units was limited to that which would be feasible to dig, given time requirements. Percentage of site area sampled was not fixed, since anything but a cursory examination was impossible. The strategy was to cover the area adequately, with the assumption that a systematic placement of units would reveal any significant patterning in the artifact assemblage. Another assumption made was that patterning in the archaeological record was not consistent, i.e. coincident with the systematic placement of sampling units, and as a corollary, that cultural features would be of sufficient size to be penetrated by 1 x 2-m or 2 x 2-m grid units laid out with a gap of not more than eight meters between them. These assumptions are even more crucial given the great size of the site, which precluded excavating outside of the original sample units. At best, excavation of 45-D0-282 was expected to produce a good characterization of the nature of cultural deposits, identifying variation in artifact assemblages and site function over time and space. It was meant to yield a good sampling of vertical deposits in the site and any difterences in the structure of those deposits across the site area.

A total of 126 1 x 1-m excavation units were dug and another 1,120 1 x 1m units were surface collected. Excavation units removed 183 m³ of cultural deposit, and contained 15,170 stone artifacts, eight fire-modified rocks, 201 bone fragments, and seven pieces of shell. Surface collected units produced 1,394 stone artifacts, 90 fire-modified rocks, 32 bone fragments, and no pieces of shell. Altogether 921 tools were identified, and 13,662 pieces of debitage were sorted.

REPORT FORMAT

The following chapters survey the data recovered from 45-D0-282. Chapter 2 discusses the site's sedimentary stratigraphy and the assignment of artifacts to defined analytic zones. Chapter 3 is an analysis of the





an tanàna mandritra dia kaominina minina Ny fisiana minina mi



artifacts, characterizing technological, functional and stylistic aspects of the assemblage. Chapter 4 describes the results of faunal analysis. Chapter 5 describes cultural features identified at the site. Chapter 6 summarizes our findings and discusses both the chronology of site occupations and the nature of activities documented for site inhabitants. No betanical analysis was done for 45-D0-282, although soil samples are available for future analysis.

2. NATURAL AND CULTURAL STRATIGRAPHY

This chapter discusses the geologic setting of site 45-D0-282 with reference to local geologic history and describes the sedimentary history of the site itself in detail. Strata mapped during excavation are grouped into sitewide depositional units, which provide the basis for determining how deposition occurred and for correlating cultural materials among units. The second half of the chapter discusses the cultural strata or analytic zones defined within this framework.

GEOLOGIC SETTING

Geologic formations in the vicinity of 45-D0-282 are shown in Figure 2-1. On either side of 45-D0-282 are outcrops of granite bedrock (Mzg). These are part of the exterior Colville batholith, the main body of which lies to the north. Not shown on the map is the later Miocene basalt formation. Part of the vast basalt flows forming the entire Columbia Plateau, this formation locally covers 250 sq km on the plateau north of the Columbia River Canyon. It outcrops also on the southern rim of the canyon, indicating a greater original extent. The basalt lag blocks on the site area, if indeed deposited by advancing Cordilleran ice, may be derived either from this local basalt formation, or from a source further away. Alternatively, the large pleces of basalt may not be erratics, but simply remnants of the local basalt formation lowere uy gravity as the deposit was eroded from the canyon. This mechanism, hypothesized for basalt "havstacks" on upper terraces in the northern end of the reservoir (Hibbert 1984), probably is responsible for the basalt "erractics" at 45-D0-282.

Pleistocone events in the site vicinity include the deposition of proglacial gravels (Qpg) and the early Nespelem silt (Qne), followed by deposition of the till (Qt) as the ice advanced southward across the canyon and over the southern rim to the Waterville Plateau. On the north side of the canyon, Nespelem silt (Qn) overlies the till, but later Pleistocene deposits are missing on the south side of the canyon. The exposure of bedrock near the river level on the south side indicates that the till has been eroded away.

Since the Pleistocene, the river has been incising its channel into the giacial age deposits. The topography of the south side of the canyon suggests that the river migrated as far south as the 1,400 ft contour, cutting several small terraces as it moved deeper and northward. The site itself is located on a bench, or terrace, sloping gently from 1,000 ft down to the bank at 950 ft. Although no regional formation is mapped for this terrace, it was presumably cut by the Columbia River. Deposits of the Columbia River would be





ļ

- こうないに、ため、「「ない」ないのないでは、こうないとう

a a la caractería de la c

12

1. 19 1 expected to occur, overlying Pieistocene deposits such as the proglac'al gravels (Qpg) which outcrop on the bank of the terrace to the east. The two granite outcrops at each end of the terrace would have affected the currents, and thus the deposition, in the site area.

At this time, the terrace lies along a two mile straight reach (RM 555-557) of the Columbia River. The rapids across from Box Canyon are undoubtedly caused by a bedrock obstruction in the channel, continuous with the bedrock exposed at the upstream end of the terrace. The rapids may have been a relatively stable feature over the last few thousand years.

PROCEDURES

A. 18

The locations of profiled walls at Areas A, B, and C are shown in Figures 2-2 and 2-3. The stratigraphy crew profiled walls in 17 units in Areas A and B while the rield school students profiled all of the units in Areas C and most of those in Area B. In analyzing and describing the stratigraphic units at the site, the stratigraphy crew profiles were given more weight than the student profiles. Consequently, the data for Areas A and B are considered more reliable. Areas B and C are grouped in Table 2-1, which summarizes the depositional history, not because they are necessarily similar, but because they are close together and there is little data available for Area C. Representative profiles from Areas A and B are shown in Figures 2-4 and 2-5.

DEPOSITIONAL HISTORY

The oldest deposit encountered in excavation at the site, DU I, is a layer of rounded river cobbles found underlying Stratum 400 in 74S174E. On the basis of comparison with other sites in the project area, these are undoubtedly part of a more extensive river channel deposit.

The overlying depositional unit, DU II, is the product of alluvial fan deposition. Two site wide strata, 400 and 300, were recognized, as well as a localized ash deposit, Stratum 250, which has been identified as Mt. St. Helens tephra, and might be St. Helens S-set tephra (ca. 13,000 B.P.) or Mt. St. Helens Yn tephra (ca. 3400 B.P.) (Davis 1984).

Stratum 400 was encountered in only a few excavation units, as most units terminated above it. Apart from the river cobbles, Stratum 400 has the coarsest matrix and the most gravel of all the deposits at the site. The gravel is decomposing granite, and is more frequent in Area B, which is topographically higher and further inland than Areas A and C. The distribution of the gravel is more consistent with deposition in an alluvial fan from bedrock weathering upslope than it is with bedrock weathering in situ.

The boundary between Stratum 400 and the overlying Stratum 300 is clear, and distinguished primarily by higher color values in Stratum 300. Particle size and gravel content decrease more gradually in Stratum 300, indicating that this is a conformable boundary. The sediments of this deposit vary more than those of the other deposits, with particle size decreasing toward the

1

A CARLER CALL





15



• • •

i

 $\boldsymbol{\lambda}_{1}$

Figure 2-3. Location of profiled walls in Areas B and C, 45-D0-282.

Table 2-1. Summary of depositional units, 45-D0-282.

Depositionel	Sitewide	Physical Description
Unit	Stratum	
III overbank and acolian deposition	50 overbank and aeolian medi- ments with littermet end B horizon	At fine send, moderately to poorly sorted, greyish brown [10YR5/2] with some darker colors, dark grayish brown to very dark grayish brown [10YR4/2-3/2], hard to soft, boundary clerr. B and C: Loemy send to sendy Loem, send is fine to medium, poorly to moderately well
		eortad, brown to grayish brown (10YR5/3- 5/2), soft to hard, boundary clear.
	100 overbank and ectian deposite	As sand to sendy loam, send fine, moderately to moderately well sorted, light brownish grey to grayish brown (10YRS/2- 5/2), moft, boundary gradual to clear,
		B and C: as above except less well sorted, pale brown to brown (10YR6/3-5/3),
	200 overbank deposits	At send to sendy loem, send medium to fine, woderately to well sorted, light gray and pele brown (10YR7/2-6/3) to grayish brown and dark brown (10YR5/2 and 3/3), soft, boundary clear,
		B and C: sand to loamy sand, sand medium to fine, poorly to moderately morted, light gray and pale brown to brown (10YR7/2,8/3- 5/3), soft, boundary gredual.
II siluvisi fan deposits	250 secondary ash deposit	As clay loss with some fine gravel, 35% volcanic ash, compact when dry, soft when wet, light gray (10YR7/2), boundary clear, found only in BSM200E,
	300 fine aliuvial fan deposita	At variable textures from send to clay loss but predominently fine grained, moderately sorted, colors range from pale brown [10YRS/3] to dark grayish brown (10YR4/3], finer mediments are hard, coarser mediments soft, boundary clear, more gravel than Stratum 200,
		B and C: send to sandy loss, mands medium to fine, poorly to moderately well sorted, wide color ranges with very pale and pale brown (10YR7/3-6/3), light bray (10YR7/2) and grayish brown (10YR5/2) dominant, loose to soft, clear boundary, more gravel than Stratum 200,
	400 coerse alluvial fen deposits	All: coarse to fine-grained send with some decomposing granite gravel, poorly to moderately sortad, soft to hard, brown to dark grayiah brown (10YR5/3-4/2), more gravel than Stratum 300,
I Columbie River gravels		B: rounded river cobbles underlying Stratum 400, encountered only in 745174E.



STRATA DESCRIPTIONS

- A: Dark greyish brown (10YR4/2) Loamy fine grained send. Organic mat-roots, twigs, grasses, Soft consistence, Moderately sorted, Boundary: abrupt, wavy.
 I: Greyish brown (10YR5/1-5/2) fine grained sendy Loam. Herd consistence, Moderatily sorted (high silt content). Boundary: abrupt, wavy.
 II: Dark greyish brown (10YR4/2) sendy Loam. Herd consistence, Moderately content. During a start of the sendy Loam. Herd consistence, Moderately content.
- sorted. Boundarys abrupt, wavy (but disturbed). This is the Al
- Greyish grown (10YR5/2) fine greined sendy Loss, Soft consistence, Moderstely sorted, Contains possible feature, Boundarys clear, wavy. This is the A2 horizon. IIII
- Light brownish gray [107R6/2] sandy loam. Soft consistence. Moderate to poorly corted. Occasional fine gravel. Boundary: Clear to graduel, smooth. This is the A3 horizon. Grayish brown [107R5/2] sandy Loam. Soft consistence. Moderately to IV:
- V:
- pourly sorted. Feirly abundant fine graveL. Boundarys clear, wavy. Light gray [10YR7/2] cluy Loam. Feirly compact but soft consistence when dry. Occasional fine graveL. Moderately sorted. Boundarys VI: clear. wavy. VII: Pale brown (10YNE/3) sandy Loss to Lossy sand,
- Coarser then any of the above strata, Soft consistence, Occasional fine gravel. Moderately sorted, Boundary: unknown,

Figure 2-4. Frofile of 83N200E, Area A, 45-D0-282.



STRATA DESCRIPTIONS

- A: Brown (10YR5/3) Loamy send. Organic Litter mat- roots, gresses, Boundary: shrupt. wevy. Brown (10YR5/3) Losmy sand. Moderately sorted. Soft consistence.
- II Slightly harder and darker in color than Level IL. This is the A Horizon, Boundary: gradual, way. II: Brown (10YR5/3) Losmy send. Contains some coarse send, but is
- predominantly fine send. Somewhat more friable in the upper half of the level. Same as Level I but lighter in color,
- III: Yellowish brown (10YR5/4) Losmy send. Moist, slightly harder than Level II, more compact, different in color, and somewhat coarser grained. Boundary: clear to gradual, wavy. Brown to dark brown (10784/3 when wet) Losmy sand. Same as Level III,
- IIIA: but darker in color, slightly finer grained, and more compact,
 - Boundary: clear, wevy. IV: Brown (10YR5/3) fine and medium mand. Single grain to fine blocky structure. Moderately sorted. Very wet. Lighter in color than Level
- III. Boundary: greduel, irregular. IVA: Brown (10YR5/3 to 4/3 when wet) send. Similar to Level IV but derker in color and somewhat finer grained. Boundary: unknown.

Figure 2-5. Profile of 16N6W, Area B, 45-D0-282.

الرسوم ترمان مارم وجرم الرمان موجوع فراغو المواجر المواجر المراجر الراجر المراجر الموجر المواجر المواجر المراجر

river. Stratum 300 appears to be a continuation, at a diminished energy level, of the alluvial processes at work in Stratum 400. It is also possible that there is some some overbank contribution, especially close to the river. The great variability in color in this stratum is due to variation in both texture and moistness of the sediments.

During laboratory analysis, Stratum VI from 83N200E (Figure 2-4) was found to contain 35% volcanic ash identified as Mt. St. Helens tephra by Davis (1984). On the basis of other characteristics, the stratum in which the ash occurred is part of sitewide Stratum 300, but because of its unique content it was labelled Stratum 250. Although Stratum 300 in the adjacent units has a similar matrix, ash was not noted. The isolated horizontal occurrence of the ash indicates that it is a secondary deposit. It therefore provides only a maximum date for the age of Stratum 300.

でたたたとう

Above the alluvial fan deposits are a series of overbank deposits, DU 111. Three sitewide strata, 200, 100, and 50, were recognized.

Stratum 200, inferred to be overbank deposition, is a more massive deposit with more uniform characteristics sitewide than Stratum 300. In Area A, Stratum 200 is coarser, sandier, better sorted, and looser than the underlying units, although similar in color, while in Area B, Stratum 200 is finer, loamier, harder, and more poorly sorted than the underlying unit. These contrasts indicate a change in the direction of deposition, which would be the case if overbank deposition was more prevalent than alluvial fan deposition. Despite the greater uniformity in texture, a wide range of colors were recorded for this stratum. This is due primarily to variations in the moisture of the sediments.

The sediments of Stratum 100 are yet more uniform, markedly finer, somewhat loamier, and somewhat more poorly sorted than Stratum 200. The textural differences may indicate a fining upwards in the overbank sequence, or an increase in aeolian contribution. In either case, the gradual boundary indicates that Stratum 100 overlies Stratum 200 conformably. The colors of the sediments in Area A are the same in value, but consistently grayer than those in Areas B and C, indicating a more reducing environment at Area A. It is possible, therefore, that the observed physical difference is more a function of soil development than a change in depositional regime. While soil formation is a postdepositional process, it is generally uniform relative to the topography, and may develop along textural lines. In the absence of a distinct boundary between Stratum 100 and 200, this soil horizon is probably an adequate time marker.

Stratum 50 is coarser, loamier, and more poorly sorted than Stratum 100. The litter mat and underlying B horizon are contained in this depositional unit. The sediments have been affected by soil forming processes and are dark grayish brown and darker colors. It is possible that the upper 15-25 cm of this unit are a plow zone but this was not corroborated by local informants. Although there are textural distinctions between Strata 100 and 200, the boundary between them, generally found at 75-100 cm below the surface, is gradual and recognized largely by color differences.

»»» (> \$\${\${\${}}}

In summary, DU I represents the passage of the Columbia River channel across the bench. After the river withdrew, an alluvial fan began to aggrade, protected from fluvial erosion by the granite outcrop at the upstream end of the bench. DU II was the result. The fan aggradation rate diminished, and came into equilibrium with the local base level, and overbank deposition became dominant (Stratum 200). As the elevation of the overbank deposits relative to the river channel increased, the rate of deposition slowed, and aeolian deposition made a greater contribution (Stratum 100). The relative stability of the surface in Stratum 100 and 50 allowed soil development to begin.

ANALYTIC ZONES

The four sitewide stratigraphic units containing separate peaks of cultural material were defined as cultural analytic zones. Table 2-2 summarizes the stratigraphic definitions and cultural contents of each zone. Due to a lack of charcoal or other datable organic material, no radiocarbon dates were obtained for any of the deposits. A lack of patterning in artifact distribution, a lack of bounded cultural features, and a lack of lighter materials such as charcoal, bone, and shell suggests that they may be secondary or deflated deposits. However, the small assemblage of projectile point types (see stylistic analysis) indicates that there are discernible differences in the temporal distribution of the cultural deposits, and justify division of these strata into four analytic zones. Each zone is discussed individually below.

ZONE 4

The cultural materials from DU II (Strata 250, 300, 400) have been assigned to Zone 4. This is the smallest of all four assemblages, although it has the second largest excavated volume. The cultural assamblage includes lithic and bone scatters. No FMR (fire modified rock) or shell was recovered. Although cultural features were recorded, they are all unstructured features identified on the basis of content alone and defined within arbitrary unit levels within 2×2 -m excavation units. The Zone 4 materials occur with the redeposited Mt. St. Helens ash deposit (Stratum 250). Stylistic evidence indicates this zone dates before 6000 B.P., which implies that the ash must be the S-set tephra (ca. 13,000 B.P.). Zone 4 was excavated in all but a tew units in Areas B and C. These were terminated above DU I.

ZONE 3

国家としてものなるのからのなど

Stratum 200 (DU III), containing the largest cultural assemblage of the four depositional units, is defined as cultural Analytic Zone 3. Like the other zones, the assemblage is dominated by lithics and only a very small percentage bone, shell, FMR and miscellaneous items. The feature assemblage is the largest of all the zones. Although features were recorded, they are all

Table 2-2. The analytic zones, 45-D0-282: their stratigraphic definition and contents.

Zone	3	ßtratum	Description	Lithic ¹ number	Nor-Lithic ² number		Shell ³ 9		Total	Miscel Lansour/ Historic	feetures	Vot une [13]	Denaity (objecta/e ³)
-	•	8	Overbank and seolian	2,763	24	68 24			2,860	-		45.0	63.6
Q I	0	100	Upper overbank deposit	3,820	35	52	^ی ۳	6 844	4,021	ø	e	40.7	8.8
~	**	500	Lower overbenk deposit	6,877	"	51 17	58 C	1.6	7,037	ņ	œ	53,1	132.5
•	9	820 300 401	Alluviel for	1,610	36	~\$	i cu	i i	1,655	Cu	Q	£°.74	35.0
8	ł	1	Beach Leg	1,394	13	32 145	ius	90 32,687	1,527	0	1	1,1204	1
1 Incl. 2 Incl. 3 Count		/4 in flak	es. ts, ochre, histor sight is totel ah	ric, and mi neil weight	iscel Lanaoua.								

21

VALUE AND A CONTRACTOR

in i
unstructured and defined on the basis of content alone. Cultural materials in this zone may be less disturbed or reworked than those in Zone 4 because of the dominance of overbank deposition. This zone was excavated in all units.

ZONE 2

Zone 2 corresponds to Stratum 100 (DU III). This zone has a larger and more diverse assemblage than that of Zone 1, even though it represents a slightly smaller excavated volume. The assemblage is dominated by lithics, but also includes FMR, bone, shell and cultural features. The features recorded were all unstructured and defined by content alone. The sediments are overbank and aeolian deposits resulting from processes which should not have disturbed the cultural materials. This zone was excavated in all units.

ZONE 1

「「いい」であった。ここには国際にたっていたとうです。

Zone 1 includes the cultural materials recovered from Stratum 50 (DU III). The assemblage consists largely of lithics, with but a few bones. No features, shell or FMR were recovered. This zone was excavated in all units. A plow zone has reworked the upper 20-30 cm of deposit.

THE BEACH CULLECTION (ZONE 25)

The surface on which the beach lag collection occurred was labeled Zone 25. It is an erosional beach surface cross-cutting several of the deposits at the site. Although not comparable to the other zones in duration and formation processes, the zone yielded a large assemblage containing valuable information on artifact types and morphology.

3. ARTIFACT ANALYSES

BARGELING DOWNOOD PRODUCE

NCOCCOLUMN

BARETAL ALL TO BAR

The second of th

Artifacts recovered from site 45-D0-282 have been subjected to three separate analyses. <u>Technological analysis</u> describes elements of prehistoric tool manufacture, detailing processes of lithic reduction. <u>Functional</u> <u>analysis</u> describes attributes of wear on tools and develops inferences concerning the use of tools at the site. <u>Stylistic analysis</u> describes morphological elements that have demonstrated temporal and regional significance and compares recovered artifacts with types defined outside of the project area.

Stone artifacts are treated in the most detail, with other materials entering the classification only when specified attributes are applicable. Analyses were intentionally biased towards lithics with the assumption that these artifact classes would be of the most value in comparisons with other researchers' work and in developing reconstructions of site activities. Artifacts of bone, shell, and other non-lithic materials, though included in the classifications wherever appropriate, are only described in detail selectively.

All artifact analyses take the form of paradigmatic classifications as defined by Dunnell (1971, 1979). In this system, dimensions are selected which can describe morphological variation in the collection. These dimensions may correspond to defined stages of tool manufacture, be characteristic of specific tool uses, or be indicative of limited periods of time depending on the purpose of the classification. Dimensions are divided into sets of attributes. In analysis each artifact is identified by a single attribute from each dimension. By cross-tabulating the dimensions, sets of comparable but mutually exclusive classes are formed. From study of these classes, inferences may be drawn concerning the nature of tool manufacture, use, and distribution in time and space.

However, classificatory dimensions and constituent attributes are not always truly exhaustive in practice and must be viewed as gross analytic categories designed to signal obvious morphological variation. Whenever possible, our defined attributes approximate characteristics identified in prior research as important technological, functional, or stylistic indicators. Further, it will be apparent that analytic levels within the paradigmatic classifications often preclude direct comparison with more traditional typological approaches. For example, in several instances these analyses will focus on the tool, and not on the artifacts, because an artifact may have more than one tool or use. These classes are then only related to more standard classifications by cross-correlation with more traditional artifact designations (e.g., biface, drill, or chopper). Discussion,

. 1

R

therefore, involves analysis both at the level of the tool and of the artifact. This distinction will be used throughout.

In the following subsections we present the descriptive data from technological, functional, and stylistic analysis. Most data are summarized in tables with text largely reserved for discussion and interpretation of major points. Brief explanations of dimensions and attributes used in each analysis are presented at the beginning of each subsection. Introductory tables list the attributes corresponding to each classificatory dimension. All data tables are confined to the appendix. Only interpretive illustrations are included within the text proper.

Artifact analyses will be presented with reference to the five analytic zones outlined in Chapter 2. All four excavated zones correspond to the Kartar Phase (ca. 7000-4000 B.P.) defined for the Rufus Woods Lake project area. The beach collection (Zone 25), represents an eroded remnant surface, with diagnostic projectile point types that indicate a temporal range comparable to that documented for the four excavated zones. We include the beach collection in all the following descriptive tables since it is of comparable age, and can be used to assess the validity of patterns observed in the other four zonal assemblages. Because the site had neither activity surfaces nor bounded cultural features, the artifact assemblages within the zones cannot be divided further; consequently, the zones must serve as our finest units of analysis. Discussion of artifacts in association with the arbitrarily defined unit level features is reserved for Chapter 5. This chapter will also treat the spatial patterning in the artifact classes.

TECHNOLOGICAL ANALYSIS

の目的ななので、「ならられる」となっていた。そののないないで、「ないないない」ではないない。「「ないないない」」となったので、「「ないないない」」で、「ないないない」で、「ないないない」で、「ないないない」

Prior researchers have described general manufacturing sequences in the production of stone tools, and have thereby identified specific morphological elements associated with certain methods of production and particular steps in the reductive sequence (e.g., Crabtree 1972, 1967a,b; Flenniken and Garrison 1975; Muto 1971, 1976; Smith and Goodyear 1976; Speth 1972; Stafford 1977; Swanson 1975).

While the process of lithic reduction may vary greatly even within defined industries, an idealized trajectory of reduction, with certain fundamental steps, can be constructed. First, the knapper selects a nodule which will serve as a core for the production of flakes of suitable size and shape. The first flakes removed exhibit the weathered surface of the stone. Later flakes show little or no weathered surface, and may have flake scars from the initial flaking. All of these flakes may be removed with a hard hammer of stone, and this creates distinctive large flakes with pronounced bulbs of percussion, strong stress lines, and crushed striking platforms. Once flakes are of a suitable size, the knapper modifies them further with a soft hammer of antier or wood, producing smaller flakes with less pronounced bulbs of percussion, finer stress lines, and little or no crushing of the striking platforms. Later, after the artifact has been roughed out to the desired shape, the knapper may remove still smaller flakes with an antier time to sharpen, finely shape, and maintain working edges on the tool.

This is, of course, an extreme simplification. Not only are there innumerable variations in the sequence of steps and tools used, there are also several related processes with distinctive steps and products. The above description characterizes a flake tool technology, wherein hammers of different materials are used to detach thin, lamellar flakes by direct percussion. There is a related blade industry, where hammers or punches are used to create long, harrow flakes with prismatic cross sections. This technique requires a more prepares core, and may involve indirect as well as direct percussion (cf., Leonhardy and Muto 1972; Muto 1976). In turn, these industries may be contrasted with a microblade industry which calls for the creation of small, carefully prepared wedge-shaped cores and use of fine fabricators for detachment of thin, parallel-sided flakes. The small, thin blades have one or more arrises on the dorsal surface, and are themselves finished tool forms requiring no further modification (cf., Sanger 1968, 1970). While clearly distinct, these three industries need not have been independent, as one could easily complement the others as part of a more comprehensive industry. That this is in fact the case is suggested by the presence of tlake and blade industries in early ascemblages on the Columbia Plateau (Leonhardy and Rice 1970; Leo.hardy et al. 1971; Munsell 1968; Muto 1976).

Artifact types are the most obvious practical indicators of lithic industries (e.g., cores, blades and flakes, and tools made from blades or flakes). Core configuration is distinctive; flakes, blades, and microblades are also readily distinguished. Tools often evidence attributes of origin like arris remnants or striking platforms. Other characteristics, though quite recognizable, are less certain diagnostic indicators, and often blend into the general signposts of lithic reduction outlined above (e.g., detritus, flake size, presence or absence of cortex, etc.).

であっている。このでは、「「「「」」というという。「「」」というという。「「」」というという。「「」」というという」「「」」というという」「「」」というという」「「」」」というという」「「」」」というという

In technological analysis, we record attributes indicative of these steps in stone tool manufacture, and characteristic of these varied reduction techniques.

Technological acaiysis makes use of seven dimensions: OBJECT TYPE, MATERIAL, CONDITION, DORSAL TOPOCCAPHY, TREATMENT, KIND OF MANUFACTURE, and MANUFACTURE DISPOSITION. These describe the kind and condition of artifacts and the materials from which they are made. Descriptive attributes of WEIGHT, LENGTH, WIDTH, and THICKNESS are also measured, and supplement the classificatory dimensions. Table 3-1 lists these dimensions and attributes.

Technological analysis at 45-D0-282 utilized an abbreviated form termed LITHAN-X, which applied standard analytic procedures only to the NW quadrant of any excavated unit. Other quadrants within the unit received only partial analysis. Artifacts in the NW quadrant were identified according to all five dimensions. Artifacts recovered elsewhere in the square were only identified according to MATERIAL TYPE and OBJECT TYPE. Exceptions are field-catalogued objects and flakes smaller than 1/4 in. These were subjected to standard analysis, irrespective of provenience. This means that counts of material types and object types will be higher than those tabulated for condition, dorsa! topography and treatment. Worn and manufactured objects and flakes smaller than 1/4 in are subjected to full analysis, and included in the above

Table 3-1. Technological dimensions, 45-D0-282.

A REAL PROPERTY OF THE REAL

DINENSION I: OBJECT TYPE DIMENSION V: TREATMENT Conchoidel flake Definitaly burned Dehydrated (heat trestment) Chunk Core Linear flake ATTRIBUTE I: WEIGHT Unmodified Tabular flake Recorded weight in grams Formed object Vaathared ATTRIBUTE II: LENGTH Indeterminate Flakest Longth is measured between the point of impact and the distal and along the bulber exis DIMENSION II: RAW MATERIAL* Jacost Chal cedony Other: Length is taken as the Longest dimension Patrified Wood Obsidian ATTRIBUTE III: WIDTH Cont Quertzite Fine-grained quartzita Flekast width is measured at the Beselt widest point perpendicular to the Fine-greined beselt bulber axis Silicized mudstone Argillite Other; width is taken as the Grenite maximum measurement along an axis perpendicular to the axis of Length Sil tatons/mudstone Bone/entler ATTRIBUTE IV: THICKNESS Ocher DIMENSION III: CONDITION Flakes: thickness is taken at the thickest point on the object, Complete excluding the bulb of percussion and Proximal fragment the striking platform Proximal flake Less than 1/4 inch Other: thickness is taken as the measurement perpendicular to the Broken width measurement along an exis Indeterminete perpendicular to the axis of Length DIMENSION IV: DORSAL TOPOGRAPHY None Partial cortex Complete contex Indeterminate/not applicable * Only those raw materials recorded from the site are listed

' Only those raw materials recorded from the site are listed here; a complete list is evailable in the Project's Research Design (Campbell 1984d).

26

1. N. 1.

19 M

.

į.

k,

ę

ţ.

£

.

с. 1

,

,

• 20

.

•

5

...

ŧ.

1. 1. W. L. V.

categories.

We will first examine the range of material types recovered, and then the types of objects present. By examining such technological dimensions as material, object type, type of manufacture, treatment, dorsal topography, and flake size, we will make inferences about the nature of lithic reduction at the site. These are admittedly crude indicators, but they should provide the data necessary to describe the sorts of stone tool production present at the site. When analyzed by distribution over analytic zones and in cultural features, these diagnostics will also allow us to make inferences about changes over time and the use of space in any defined period.

MATERIAL TYPES

Ninety-five percent of the worked stone recovered from site 45-D0-282 was either jasper (57%) or chalcedony (38%) (Table 3-2). No other material type exceeds 2% of any zonal assemblage nor 1% of the total assemblage. Distribution across zones is very even, except for a dramatic increase in jasper in Zone 25, the beach collection. This zone also produced slightly higher quantities of quartzite, fine-grained quartzite and basait, and markedly lower quanties of chalcedony than found in the excavated zones.

Non-lithic materials were rare a: this site (Table 3-3), consisting solely of ocher, three glass fragments, and two poorly preserved bone/antier tool fragments. Again, none of these exceeded 2% of any zonal assemblage and only ocher tallied 1% or above.

Material			Zan	•		Tatal
	1	2	3	1	25	iocat
Ocher Column %	24 01	35 01	78 01	36 02	1	174
Glass Column X	1	1	1	-	-	3
Bane/Antier Column %	-	-	-	-	2	2
TOTAL	25	38	79	36	З	179

Table 3-3. Count of non-lithic material by zone, 45-D0-282.

OBJECT TYPES

Jasper and chalcedony conchoidal flakes are easily the most common object type in the collection, comprising 88% of the artifacts recovered (jasper, 52%; chalcedony, 36%) (Table 3-4). Other object types occur in the following order of descending frequency: chunks, of all material types (5%), conchoidal flakes of other than jasper and chalcedony (3%), formed objects (2%), microblades (1%), cores (.3%) and unmodified objects (.2%). Jasper and chalcedony are the most frequent minerals for all object types except the unmodified category, which includes hammerstones and pesties. Chalcedony is

			Zone			
Material	1	2	3	4	25	Totel
Jatper	349	518	819	166	1,111	2,962
Column %	50.5	50.3	47.7	47,3	81.2	
Chelcedony	315	473	830	171	158	1,947
Caluen 3	45,8	48,0	48,4	49.0	11,5	
Petrified wood Column %	3 0,4	0.2 0	13 0.8	0.0	9 0.6	26
Obsidian	0	3	4	1	1	9
Column X	0.0	0.3	0.2	0,3	0.1	
Opal	8	3	6	3	0	20
Column X	1.2	0.3	0.3	0,9	0.0	
Quertzite	1	0	4	0	22	27
Column X	0,1	0.0	0,2	0.0	1.8	
Fine-grained quertzite Column %	0 0.0	5 0,5	1 0.1	1 0.3	17 1.2	24
Baselt	2	12	17	3	27	81
Column X	0,3	1 . 2	1,0	0.9	2.0	
Fine-grained baselt	5	5	5	2	2	19
Silicized	0	0.5	2	0.8	2	4
Column %	0.0	ə. 0	0.1	0.0	0.1	
Argillita	6	5	11	3	4	29
Column X	0.9	0,5	0.5	0,9	0,3	
Granitic	0	0	3	0	7	10
Column X	0.0	0,0	0.2	0.0	0,5	
Sendstone	0	0	0	0	1	1
Column %	0.0	0,0	0.0	0.0	0.1	
Indeterminate Column %	2 0.3	3 0.3	1 0.1	0.0	8 0.6	14
TOTAL	891	1,029	1,718	349	1,368	5,153

Table 3-2. Count of lithic material by zone, 45-D0-282¹.

1<1/4 in flakes deleted.

1. S. W.

1 Martine Strange

1. S. S. S. S.

more prevalent than jusper in the microbiades (chalcedony 69%; jasper 31%) and equals jasper in the cores (chalcedony 44%; jasper 44%). Jasper is the more common in conchoidal flakes, chunks and formed objects.

Material by			Zone			Tanal
	1	2	3	4	25	
Conchat del fi elle						
Jasper	321	478	766	151	944	2,660
Chal cedony	289	454	799	159	138	1,839
Baselt	7	13	19	5	8	52
Quertzite	1	5	5	1	15	27
Argillite	5	5	11	3	1	25
Petrified wood	3	2	13	0	5	23
Obsidien	-	3	4	1	-	8
Opel	6	2	8	2	-	16
Silicized						-
nudstons	-	-	2	-	-	3
Hi crobi ade						
Jasper	4	3	4	2	4	17
Chal cadony	8	3	18	6	3	38
Chunk						[
Jaspar	16	31	43	10	83	183
Chel cedany	15	8	11	4	7	45
Bamelt	-	-	5	-	-	2
Quertzite	-	-	-	-	4	•
Argillite	-	-	-	-	1	1
Petrified wood	-	-	-	-	1	1
Obsidian	-	-	-	-	1	1
Opel	1	1	-	-	-	2
Core						
Jasper	-	1	1	1	4	7
Chel cedony	1	1	2	1	2	7
Beneit	-	-	1	-	-	1
Argillite	-	-	-	-	-	1
Furmed object						
Jesper	6	5	5	1	67	64
Chal cadr ny	1	7	-	1	7	16
Beselt	1	2	-	-	11	14
Quertzite	-	-	-	-	3	3
Argillite	1	-	-	-	1	2
Petrified wood	-	-	-	-	1	1 j
Opel	1	-	-	1	-	2
Silicized						
nuditom	-	-	-	-	1	1
Urmodified						
Jasper	-	-	-	-	1	1 [
Basel t	-	-	-	-	6	6
Quertzite	-	-	-	-	1	1
Grani ti c	-	•	1	-	3	4
TOTAL	687	1,024 1	,713	349 1	,329	5,092

Table 3-4. Material by object type by zone, 45-D0-282.

Zonal distributions of object types are remarkably regular, with the percentage of object types very uniform across the four excavated zones and the Zone 25 beach collection. The only exception is the formed object category: in Zone 25 formed objects make up 7% of the total assemblage; in the other zones, this category makes up less than 1.5% of the zonal assemblage.

والمرجوع والمرجوع والمرجوع والمرجوع والمرجوع والمتعاد المتعاد والمعاد والمعاد والمعاد والمعاد والمعاد

MANUFACTURE

Chipping accounts for 99% of the manufacture observed for objects in the collection (Table 3-5). However, only 4% of the total number of objects recovered show any manufacture beyond initial detachment from a core or blank, or resharpened tool. Chipped objects constitute no more than 3% of the objects in any zonal assemblage, with the exception of Zone 25, where the 10.2% tally reflects the higher number of formed objects recovered.

•			Zone			
iy pe	1	2	2 3 4		25	Total
Mone Calumn %	671 97,1	1,009 98,1	1,698 89,0	344 98.8	1,227 89,7	4,949
Chtpping Column %	20 2,9	20 1.9	18 1.0	5 1.4	139 10,2	202
Indeterminet Column %	• -	-	-	-	2 0.1	2
TOT AL	691	1,029	1,716	349	1,358	5,153

Table 3-5. Type of manufacture by zone, 45-D0-282¹.

1<1/4 in flakes and non-lithics deletad,

Heat treatment prior to manufacture appears to have been common, with 4-7% of each zonal assemblage listed as burned or dehydrated (Table 3-6). Jasper and chaldedony frequently exhibit burning (Jasper 75%; chalcedony 23%) (Table 3-7). Two examples of opal and fine-grained basait are also listed as burned. All four specimens listed as dehydrated are opal; however, this is probably the result of natural dehydration since opal tends to dehydrate when exposed.

Table 3-6. Count of treatment by zone, 45-D0-2821.

**							
1100100015	1	2	3	4	25	Total	
None Column 12	548 93.8	978 94,9	1,843 95,7	323 82,5	1,314 96,1	4,904	
Burned Column X	40 5,8	53 5,2	72 4,2	26 7,4	54 3,9	245	
Dehydrated Goluen X	3 0.4	-	1 0.1	-	-	4	
TOTAL	691	1,029	1,716	349	1,368	5,153	

1<1/4 in flakss and non-lithics deleted,

Treatment by			Zone			
mterial	1	2	3	4	25	Total
Jesper	394	470	787	1.67	9.087	2.778
Russed	28	49	52	18	24	184
Tatal	348	518	81	185	1.111	2.063
Chal cadeny						
Norm	304	463	811	154	146	1,810
Surned	11	10	18	7	10	57
Tetal	315	473	81 0	171	156	1,847
Petrified wood		_			_	
Nore	3	2	13	-		
Total	5	8	13	-		
UBAIGIAN		•			•	
Tone	-	3		1		2
19181	۲	٤	13			
			5	2	-	14
Runna		-	-		-	2
Debudeeted		-	1	-	-	- I I
Total	ĭ	3	ė	3	-	20
Quertrite	•	•	•	-		
Xeen	1	-	4	-	22	27
Total	1	1	4	-	22	27
Fine-gratned						
quertaite						
None	-	5	1	1	17	24
Total	-	5	1	1		24
Beenlt	•		47		-	
Tatal	2	12	17	3	2	
Finnensing	•	14	.,		•	•••
been t						
Nore	5	4	4	2	2	17
Burned	-	1	1	-	-	2
Total	5	5	5	2	5	19
Stlicized						1
	_	_		-		
Totol	-	-	2	-	2	
Jacal Appli 1444	-	-	6	-	•	~ {
Mana		5	11	3	4	28
Tetal	ž	š	11	ā	Ă	29
Granitic	•	-	••	-	•	(
None	-	-	3	-	7	10
Total	-	-	3	-	7	10
Sandstans						
None	-	-	-	-	1	1 1
Total	-	-	-	-	1	1
Inde termine to	-				-	
None	2	3	1	-		14
Totel	2	3	1	-		14

Table 3-7. Heat treatment by material by zone, 45-D0-2821.

1<1/4 in fiches and non-lithics deleted.

Primary reduction of all material types occurred on the site, with 4-8% of all object types recovered per zone having either partial or complete cortex (Table 3-8). Of the stones with cortex, 62.6% are jasper and 17.8% are chalcedony (Table 3-9). The remainder are primarily locally available quartzite and basalt. The majority of object types with cortex are cryptocrystalline conchoidal flakes (73%) (Table 3-10). Cryptocrystalline chunks comprise 5.7% and non-cryptocrystalline formed objects make up another 5%. Other categories never exceed 3%. Only five of the sixteen recovered cores show any cortex remnants, and four of these five are cryptocrystalline stones.

Dorset			Zone			
topogrepny	1	2	3	4	25	IOTAL
None	633 91.9	977 94,9	1,599 83,2	321 92.0	1,291 94,4	4,823
Partial cortex	53 7.7	45 4,5	112 6.8	28 8.0	48 3.5	291
Complete cortex	- 0.0	1 0.1	1 0.1	- 0.0	4 0.3	6
Indaterainete	3 0.4	5 0,5	0.0	 0.0	25 1.8	33
IDTAL	691	1,029	1,718	349	1,368	5,153

Table 3-8. Count of dorsal topography by zone, 45-00-282¹.

1<1/4 in flakes and non-lithics deleted.

Secondary reduction and finishing/maintenance of stone tools on the site is documented in Table 3-11, which lists conchodial flakes by size. Most (88%) of the conchoidal flakes are larger than 1/4 in, 12% are less than 1/4 in, and only .04% are less than 1/8 in. Of the less than 1/4 in and 1/8 in flakes, 34% are jasper, 65% are chaicedony, with the rest, opal, obsidian and various other noncryptocrystalline stones. This closely replicates the pattern observed in the distribution of material types and object types.

Weight, length, thickness and weight measurements taken on conchoidal flakes reflect the use of cryptocrystalline stones for most tool forms, as well as providing insights into the reduction process used. For example, Table 3-12, which lists average weights by material types for the five analytic zones, underscores the differential use of CCS and non-CCS stones. CCS--predominantly jasper and chalcedony--is consistently present in the lowest average weights, and shows very low standard deviations, reflecting the use of these stones for the widest variety of small tools and, perhaps, either very little variation in the size of cores or the kinds of tools made. It will be seen that the quartzite and basait specimens recovered are much larger, indicating very different use patterns for CCS and non-CCS stones. We also observe much larger CCS flakes of more variable size in the beach collection (Zone 25), accompanied by smaller basalt flakes of less variable size. This may indicate some difference in the natu of reduction between that area and those zones still in stratigraphic context, although the possibility of size sorting by wave action must also be taken into account. These patterns in flake size are also represented in Tables 3-13, 3-14, and 3-15. In them, we see a marked consistency in the size of CCS flakes relative to the presence or absence of cortex in all zones except Zone 25, where the flakes are larger and more variable in size. It is also apparent that CCS and non-CCS flakes are generally comparable in length and width, except again, in Zone 25, where both CCS and non-CCS flakes show marked increases in size.

and the second secon

-
• •
•.*
-
·,`
••
27
÷.
تبيلا
ر بر الحا ر
-
· · _]
5
- 1
- V-1
. {]
- 20
- 7
- 23
A.
- ÈS
]
- ° -
2.4
1
. 1
· 1
3
74
1
4
-
<u> </u>
-
1000

Table 3-9.	Amount	of	contex	by	material	by	zone,
45-D0-282							

Heterial			Zone			Total
debi cage	1	2	3	4	25	TUGEL
Jasper						
Nom	309	480	728	142	1,095	2,/59
Part cortex	38	32	91	23	2	188
Indeterminete		1	-	-	14	15
Total	347	518	819	155	3,111	5,860
LTAL CEDONY						
None	302	459	611	168	155	1,893
Part Cortex	13	13	18	0	1	51
Complete cortex	-	r	-	-	1	2
	945	479		474	450	1 0.67
	313	4/3	030	1/1	136	1,84/
Normal Norma	2	2	43	-		08
Total	3	2	13	474		20
Obsidias	3	6	15	17.1	9	20
Nom	-	3		4	4	0
Total	-	3	7			a 3
Onal	-	3	-	1	•	
Non	7	3 .	8	3	-	40
Dert Coster	4	-	Ŭ,	-	-	4
Total		3		2	-	20
Questrite			•	J		20
None	1	-		_	*	44
Part cortex	-	-	-	-	15	15
Indeterainete	-	-	-	-	1	1
Total	1	-	4	-	22	27
Fine-grained quartzi	ta		•			
None	-	5	1	1	14	21
Pert cortex	~	_	-	-	3	3
Totel		5	1	1	17	24
Basalt						
None	1	11	12	3	8	33
Part cortex	1	1	5	-	18	25
Complete cortax	-	-	-	-	5	2
Indeterninete	-	-	-	-	1	1
ICTAL	5	12	17	3	27	61
				•	•	40
Todete mi rete	3			2	<u>د</u>	10
Total		, k	5	-	3	40
Silicized sudstans	J	9		-	-	
Non	-	-	2	-	1	3
Part cortex	-	-	=	-	1	1
Total	-	-	2	-	ż	Á
Argillite					-	
None	6	5	11	3	1	28
Part cortex	-	-	-	-	2	2
Indetersinete	-		-		1	1
Total	8	5	11	5	4	29
Grenitic						
None	-	-	1	-	-	1
Pert cortex	-	~	1	-	6	7
Complete cortex	-	-	1	- .	1	5
Total	-	· -	3	-	7	10
Sandatone					-	
Indeterminete	-	-	-	-	1	1
i U CBL Todo tem tecto	-	-	-	-	7	וי
		-	4	_	•	
Todatami mta		3	_	-		
Total	2	3	4	-	0 0	44
	4		•	_		

1<1/4 in flakes and non-lithics deleted.

.

-

Table 3-10. Cortex on cryptocrystalline and other material by object type and zone, 45-D0-262¹.

Object type			Zone			7	
by cortex by meterial	1	2	3	4	25	lotal	
Conchoidel flekss							
No cortex							
Cryptocrystalline	572	896	1,488	285	1,081	4,323	
Other	12	23	35	9	24	103	
Pertial cortex				-			
Cryptocrystal Line	4/	41	100	27	-	215	
	1	-	•	-	1	•	
Cryptocrystalline		1	-	-	-	1	
Hi crobLades							
No cortex		-		_	-		
Cryptocrystalline	12	5	22	8	6	53	
Partial Cortex	-		-	_	_	_ 1	
Complete coster	-	1		-	-	1	
Cryptocrystalline	-	-	-	-	1	1	
Tabular flakes							
No cortex						1	
Cryptocrystalline	-	-		-	1	1	
Other	-	-	-	-	4	4	
Partial cortax							
Other		-	-	-	10	10	
Chunks							
No cortex							
Cryptocrystallins	27	38	46	14	83	208	
Other	-	-	1	-	2	3 [
Pertial Cortex	•	•	•		•		
Cryptocrystelline	5	2		-	2	12	
Veler	-	-	•	-	•	2	
Cores							
NO COPLEX	•		•				
Other	1	1	<u>د</u>	1	0	ער	
Partial corcar		-	•	-	-	'	
Cryptocrystelling	-	1	1	1	1	4	
Other	-	-	-	-	1	1	
Fa.wed objects							
No cortex						1	
Cryptocrystallins	8	12	4	3	74	101	
Cher	1	2	-	-	-	3	
Partial cortax			•			.	
Cryptocrystalline	-	-	1	-	-	1	
Uther	-	-	-	-	15	סו	
Urmodified							
No cortax					-		
Urypxoorystalline	-	-	-	-	٦	וי	
Purtiel Cortex	-	_	-	_	7	,	
Complete contex	-	-		-	1		
Other	-	-	1	-	3		
			•		-	-	

1<1/a - n flakes and non-lithics delated.

، د ر ر

	64- a		Za	ine			
Maseri al	8120 (tn)	1	2	3	4	25	Total
Jaapar	>1/4 <1/4 <1/8	1,253	1,709	2,909 341	637 82	1,111 29 1	7,529 675 1
Chal cadony	>1/4 <1/4 <1/8	1,136 199 -	1,816 308 7	2,881 633 -	639 136	158 4 -	8,410 1,278 7
Petrified wood	<1/4	7	6	28	3	8	52
Obsidian	>1/4 <1/4	7 1	14 2	7 2	8 1	1 -	35 6
Opel	>1/4 <1/4	12 1	7	9 1	8	-	34 2
Quertzite	>1/4 <1/4	3 -	1	6	1	22 1	33 1
Fine-grained quartzite	>1/4 <1/4	2 1	10 1	8 2	3	17	40
Beeslt	>1/4	15	29	68	13	27	150
Fine-gristined basels	t >1/4 <1/4	8	7	13 2	4	2 -	34 3
Silicized mudstone	>1/4	-	1	5	2	2	10
Argillite	>1/4 <1/4	:3 -	13 1	28 1	11 1	4	69 3
Granitic	>1/4	-	2	5	1	7	15
Sendstone	>1/4	-	-	-	-	1	1
Sil t/nudstons	>1/4	1	-	-	1	-	2
Hica	>1/4	12	46	49	62	-	169
Total	>1/4 <1/4 <1/8	2,479 284 -	3,481 452 7	5,894 982 -	1,389	1,360 33 1	14,583 1,972 8

1

Table 3-11. Count of flake size by material and zone, 45-D0-282.

KOOSSER EDECKAL ENDARM PARAMENTAR PARA

EXERCISE RESERVED FOR SECTION SERVICES REPORTED REPORTED

Ma & a al al			Zone			
ME COMI DI	1	2	3	4	25	Ideal
Cryptocrystalline						
x Budu	3.3	3.∠ 9.0	3.5 11.9	4.5	18.4	4.0
n	2,273	3,145	5,400	1,198	1,087	13,103
Quertzite						
x	3.7	94.0	47.8	7.0	115.5	48.5
s.d.	3.8	-,	93,5	-,	153.4	83.9
38	3	1	3	1	£	12
Fine-grained quartzite	4 0		47.0	4 0	05 O	42.0
s.d.		0.6	31.3	-	25.8	23.3
n	2	10	8	3	13	38
Basel t						
x	34.8	104.7	190.8	76.8	68.9	130.1
s.d.	83.1	382.5	555.4	261.0	107.3	433.8
n	21	38	67	61	8	141
Granitic						447.0
x a.d.	-	2/3.0	39.5	-	-	117.3
0	-	1	2	-	-	3
Obsidian						
x	1.7	1.2	1.0	2.0	-	1.4
s.d.	1.9	0.8	-,	2.5	-	1.4
n	'	13		0	-	33
Other Lithics						
*	2.5	4.8	32.8	4.5 40 5	9.5 2.4	17.0
n	12	13	32	13	2	72
Indeterminate Lithica						
x	3.5	-	7.0	-	36.0	17.2
s. d.	3.5	-	-,	-	14.1	18.7
n 	<u>د</u>	-	1	-	2	>
Total						
* • 4	3.5	4,2	8.0 es o	5.4	17.1	6.0
0	2.320	3.213	5.522	1.238	1.114	13.405
	_,					

9

i

Table 3-12. Average weight (in 0.1 g) of concholdally flaked material by zone, 45-D0-282¹.

1<1/4 in flakes, non-lithics, and non-conchoidal flakes deleted.

ļ

Mataria)			Zone			Total
	1	2	3	4	25	
No cortex						
Cryptocrystalline						
x	10.4	10.8	11.0	11.2	18.8	11.9
s. d.	4.4	6.2	5.7	5.1	11.0	7.1
n	305	4/2	814	143	270	2,004
uner T	10.7	10.7	10 7	11 2	23.4	12 6
s.d.	3.9	5.3	8.0	3.6	11.4	8.5
0	9	16	28	5	11	69
••	-			-	••	~~
Tatal						
x	10.4	10.8	11.0	11.2	18.9	12.0
s. d.	4.3	6.2	5.8	5.0	11.1	7.2
n	314	488	842	148	281	2,073
Partial cortax						
Cryptocrystalline						
x	15.6	14.8	16.4	21.7	-	16.7
a.d.	7.6	9.2	9.6	15.2	-	10.2
n	21	17	56	15	-	109
Other						
×	14.0	-	65.0	-	17.0	45,2
8. d.	-	-	/4.8	-	-	59.3
n	1	-	3	-	T	5
Total						
Ĩ	15.5	14.8	18.8	24.7	17 D	19.0
a. d.	7.4	9.2	19.9	15.2	~	16.1
n	22	17	59	15	1	114
Cortex						
Cryptocrystalline						
X	-	15,0	-	-	~	15.0
s. d.	-	-	-	-	~	-
n	-	1	-	-	-	1
Total						
x	-	15.0	-	-	-	15.0
s.d.	-		-	-	-	-
n	-	1	-	-		1 1

.

Table 3-13. Average length (in mm) of conchoidally flaked material by zone, 45-D0-282¹.

 1 <1/4 in flakes, non-lithics, and non-conchoidal flakes delated.

No b o ol o l			Zone			
	1	5	3	4	25	IOTAL
No cortex						
Cryptocrystalline						
x	10.8	10.4	10.2	10.8	16.9	11.3
	274	433	754	433	285	1 950
Other	£7 4	-00	704	100	200	1,000
x	8.8	12.1	15.0	13.0	31.9	15.1
s.d.	3.5	8.1	10.0	8.4	18.6	11.7
n	9	16	26	4	7	61
Total						
x	10.7	10.5	10.4	10.9	17.3	11.5
s.d.	5.2	5.4	5.4	5.1	9.1	6.5
n	283	449	779	137	272	1,920
Partial cortex						
Cryptocrystalline						ļ
×	15.1	15.2	15.0	18.8	-	15.5
s. d.	7.9	8.2	8.4	7.8	-	8.7
Other	C 1	10	48	12	-	30
Ĩ	18.0	-	19.3	_	25.0	19.8
s.d.	-	-	4.9	-	-	4.8
n	1	-	3	-	1	5
Total						
x	15.1	15.2	15.2	18.8	25.0	15.7
s. d.	7.7	8.2	9.2	7.8	_	8.5
n	22	16	52	12	1	103
Cortex						
Cryptocrystalline						
X.	-	29.0	-	-	-	29.0
s.d.	-	-	-	-	-	-
n	-	٦	.=	-	-	1
Totel_						ľ
, X	-	29.0		-	-	29.0
.	-		-	-	-	-
n	***	r	-	-	-	τļ

÷.,

.1

の一般になったのないのないのでは、そのないないで、ためのないで、ためのないで、「ないないない」ではないないでは、「ないないない」では、「ないないない」では、「ないないない」では、「ないないない」では、

Table 3-14. Average width (in mm) of conchoidally flaked material by zone, 45-DC-282¹.

1<1/4 in flakes, non-lithics, and non-conchoidal flakes deleted.

	1		Zone			1
Hatarial	11	12	13	14	25	Total
No Cortex Crystocrystal 1 m						
x	20.3	18.9	17.9	18.3	36.5	23.0
s.d.	13.5	12.6	11.7	10.4	25.2	18.0
Other	4.0					01200
x .	17.3	24.0	30.8	28.0	55.8	32.8
N , C,	12	20	33.2	20.3	3≪.8 20	- 21.0 90
IOTEL	20.2	19.1	18.2	18.8	37.0	23.3
s.d.	13.4	12.8	12.9	10.8	25.5	18.5
n	428	682	1,174	209	803	3,296
Partial Cortex						
Cryptocrystalline		07.0		40.0		
x s.d.	20.5	13.7	23.9	42.0	-	21.3
n	35	31	75	22	-	163
uther T	30.0	-	86.0	-	41.0	69.2
s.d.	-	-	81.4	-	-	68.3
n	1	-	4	-	1	6
Total		~ ~				
x a.d.	32,9	27.8	38,4 30.3	42.0	41.0	35.7
n	36	31	79	22	1	169
C						
Cryptocrystalline						
X	-	30.0	-	-	-	30.0
#,4. a	-	-1	-	-	-	-
		-				
iatel X	-	30.0	-	-	-	30.0
s, d,	-		-	-	-	
n		1	-	-	•••	1

Table 3-15. Average thickness (in 0.1 mm) of conchoidally flaked material by zone, 45-00-282¹.

1<1/4 in flakes, non-lithics, and non-conchoidal flakes .stated.

INDUSTRIES

There are at least three recognizable stone tool industries at 45-00-282. All three primarily utilized jasper and chalcedony transported to the site as nodules, cores or blanks. The pervasive industry is a generalized flake tool technology, represented by cores, flakes, finished tools and a vast amount of chipping detritus. Heat treatment appears to have been common. From the frequency of objects with complete or partial cortex, we can conclude that primary reduction also was commonly practiced, indicating that jasper and chalcedony often were transported to the site as weathered nodules. The number of tools present and the numerous fine finishing flakes show considerable investment of effort, and the importance of this generalized reductive technique in the manufacture of most tool forms at the site (Table 3-16). We cannot describe actual steps in this reduction sequence nor the fundamental characteristics necessary to delineate the nature of reduction (e.g., hard hammer versus soft hammer percussors, the angle of flake detachment, core and platform preparation, etc.). We can state, however, that the lamellar flake was the most common tool form produced; that flake dimensions are remarkably consistent over time, either representing consistent knapping techniques and an idealized product or uniform core sizes; and that, over the period of site occupation, this basic reductive technique apparently did not change.

A second tool industry is also in evidence--a Levallois-like method of blade production comparable to that described by Leonhardy and Muto (1972) and Muto (1976). Evidence consists solely of large blades and tools made on blades. No cores were recovered, nor does this analysis recognize the characteristics of manufacture detailed by Muto (1976). We may assume that some core preparation and attendant blade production went on at the site, but we cannot assess its prevalence, nor its relationship to the more generalized flake tool technology. The two large blades recovered appear to be what Muto (1976) has termed corner-removed blades. Evidence preserved in projectile point configuration suggests that these and so-called "A-blades" were commoniy used in tool manufacture.

The third tool industry is better described, if only because its products are numerous. This is a microblade industry, which entails the detachment of small, parallel-sided blades from carefully prepared, tiny wedge-shaped cores. Represented by 13 cryptocrystalline cores and 173 microblades, its appears to have been a common form of stone tool production throughout the span of occupation at 45-D0-282. Plate 3-1 shows microblade cores recovered from 45-D0-282; Plate 3-2 shows the microblades thembelves.

Morphological attributes and descriptive terminology for microblades and cores are illustrated in Figure 3-1. The production of microblades requires quite different core preparation than that involved in the production of conchoidal flakes. Striking platforms must be broad and flat, with angular margins that approach a 90 degree or sub-90 degree angle to the striking platform. This results in a plane of detachment for blades that carries from the point of impact well down toward the ventral midline of the core. Blades may be detached by percussion or pressure flaking. The focused force will remove a long narrow flake that feathers out as the force carries across the core's lateral surface or terminates abruptly at some surface irregularity. This reductive process is more controlled and intricate than that required for the simple detachment of lamellar flakes. However, the two techniques may not be exclusive, since cores or chunks that are products of the one process can be readily adapted for use in the other.

Table 3-17 describes microblades and microblades cores. Table 3-18 lists measurements of microblades. Table 3-19 describes microblade attributes by analytic zone. Most microblades have prismatic cross sections (two arrises on the dorsal surface) although many have a triangular cross section (a single arris). Gnly 41 specimens do not terminate in a snap fracture. A few have been snapped scross both the dorsal and proximal ends. Except for these few

Table 3-16. Material by object type¹, functional type, dorsal topography, and zone, 45-D0-282.

Meterial	Object type	Finctional tuna	International Internation			Zane			
				-	8	3	•	25	10101
Cryptocrystalline	Conchafdat ftaka	Prolectile		•	-	l	1		•
		point tip			-)	I	1	-
		Biface	None	-	-	3	-	••	2
		Burin	None	-	0	-	0	• •	. 40
		Drill	None	ı	•	0	-	63	-
			Indeterminate	1	-	•	• •	• •	
		Gravar	Nane	ı	-	a	١	١	. 64
		Bcreper	None	-	1	0	-	4	
		Burin spell	None	ł	••	CN	1	1	3
		Core	None	•	-	-	ł	ı	0
			Partial cortex	1	1	-	ı	ı) 4-
		Resharpening flake	None	~	••	•	01	Ø	18
		Fragmant from a	None	-	• 1	13	1	• •	•
		blade core							•
		Bifacially retouched flake	Nane	6	01	69	01	~	କ୍ଷ
			Partial cortex	1	-	i	ı	•	*
		Unifacially ratouched flake	None	•	•	12	•	•	38
			Partial cortax	1	•	•	1	1	Ŧ
			Indeterminate	1	ł	• 1	•	ı	•
		Utilization only	None	54	78	128	30	144	ABA
		•	Partiel cortex	-	04		-	•	8
			Indeterminate	-	+	••	1	ı	
		Indeterminate	None	-	ŀ	1	ı	ı	•
		None	None	574	906	1,669	286	916	1351
			Partial cortax	4	4	116	8	•	225
			Complete cortex	۱	-	-	ı	·	0
			Indeterrinate	1,633	2,029	3,438	053	6	7,900
	Microblede	Microbi ade	None	28	30	53	4	•	162
			Partial cortax	ł	-	ı	1	•	-
		:	Complete cortex	1	ı	۱	,	•	۴
-		None	Indeterminate	10	15	35	10	1	70

41

盛

.

Table 3-16. Cont'd.

11111

	_		_		-	-			_	_	-																_			-		_	_	_	_			_					-
	10101	•••		- (30		•		10		=	Q	150	18	225	4	-	-	~	4	-	2		-	u)	a	10	91	2	69	-	ន	-	CM	23		•	-	=	(3	-	
	25	C ø .	ł	1 •	-	•	-		4		4	C 4	20	-	ю	••	-	t	-	ı	,	-		ı	ı	Ð	Q	2	•	34	1	10	1	١	-0		ı	-	19	•	•••	ı	1
	•	1	1	I	ı	ı	•		1		-	ı	13	ı	24	ı	•	-	•	-	-	ŀ		ł	ı	-	-	÷	•	4	ı	I	ł	ı	•••		ł	;	•	•		-	•
Zone	e	m	1	1	1 -	-	ł		-	I	n	ı	84	ç	63	-	ı	ı	-	Q	ı	-		ı	C 4	-	-	a	•	14	-	ŝ	•	ł	8	•	,	ı	ı		ı	ı	ı
	a		- •	-	ŀ	•	m)		C 1		ı	ł	38	C ¥	63	ł	1	ł	-	-	ı	ı		ł	Q I	-	-	e	•	12	ł	m	ı	-	8		ı	ı	ni		•	1 •	-
	-	Cu	1	. •	-	1	-		m		•••	ı	ន	'n	4	1	i	ł	4	1	ı	ı		-	-	1	ł	0	4	4	ı	C 4	ı	-	4		-	ł	ł	•	-	ł	;
	reet topography	eu	de terminete			rtial cortex	2		2		ana	ida termi nete	92	irtiel cortex	ide term ine te	90	Itial cortex	•••		rtial cortax	ide term i ne te	•u•			ide termi nate	B u		2		Put of the second se	irtiel cortex	ene	•		• •		irtial cortax	ide tarrai ne te			2		de l'entit ne te
	3	Ŷ,	5:	2:	2	.	Ŷ		2		o No	5	No.	4	5	No.	đ	¥	No.	ď	ų	Ŷ		Ŷ	Ľ	Ŷ	No.	, A	2	No	4	Ŷ	£	No.	£		2	5	Ŵ	;	₽:	2 ·	u,
		Biface		screper	0.00		Bifacially	retouched flake	Unifacially	retouched fleke	Utilization only	•	None			Biface		Drill	Core			Unifacially	retouched flake	utilization only	Nane	Projectile point	Projectile point	Deciertie octat	tip	Biface	Chopper	Screper	Core	Rechargening flake	Bifaciaily	retouched flake			Unifacially	retouched fisks	Utilization only	None	
	ed in the	Chunk														Core										Formed object																	
		Cryptocrystalline																																									

42

È.

Table 3-16. Cont'd.

	Total	•	• 6		- 1	0	 -	-		-	-	142	.	•	- 1	- 6		~	-	-	•	-		a i	-	1	-	7	0	•	•		-	-			-	1	3		[•	-	•	-	•••	-	
	25	•	- 1		•	-	1	•	•	**	-		ø	• •	- 4	- (•		-	-	*	•	I	01	•••	•	-	•	9	-	•		ı	1		ı	ı	8	•	1	. 1	•	i	•	ł	•	
	•		•	1	1	1	•	ı		•	ł	•	• •	;	1	1	•	1	1	•	•	ı		ı	•	•	•	9	1	1	1		•	1		ı	i	•	•	đ	• 1	1	. 1	ı	ı	1	
Zone	8	-	- 1	Ŧ	- 0	4	•	ı		•	4	•	• 1	1	ı	4		•	ı	ı	ł	-		1	ł	•••	ł	~	ı	ı	ı	•	-	ł		•••	٠	20	10	38	} 1	1	I	i •			
	2	•	ı	1	1	1	ł	ł		1	•	-	• 1	•	ı	I	. 1	1	•	۱	ı	ı		•	۱	Q	1	۵	1	ł	•		4	1		ı	•••	14	-		: I	i	Ŧ	-	1	i	
	-	•	a		C	4 4	-	ı		•	**	0	1	ı	•	1	i 1		1	1	1	•		1	1	ı	1	C4	ł	1	1	i		-		ŧ	1	ø	04	13	•	• 1	ı	I	1	I I	
	uoraal topography	Indeterni nete	Indeterminate	Partial cortax				Nora	More		Nore	Indeterninete	None	Partial cortax	None	Partial contac	Partial Cortar		Tartial Cortex	Indeterminate	Partial cortex	Complete cortex	1	Buon .		None	Pertiel cortex	Indeterminate	None	Partial cortex	Partial cortax	Bartist sector	rerblet vortex	None		Partiel cortex	None	None	Partial cortex	Indeterminate	None	Indeterminate	Mone		Bastisi second		
	runctional type	Indeterminate	None	Hunsteine	Microhiada	Basharosofing fishe		1111281100 001Å	Tabular baile		None		Tabular knife		Nane					Nord	Chopper	Kenserstone	lintfactal tu	retricted flate	UCILIZACION ONLY	None					Chopper			Unit act at Ly	retouched flake		Utilizetion only	None			Microbiade	None	Core	More			
		Westhered		Unmodified		Indeterainete			Conchoidai flaka				Tebul ar				Chunk	Formed object					Conchoidal flake						Chunk		Indeterminate	Conchoidel fieke									Microblade		Chunk			Gare	
Material		Cryptocry stalling							Quertz i te														Fina-orainad	quartzita								Besel t															

Material	Chiect tune	Functional type	Doced toooceahy			Zone		1	Total
	adda ana fan		urrest tupography	-	8	e	•	25	
Beeel t	Formed object	Biface	Nane	ı	-	ł	ı	ı	-
		i	Partial cortex	ł	1	-	ı	I ;	-
		Chopper	Partial cortex	1	1	ni -	1	-	<u>n</u>
			Indetersinate	,	ı	u 1	1	-	De (
		Hummerstone	Indeterminete		1	-	1	1	.
		Resharpening flake	None	4	ı	1	1	1	**
		bifacially	None	•	-	ı	1	ł	-
		retouched flake							
	Westhered	None	Indeterminate	I	-	1	1	,	•-
	limend if tad	Choner	Partial cortax	ł	• 1	•	•	•	• 64
		Partie	Complete Contex	1	•	ł	ı	• •	•
				I	•	,	1	•	• •
				ı	ı	•	1 4	- (- •
			Complete cortex	•	•	-	-	M	•
		Utilization only	Indeterminete	•	1	00	ı	1	Cu -
	Indeterminate	Chopper	Partial cortex	1	1	•	ł	4	4
		Utilization only	Pertial cortex	ı	-	1	•	,	-
		None	Indeterminate	ı	1	-	•	ı	•••
Granteic	Conchoidal flake		Mine	4	1	Ŧ	I	,	•
			Bestel costev	1	ł	- •	1	1	- •
				1	•	-	1	Ι.	
				I	-	1	1		- •
		i	Partial Cortax	1	ł	1	•	-	-
	Formed object	Chopper	Partial cortex	1	1	,	ı	•••	-
	Unmodi fied	Pastle	Compliate cortex	ı	•	ł	•	ı	-
		Hamaratone	Partial cortax	ı	ı	1	1	0	C 1
			Complete cortex	1	ł	••	•	•	ю
	Indeterminate		Partial cortex	ı	ı	ı	ı	-	-
		Indeterminate	Partiel cortex	1	ı	ı	ŀ	•••	-
an the state of the second sec	Conchoddel flate	litilitation and v			,	1	Ŧ	I	•
				,	•	•	• •	1	- d
			Tride to read to to	~	, c	łe	- 4		•
	Microbi ada	Microbi ada	Nora	. 1	; .	••	• •	1	-
	Chunk	Nana	Non	ł	• 1	1	ı	-	•
								•	•
Other Lithice	Conchoidel flake	Utilization only	None	ı	1	-	•	, 1	-
		Nane	None	ŝ	Ð	13	n	ରା	2 8 3
			Indeterminate	~	9	18	10	1	4
	Chunk		Pertial cortex	1	ł	•	1	••	-
			Inde term i ne te	13	47	20	53	•	173
	Core	Core	Partial cortex	ı	ı	ı	,	-	•
	Formed object	Projectile point	Indetersinate	ı	1	ı	•	ał	~
		Biface	Nane	•	ı	ł	1	1	-
		Chopper	Partial cortex	•	·	ł	•	•	-
								•	

CARE THE CARE AND A CONTRACT OF A C

Table 3-16. Contid.

44

Experience Boursers

3

1.1 4.1

Table 3-16. Contid.

ka tantal		Functional type				Zone			Total
				•	2	8	•	25	10161
Indeterminate Lithica	Concholdel flake	Indeterminete	Inde termi nete	-	1	ŧ	•	1	Ŧ
	ant S	None	None Toda to and page	- 1	1 4	** *	1	a, i	~ c
	Indetersinate	Utilizetion only None	Inde termine te Inde termine te	i i -		- 1 1	1 1 04	1 423	4- <u>6</u>
Totel				.482	3,468	5,896	1.301	1,368	14,603
1/4/4 40 81 abov -									









Figure 3-1. Microblade core and microblade terms (from Sanger 1986:95, Figure 2).

も国際がなかれ、より展開によったという意味であるとなるが言語になったのです。

キャイトのというの範疇であるとなるともので

45-00-282.
cores,
l ade
microb
<u>0</u>
Attributes
3-17.
Φ
Tabl

and a second and the second and the

Mater number	Zone	Matarial	Platfors Length [mm]	Platfors width [mm]	Core height [Core adge angle [degrass]	Number of flutes	Neen width of flutes [=-)	Nean Length of flutes [mm]	Number of striking directions
11/1160	8	Jasper	4	22	8	87	80	8.7	12.2	C4
11/1152	22	Chal cadony	58	15	17	63	~	5.3	12.6	•••
11/314	25	Chalcadony	ಜ	12	15	82	8	6.2 9	12.4	•
11/115	25	- and a labor	56	13	8	67,81	60	7.0	17.0	•
11/651	-	Jacob	₽	80	17	78	ß	3.2	14.5	-
11/1055	- 04	Chalcadom	17		2	54	ß	3.7	8.7	-
11/900	0	Chal de cony	27	ຂ	30	114,64	12	6.3	17.7	•9
11/861	3	Chel cedom	20	:	ce	73	12	3.6	0.8	-
11/418	3	Chalcadony	58	13	24	62	4	8.8 2	ດ. ສ	-
11/806	3	Chal de dom	27	8	54	63,73,80	12	6.6	15.0	63
11/781	0	Chal cedony	80 02	17	ន	78	~	5.5	13.0	0
11/964	g	Chal cedomy	17	1	54	77	'n	7.6	22.5	-
1/475	ß	Chal cadony	30	8	28	78	10	7.5	14.5	-
Rance			10-42	8-98	8-30	54-114	4-12	1-12	5-28	1-3
			24.1	15.1	21.0	74	7.8	5.4	13.7	ł
Standard			7.8	8°2	6.7	14.4	5.9	6.3 6	4.7	1
deviation	5									

TATATAN KALATAN PARATAN PARATAN PARATAN PALATAN PARATAN PARATAN PARATAN PARATAN PARATAN PAR

specimens that exhibit both proximal and distal fractures, it is virtually impossible to classify breakage as intentional, or as an accidental product of manufacture. Many of the core flutes terminate abruptly in hinged fractures, and one may conclude that a large number of microblades with either end snapped off are simply products of manufacture.

Managemente	ALL	microbled	es N=173	Complet	a microbl	ades N=41
ndi sur di di Co	Length	Width	Thickness	Longth	Width	Thickness
	(mm)	(mm)	(wm)	(mm)	[mm]	(mm)
Range	6-34	3-12	0.5-2.0	9-34	4-9	0.5-2.0
Tan	13.2	5.8	1.1	16.2	6.1	1.1
S.D.	5.1	1.5	0.4	6.2	1.2	0.4

Table 3-18. Microblade measurements, 45-D0-282.

Table 3-19. Microblade attributes by zone, 45-D0-282.

	Zone					Tabal	
	1	5	3	4	25	IDENE	
Prismatic cross mection	17	22	41	13	7	100	
Distal end snap	8	10	22	7	3	5U	
Proximal and snap	2	3	5	1		11	
Distal-proximal and snap	3	4	5	2	3	17	
Complete	4	5	9	З	1	22	
Triangular cross section	13	12	35	9	4	73	
Distal and snap	8	4	17	4	2	33	
Proximal and anap	3	2	3	2		10	
Distal-proximal and snap	1	2	7	1	-	11	
Complete	3	4	8	2	2	19	
Total	30	34	78	22	11	173	

のないない。このないないでは、「ないないない」ではないないない。「ないないない」ではないないで、「「ないないない」」ではないないない。「ないないないない」ではないないない。「ないないない」では、「ないないない」

As shown in Table 3-17, the sizes and platform edge angles of cores are very consistent, regardless of associated analytic zone. Measurements of flutes or blade scars are also fairly consistent. This consistency is further demonstrated if one compares Tables 3-18 and 3-19. Mean lengths and widths of blades and blade scars are very close, indicating a fairly well controlled industry. One may infer with some confidence that knappers were attempting to produce blades of uniform proportions.

Figure 3-2 depicts the consistency observed in microblade dimensions. The upper graph shows the relationship of width and length; width shows a very narrow distribution (4-6 mm), regardless of length. The bottom graph depicts the lengths of specimens as a frequency distribution. Length, it will be noted, shows much greater variability than width. The control of width would seem to have been a cultural variable, related to selection of punch or percussor size and sense of ideal microblade size. Length, on the other hand, seemed to depend on the size and quality of the nodules being used as cores.







Dimensions recorded for both microblade cores and microblades in this collection are very similar to those recorded by Sanger (1968, 1970) and Munsell (1968) for microblades on the Columbia Plateau and by Taylor (1962) for microblades in the American Arctic. Blade widths average 5.8 mm; lengths average 13.2 mm. This consistency in blade size across the Northwest and into the Arctic has led Sanger (1968, 1970) to postulate a "Plateau Microblade Tradition" and to speculate that there are direct historical ties to microblade traditions to the north.

While this microblade industry occurred in the context of a more generalized flake tool industry which was also associated with a Levalloislike blade tool industry, and all three industries were based on the reduction of jasper and chalcedony, it required more careful, controlled techniques of tool production. All three industries at 45-D0-282 are distinct, but, as shown by their association here and elsewhere on the Columbia Plateau, they were complementary facets of the same general stone tool technology.

TEMPORAL AND SPATIAL DISTRIBUTION

2月に、このために、1955のから、1955のためになったので、1955のために、1955ので、1955ので、1955のために、1955のために、1955のために、1955のために、1955のために、1955のために、195

Stone tool manufacture, primarily as part of a generalized flake tool industry utilizing imported cryptocrystalline stones, was remarkably consistent over the span of occupation at 45-D0-282.

Together, the three stone tool industries mark the 45-D0-282 artifact assemblage as Cascade-like, probably dating to the latter part of the defined Cascade Phase (cf., Leonhardy and Rice 1970; Bense 1972). The presence of Levallois-like blades, with a generalized flake tool industry, and a microblade industry are characteristic of these early assemblages (Leonhardy and Muto 1972; Muto 1976; Sanger 1968, 1970; Munsell 1968). The closest correlate is probably the assemblage recovered from the Ryegrass Coulee Site (Munsell 1968), where radiocarbon assay and diagnostic artifact types supply a probable time frame of ca. 6500-3500 B.P. or the late Cascade Phase. To the north, the earliest dated site with microblades is the Drynotch Slide (ca. 7500 B.P.) (Sanger 1968). Microblades and microblade cores recovered from 45-D0-282 appear representative of the defined "Plateau Microblade Tradition."

The technological analysis presented in this report is too cursory to describe adequately the full range of variation in these separate manufacturing techniques, or to delineate fully the relationship between the three. We can safely state that they are present, but we cannot adequately describe them nor place them in proper regional perspective, without a much more detailed, careful analysis of the materials and the varied halimarks of reductive strategy present in the collection.

FUNCTIONAL ANALYSIS

Functional analysis examines the physical characteristics of artifacts in order to identify patterns of wear diagnostic of specific tool uses. Past research has pointed out the possibility of interpreting tool use by examining edge damage and general attrition of working surfaces (e.g., Hayden 1979; Statford and Stafford 1979; Keeley 1978, 1974; Odell 1977; Crabtree 1973; Wilmsen 1968, 1970; Frison 1968; Semenov 1964). Wear patterns have been shown to reveal both the manner of tool use and the nature of the materials worked.

All artifacts were examined with a 10X hand-lens (cf. Hayden 1979; Stafford and Stafford 1979). During analysis, each artifact was classified as to tool shape, wear or surface damage, and edge angle. Making use of established correlations between specific wear patterns on certain materials and types of tool use, we can hypothesize the intended and actual use of collected tools. Most distinctions will be based on hardness--on the nature of edge attrition given softer and harder working mediums.

The surfaces of many of the lithics from 45-D0-282 had a thin deposit of an unidentified substance, which tended to be concentrated on the edges. Attempts were made to remove this substance with a variety of acids, bases, other solvents, and ultrasound, but to no avail. The deposit may have obscured light chipping wear in approximately 10% of the cases.

Eight classificatory dimensions are used to describe functional attributes: UTILIZATION-MODIFICATION, CONDITION OF WEAR, WEAR/MANUFACTURE RELATIONSHIP, KIND OF WEAR, LOCATION OF WEAR, SHAPE OF WORN AREA, ORIENTATION OF WEAR, and EDGE ANGLE. The first dimensions describes objects, the next six describe tools on objects, and the last describes variation within object/tool types through measurement of the working edges. Table 3-20 outlines these dimensions and constituent attributes.

Description will initially focus on functional object types. Objectspecific dimensions will be used to introduce the occurrences of wear on functional object types. Tool-specific dimensions will outline the relationship of wear to manufacture and explicate the kinds of wear observed. Analysis will therefore proceed from the object to examination of tools on the object. Summary tables will deal with tools and the attributes of wear and manufacture which characterize them, rather than with simple descriptions of traditional formal-functional categories.

As in the preceding section on Technological Analysis, all discussion will focus on the distribution of functional types and tool types within the five defined analytic zones.

FUNCTIONAL OBJECT TYPES

A total of 1,110 stone tools was recovered from site 45-D0-282. These include a broad range of functional forms: light piercing and cutting tools, cruder, thicker cutting and scraping tools, and heavy chopping and pounding implements. Chipping was the only type of manufacture recorded (Table 3-21). Simple utilized flakes are by far the most frequent tool form (48\$, N=539). Other tools showing wear only include burins, choppers, drills, gravers, pestles, scrapers, hammers, microblades and cores (4\$, N=49). Tools with manufacture only comprise 14\$ of the assemblage (N=156); they include projectile points, bifaces, choppers, scrapers, resharpened flakes, bifacially retouched flakes and unifacially retouched flakes. Tools with wear and manufacture constitute another 16\$ (N=174); they include projectile points, bifaces, choppers, scrapers, tabular knives, hammers, cores, resharpened flakes, bifacially retouched flakes and unifacially retouched Table 3-20. Functional dimensions. 45-D0-282.

DIMENSION I: UTILIZATION/MODIFICATION	DIMENSION VI: Continued
None	Feathered chipping
Wear only	Feathered chipping/abrasion
Manufacture only	Feathered chipping/smoothing
Manufacture and weer	Fee thered chipping/crushing
Modified/indeterminete	Feathered chipping/polishing
Indeterminate	Hinged chipping
	Hinged chipping/abresion
DIMENSION II: TYPE OF MANUFACTURE	Hinged chipping/moothing
	Hinged chipping/crushing
None	Hinged chipping/polishing
Chipping	None
Packing	
Grinding	DIMENSION VII: LOCATION OF WEAR
Chipping and packing	-
Chipping and grinding	Edge only
Pecking and grinding	Unifacial edge
Chipping, packing, grinding	Bifacial adge
Indetensingts/not applicable	Point only
	Point and Unitacial edge
DIMENSION III: NANUFACTURE DISPOSITION	Point and Difacial adge
	Point and any compination
None	
Pertial	None
iotel Tada harmian ha fant, anni ionti a	
Tuge celearue cel uper abbit scant e	DIMENSION VITTE SHAPE OF WORN AREA
DIMONSTON THE WEAR CONDITION	
DIMENSION INT WEAK CONDITION	Not applicable
None	Convex
Deenlate	Conceve
Frament	Streight
	Point
DIMENSION V: WEAR/HANUFACTURE	Notch
REATIONSHIP	Slightly corvex
	Slightly conceve
None	Irregular
Independent	
Overlapping - total	DIMENSION IX: ORIENTATION OF WEAR
Overlapping - partial	
Independent - opposite	Not applicable
Indetermineta/not applicable	Pereilel
	Obt ique
DIMENSION VI: KIND OF WEAR	Perpendicular
	Diffuse
Abrasion/grinding	Indeterminete
Smoothing	
Crushing/pecking	UIMENSION XI OBJECT EUGE ANGLE
Polishing	And unit and an analysis
	ACTUBI BOGB ANGLE

0bj ect	Utilization/	Type of 2	Zone					
type modification' manufacture		MERUTECTURE	1	5	3	4	25	
Projectile point	3	2	-	1	-	1	4	
Projectile point	-	E	_		•	-		
bese Projectije point	3	2	-	1	1	1	5	
tip	3	2	1	5	2	1	3	
	4	2	1	5	-	-	4	
BITACE	3 4	2	4	10	15	4	25	
Burin	2	1	1	ž	í	2		
Chopper	2	1	-	-	1	-	-	
	3	2	-	-	4	-	15	
Devil	4	2	-	-		-	10	
UFILL	4	2	-	1	<u>د</u>	2	1	
Graver	2	1	-	1	1	-	- 1	
	4	2	-	-	1	-	-	
Pestle	2	1	-	2		-	-	
scraper	3	2	_	_	1	-	1	
	Ă.	2	3	4	Á.	-	11	
Tabular knife	4	2	-	-	-	-	5	
Homerstone	2	1	-	-	7	2	5	
Burin Spail	1	1	-	1	2	-		
Hicrobi ade	1	1	30	31	73	24	6 j	
_	2	1	2	1	2	-	2	
Core	1	1	4	3	7	2	3	
	2 A	1	-	1	-	-	2)	
Resharpening flake	2	ī	Å	-	-	-	1	
	3	2	3	-	5	1	1	
	. 4	5	3	2	1	1	3	
CORE	1	1	1	-	-	-	_	
Bifacially	•	•	•					
retouched flake	3	2	5	9	8	4	4	
lind front of the	4	2	7	6	8	-	6	
Unit&Clatty retouched flake	3	2		1	ß	1	A	
	4	ž	6	12	10	Å	13	
Utilized flake	2	1	104	85	145	46	159	
Indeterminate	4 5	2 9	2	-	1	-	2	
Total	<u></u>		176	152	272	85	217	

Table 3-21. Utilization/modification and type of manufacture of formed lithic objects by zone, 45-D0-282.

¹Utilization/modification

1. 2. 3. 4. 5. 6.

• •

- None Wear only Menufacture only Menufacture and wear Modified/indetaminate Indeterminate

2Type of Henufacture 1. None 2. Chipping

- 1. 2. 3. 4. 5. 8. 7. 8. 9.

- Chipping Packing Chipping and packing Chipping and grinding Packing and grinding Chipping, packing, grinding Not applicable/indeterminate

flakes. Together, these tool forms and associated attributes of wear and manufacture show a broad range of functions and an emphasis on tool production and tool maintenance at the site. Plates 3-3 through 3-5 illustrate flake cores and bifaces; projectile points and drills, burins, blades, and gravers. Plate 3-6 illustrates large cobble tools, and two pestles are shown in Plate 3-7.

WEAR PATTERNS

Many of the 1,110 stone objects exhibit more than one instance of wear or more than one tool (25%, N=276) (Table 3-22). The highest wear area-object ratios were observed on scrapers, gravers, drills and hammerstones. Ratios for tabular knives, unifacially retouched flakes, pestles, and utilized flakes are only slightly lower. Cores, choppers, burins, resharpened flakes, and microblades have the lowest ratios. Object forms with the largest range of defined wear areas include scrapers, unifacially retouched flakes, and simple utilized flakes, with from 0-7 isolable tools. Those forms with the narrowest range are cores, choppers, burins, tabular knives, and resharpened flakes, with 0-2 wear areas present. We conclude that although simple utilized flakes were the most frequent form, and were intensively used, other object types such as scrapers, gravers, drills, and hammerstones saw more consistent use and reuse. We may also conclude that any given object may not be accurately categorized under a single functional label, as it may have multiple uses and variable potential functions.

Most tools in this collection are some combination of feathered and hinged chipping wear on a unifacial edge (79%, N=920) (Table 3-23). Other wear is predominantly smoothing on edges only and unifacial and bifacial edges (6%, N=65) or crushing of a surface (2%, N=28). Any sort of wear on points is relatively uncommon, as is crushing of unifacial or bifacial edges. In general, it seems that heavy chopping or pounding activities are represented, but do not account for a large proportion of tool types recovered, These indicate intensive cutting and scraping activities in soft, pliable materials like hides, meat, or, perhaps, plant or woody materials.

Figure 3-3 illustrates the relationship of wear types to defined functional object types. Most obvious is the rough correspondence between functional types with implicitly assumed uses and wear types indicative of those kinds of uses. Choppers and hammerstones are characterized by heavy crushing wear on edges and surfaces, indicative of work on hard materials, either bone or stone. Smaller flaked tool forms are characterized by feathered and hinged chipping wear on unifacial and bifacial edges and points. If we make finer distinctions, however, we discover discrepancies between implied and actual tool uses. For instance, projectile points show smoothing, feathered chipping and crushing wear on edges, reflecting use as general purpose cutting and scraping tools. Scrapers show predominantly hinged chipping wear on unifacial and bifacial edges. Indicative of heavy cutting or scraping uses. If these tools had, in fact, been commonly used to scrape hides or other soft materials, they would have exhibited smoothing or light, feathered chipping wear. Drills and gravers, tool forms believed to have been

÷,

2 7

 \mathcal{L}

.



スススストウス

Plate 3-3. Flake cores and bifaces, 45-D0-282.
		Mas Hor KEY His Pro Zon Mat	ter number: phological ty torical type venienca/lev B: arial:	ypei I BLI			
a. 57 Type 2 Unacsigned 103 N231E/1 5 Jasper	b. 471 Type 3 Cold Springs 63N200E/110 4 Opel	c. 89 Type 3 Obid Springe 105N233E/1 5 Argillite	d 301 Type 3 Cold Sprim 122K230E/1 5 Jesper	9. 1048 Type 3 39 Cold Spr 75S174E/ 4 Jesper	tings 110		
f. 106 Type 5 Wndust A 105NB6E/1 5 Jasper	9. 850 Yype 5 Ceatade A 8453AE/Fe13/110 3 Jasper	h, 884 Type 5 Cescade A 84535E/Fe1 3 Jasper	1. 1121 Type Casc 5∕110 545 3 Chelo	5 ide A 5E/Fe14/100 xedony	j. 216 Type 5 Cascade A 115M245E/0-40 - Jasper	k, 258 Type 5 Cascade A 117 N232E/1 5 Chel cadony	L. 522 Type 5 Hehkin Shouldered 5N5W/70 2 Jesper
239 Type 6 Mahkin Shoulderad 115N216E/1 5 Jaspar	n 116 Type 6 Mahkin Shouldered 108N243E/1 5 Jasper	Q., 277 Type 6 Mahkin Shouldered 119N239E/1 5 Jesper	P. Type 7 Surface 5	q. 206 Type 7 Nespelam B 116N245E/0 Jasper	r. 1154 Type 7 ar Nespelem -40 Surface - Jesper	B. 3005 Type 10 Ber Columbia 122x1975 5 Chelcada	∎ 8 ⊊∕10 Iny

 $\mathbb{R}^{1}_{\mathcal{A}}$

the second and a straight of the second second and a second second second second second second second second s

E.

.

Plate 3-4. Projectile points, 45-D0-282.

オイヤン語語があるのからいでは認識になっていたが、問題になっていたが、自然自己になっていたか。そうという



COMPLET

1.1.1

-72

4

a de la comencia de l

.

	KI I	Master number: Tool: EY Proventence: Zone: Material:		
6. 953	b. 805	с. 49	4 . 103	6. 25
Drill	Drill	Utilized flake	Drill	Dritl
645165E/20	5455 E/70	1031232E/1	106h213E/1	102N230E/1
2	3	5	5	5
Chalcedony	Ch al cedony	Jasper	Jaapar	Jaspar
f.	9-	N .	1.	ا.
477	95 4	431	1141	DL 38
Burtn	Burin	Burin	Blade	BLade
74H221E/0	645165E/20	83N231E/30	105H200E/1	13N6W/Beach collection
1	2	2	5	-
Chal cadeny	Ch al cado ny	Ch al. cedony	Jesper	Jesper
k.	L	n.		
926	744	708	219	
Graver	Burln	Graver	Chunk	
64514E/100	4595W/Fe7/90	345151/90	115H245E/0-40	
3	3	3	-	
un al cedo ny	Jaapar	Jasper	Ch et cedo ny	

Plate 3-5. Drills, burins, blades, and gravers, 45-D0-282.

のの言葉をいたないというです。そのできたが、「「「「」」というないでは言いないで、「」」というないで、「」」というないで、「」」というないで、「」」というないで、「」」というないで、「」」というないで、

J





ESSENT INVANA ENVERT BEEDRI DERENI DEREN



786/540 88/88 1.51 64/57 -1.12 0.67 -0.04 0/2 --0 --0 --1.80 116/82 62/30 -2.73 **\$**6. **R**tio Total Frequency 887 28 8 38/14 =2.71 241/160 30/21 -1.43 12/10 0.80 2/7 0.28 1.80 1.80 58/41 2 Re tio 8 Frequency Table 3-22. Functional object type and wear area/object ratio by zone, 45-D0-282. 84: fa tio 12/40 1.35 28. 2 0.50 0.24 5,0 1/5 28.3ª 1/8 1/8 25 4 Frequency ខ្លួយលេខ្រោ 🗣 i - i 111.1.4 227/145 20/16 1/3 =0.33 1/73 25/22 26/8 =3,25 18. 18/18 -1.19 Ratio Frequency Zone 822 1.8 8 8 1.12 28 \$8. * 22/13 -1.69 20.0 10.0 39 120/85 -1.41 Ratio 04 Fruguency 일 # 0.4 0 12/10 12/12 2/30 0.07 5 in 19 19 136/104 2 23 2% 28 **B** 10 -Frequency 25441 -Number of Weer areas Functional types Unifectally retouched flake Bifacially retouched flake Utilized fieles Tabuter knife Reshergent ng Fileke Micrabl ade Bergen 817360 Burta

Table 3-22. Contid.

			ļ			Zom						Tet	ľ
Punctional types	Rumber of	-		8		-		•		8			
		Frequency	Ruio	Frequency	Reto	Frequency	Na tio	Frequency	Re Li e	Frequency	Patio	Frequency	Mci o
Burin spail	•	ı			24	5		L		3		-	57
Ortic	- a a 4			.	5/2 -2.50		4°2 100 100	94 E F F	1/2 -0.50	₩ 4 4 1	6/3 *2.00		51.1
Cra er	-4	ı		-	21		2°5	ı		ł			58
Prejestite point	٥	-	~	~	2	•	1/4	•	\$	•	24/17	8	90/08
	- 01 19	~ 11	3.9	- 1	8. 9	-11	ୟ ନ		9	Q 42 ₹	1.4		-1.65
	07W					4 i 4	59 9			6 • «	99 87		15.95 10.47
Namer atoms						4 W I 4	1.1.	₩ i ₩ i	28.9 8		12/1-		1.75
Pet .	- 61	11			57 1-50	11		• •		• •		* *	1.50
ł	0 - -		2.0		1/4 .025	~ '	28	ee 1	3 9	n	1/4	5 .	1.14 1.14
Indecersi mu	0	Cu	21	، ۹.۹		-	29	ı		94	24	•	59

, ... -..--,

.

;

Crushed Fast thered Hinsed Same sheet FUNCTIONNE Type 16 0 54 1055 0 54 1005 0 50 1005 ۵ 50 Utilized flame adge anly unifected adge bifacted adge peint only e. Merilla II-788 Unif mit flake adge only unifectal adge bifectal edge peint only Madin A H=78 1.20 BIT ME FLAKE edge only unifectel edge bifectel edge peint enly **H=34** L rp flake adge only unifectal adge Difectal adge paint anly **#-15** Hicrobiede edge only unifectal adge bifectal edge paint anly **H=7** Stade edge only unifectal edge bifectal edge peint enly #12 Jacobergi He est Tobular knife edge only unifaciel edge bifaciel edge peint only -Screper edge only unifectel edge bifectel edge peint only H=73 Bifaca adge only unifectel edge bifectel adge paint only Burin adge only unifectal adge sifectal edge paint enly **\$**. Detti edge only unifectel edge Difectel edge paint enly **III-17** I Graver adge anly unifacial edge bifacial adge puint anly Projectile Peint odge only unifectal odge bifectal odge peint only Proj Pt TIp edge only unifectal edge bifectal edge paint only . 2 **m**11 edge only unifectel edge bifectel edge paint only i, odge only unifactel odge bifactel odge peint only surface Lorminel surface surface terminal Jurface odge anly unifacial odge bifacial odge peint enly **38-3**

Figure 3-3. Type and location of wear by functional object type, 45-D0-282.

used to perforate or incise relatively hard material like bone, do exhibit the expected heavy hinged chipping wear on points, but are characterized as well by feathered and hinged chipping wear on unifacial and bifacial edges. Some of the attrition recorded as wear may not be due to direct use, bit to preparation for hafting, or to resharpening Nonetheless, it appears that tool forms were used for purposes not necessarily defined by obvious morphological attributes of form nor by attached functional labels.

		1		Zone			Terrel
Kind of wear	Location of wear	1	2	3	4	25	10581
Shoothing	Edge only	1	2	-	-	10	13
	Unifacial adom	5	9	10	1	2	27
	Bifacial ados	1	3	7	1	43	25
	Point only	1	2	3	1	3	10
Crushing	Edge only	-	-	1	-	2	3
	Unifacial edge	-	-	-	-	1	1
	Bifacial edge		~	2	-	6	8
	Point only	-	1	-	-	A	5
	Surface	-	3	11	4	10	28
Feathered	Unifacial adom	105	85	162	~5	181	1.88
cnipping	Bifacial sdoe	11	8	18	1	28	68
	Point only	-	3	1	-	3	7
Hinged	Unifacial edge	50	58	79	21	124	332
chipping	Bifacial edge	4	8	13	4	12	42
	Point only	-	1	2	-	3	6
TOTAL		178	184	308	88	402	1,161

Table 3-23. Type and location of wear by zone, 45-D0-282.

Table 3-24 ranks functional types by the proportion of specimens within a functional type with a certain kind of wear and by the percentage of specimens within that functional type with that type of wear for the entire tool assemulage. A close correspondence in the order of the two rankings may suggest prehistoric selection for a specific tool form. A lack of correspondence may imply that use indicated by the type of wear did not require a specialized tool form.

Definitive characteristics are largely those noted in previous tables. Smoothing wear on edges only is characteristic of tabular knives. Smoothing wear on unifacial and bifacial edges is most common on projectile points, bifacially retouched flakes, choppers and bifaces. Smoothing on points only typifies gravers, and, to a much lesser extent, drills. Feathered chipping on unifacial and bifacial edges is most frequent on microblades, and utilized flakes. Feathered chipping on points is most frequent on microblades and utilized flakes. Hinged chipping on unifacial and bifacial edges is frequent on all flaked functional types except choppers. Hinged chipping on points only is most characteristic of gravers, drills and projectile points. Crushing on edges only and unifacial and bifacial edges is found most often on choppers. Crushing on points is not frequent in any functional type category, but comprises the highest percentage in cores. Crushing on surfaces and terminal surfaces, of course, characterizes hammerstones and pestles. Table 3-24. Ranking of functional tool types by wear type, 45-D0-282.

Weer type	Functional type r	enking	Functional type r	ankin
	X of assemblage a wear type	ithin	% of total assess	bl age
Smoothing			·····	
Edge only	Tabular knife	100.0	Tabular knife	- 8
	Bifacially		Bifacially	
	retouched flake	5.9	retouched flake	1
	Biface	1.7	Biface	.1
	Scraper	1.4	Scraper	.1
	UTILIZAG TLAKA	•1	Utilized flake	.1
Unifacial/	Projectile point	22.2	Utilized flake	1.9
DITACIAL AGGA	Biracially		Biface	.7
	Petouched flake	1/./	Bifacially	-
	Biface	14.3	retouched flake	.5
	Screper	6.8	ocreper Unifectulty	.4
	Resharpened fisks	6.7	Catouchad field	•
	Drill	5.9	Projectile point	•3
	Unifacially		Chopper	
	retouched flaks	5.1	Resharpened flake	.1
	Utilized flake	2.9	Drill	.1
Point only	Graver	33.3	Graver	.2
	Drill	11.8	Drill	.2
	Chopper	7.1	Utilized flake	.2
	Bifacially		Chopper	.1
	retouched flake	2.9	Bifacielly	
	BITBCS Undfootstin	1./	retouched flake	.1
	UNITECIELLY	4 3	B1Tace	.1
	Utilized flake	.2	retouched flake	.1
sethered chipping				
Unifacial/	Microbiade	85.7	Utilized flake	49.0
Bifacial edge	Utilized flake	72.4	Unifacially	
	BLede	50.0	retouched flake	2.2
	Projectile point		Biface	1.1
	tip	45.5	Bifacially	
	SUFIR Deck second distant	37.5	retouched flake	1.0
	Rifecially	33.3	Screper	.8
	retouched finks	32.4	Projectile sairt	.5
	Unifacially	JE	tin	
	retouched flake	32.1	Resharmened flave	.4
	Drill	29.4	Drill	
	Biface	22.4	Burin	.3
	Gravar	16.7	Blade	.1
	Scraper	12.3	Graver	.1
	Projectile point	11.1	Projectile point	,1
Paint only	Drill	17.6	Drill	.2
	Projectile point	11.1	Unifacially	• ·
	Unifacially		retouched flake	.2
	ratouchad flaka	2.8	Projectile point	4
			in of source boths	

1

وتعليشين المريبية والكرك كالكريات مالفا مارما مارما مالمكمس فالمابعات المانية والمرابع

Table 3-24. Conttd.

·

M	Functional type	ranking	Functional type r	enking
weer type	X of someblage wear type	#ithin	% of total assessi	bi ege
Hinged chipping				
Unifacial/	Scraper	78.1	Utilized flake	16.3
Bifacial edge	Core	6 8.7	Biface	3.0
	Burtn	62.5	Bifacially	
	Resharpened flak	60,8	retouched flake	1.3
	Biface	60.3	Resharpened flake	.8
	Unifacially		Burin	.4
	Tetoucheo flake	50 0	Projectile point	
	Projectile mint	50.0	Deciectile soint	.,
	tin	45.4	Drill	.3
	Projectile point	44.4	Screper	.2
	Bifacially		Core	.2
	retouched flake	44.1	Graver	.2
	Graver	33.4	Unifectally	
	Utilized flake	24.1	retouched flake	.1
	Drill	23.5	blade.	.1
	Hicrobiade	14.3	Ni crobi, ade	•1
Point only	Remar	48 4	Detti	2
rothe only	Deill	11.8	Graves	.1
	Projectile point	11.1	Projectile point	.1
	retouched flake	1.3	retouched fiske	-1
	Utilized flake	.1	Utilized flake	.1
Crushing				-
Edge only	Chonner	14.3	Chopper	.2
	Hammerstone	3.8	Hammerstone	.1
Unifacial/	Chopper	57.1	Chopper	.7
bifacial edge	Hannerstone	3.8	Hamerstone	.1
	Utilized flake	.1	Utilized flaka	.1
Point only	Core	33.3	Core	.1
	Projectile point	_	Projectile point	
	tip	9.0	tip	.1
	Unopper Norman	7.1	Chopper	•!
	namierstone	3.6		
	APILIZAD I FORM	•1	UFICIZED LENS	•1
Surface	Hamerstone	3.6	Hemerstone	.1
Terminel	Pestie	100.0	Hannerstone	2.1
surface	Hamerstone	85.7	Pastie	.2

ار وراین بر این از بر بر بر بر بر بر بر بر بر از از از مراوع بر و می واد و می واد و می واد و می واد و می ورد و ا

69

.

					Zo					
Proventing and the second		1	1	2		3		4		25
Functionet type	N	1	N	3	N	3	N	x	N	3
Flake tools										
Utilized only	104	58,7	85	45,9	145	46.6	46	46_9	160	50.1
Unifacially retouched	10	5.8	13	7.0	16	5.2	5	5.1	21	6.6
Bifacially retouched	12	6.5	15	8.1	16	5.2	4	4.1	10	3.1
Restarpered	-	-	2	1.1	3	1.0	2	2.0	5	1.8
Hicrobiade	30	15,9	32	17.3	73	23.5	24	24.5	7	2.2
Blede	2	1.1	-	-	-	-	-	-	-	
Tebular knife	-	-	-	-	-	-	-	-	5	1.6
Subtotel	158	89.3	1 47	79.4	523	87.6	81	82.6	208	85,2
Formed tools										
Biface	8	4.5	16	8.6	22	7.1	5	5.1	41	12.8
Scraper	ق	4.7	4	2.2	ô	2.5	1	1.0	14	4.4
Burth	1	0,6	3	1.6	1	0.3	2	2.0	-	-
Drill	-	-	2	1.1	2	0.6	5	2.0	3	0,9
Grever	-	-	1.	0,5	2	0.6	-	-	-	-
Projectile point	2	1.1	- 6	3.2	4	1.3	3	3.1	17	5.3
Chopper	-	-	-	-	5	1.6	-	-	25	7.8
Core	5	2.8	4	2.2	7	2.2	2	2.0	4	1.2
Subtotal	19	10.7	36	19.4	51	16.4	15	15.3	104	32.8
Unformed-pecked										
Hammerstone	-	-	-	-	7	2.2	2	2.0	7	2.2
Pestis	-	-	2	1.1	-	-	-	-	-	-
Subtotal	-	-	5	1.1	7	2.2	5	2.0	7	2.2
TOTAL	177		185		318		98		319	

Table 3-25. Distribution of functional types by zone, 45-D0-282.

Functional types and associated wear patterns indicate that hunting and attendant on-site butchering and processing of game were probably the primary economic activities at site 45-D0-282. Lacking definable cultural features and having only a sparse faunal collection, we must base this inference on our functional analysis. On the other hand, site activity may have been focused on the production of stone tools, and hunting was undertaken only to provide very short-term subsistence. Yet another possibility, not unrelated to a focus on hunting, is the use of on-site produced too! forms to manufacture and maintain non-lithic items of the tool kit. Utilized and retouched or resharpened flakes, as well as scrapers, bifaces and a wide variety of other formed tools, may have been used to manufacture wooden or bone implements-shafts for spears or atlatis, points for the projectile shafts, handles for knives and scrapers, etc. Weighing all the evidence--a lack of bone in the site deposits, no identified firepits or living surfaces, and incontrovertible evidence of tool production and tool maintenance, with tool types commonly associated with hunting--we conclude that 45-D0-282 was the scene of very short-term camps where tools were made preparatory to hunting, but where little butchering or processing was done. The massive erosion of the site area nearest the Columbia River adds uncertainty to this conclusion, of course. In addition to the tool forms discussed above, the beach collection also held evidence of firepits, river mussel collection, and fragmented food bone, all of which indicate a prehistoric site economy more in line with the postulated emphasis on hunting and butchering, as well as supplementing the meager evidence for consumption of other resources. It seems plausible that tools were made and used upslope from the area of camping and everyday living

se ¥₹\$

When we examine the ranking of functional types by type of wear for the entire assemblage, we find a varied lack of correspondence in most wear categories. Those rankings which are congruent include tabular knives in smoothing on edges only, gravers and drills in smoothing on points only, drills in teathered chipping on points only, drills in feathered chipping on points only, drills and gravers in hinged chipping on points only, and choppers, hammerstones and pestles in all types of crushing wear except crushing on points only. Wear types on unifacial and bifacial edges show marked variation in the proportional ranking, generally characterized by the dominance of simple utilized flakes in most categories. It seems that utilized flakes, the most frequent tool form in the collection, were also the favored multipurpose tool, used for a wide range of purposes not limited to sharp unifacial or bifacial edges, but also points, and spanning all four major wear classes from smoothing to crushing. In general, it would seem that rigid selection of a particular tool form was largely confined to the manufacture of points and thus, functional types such as gravers, drills and projectile points. Edged tools, unifacial or bifacial, seem to have had more varied uses, commensurate with a more generalized tool form. The dubious association of tabular knives and smoothing wear on edges only does not seem to be a matter of tool design since these tool forms are among the crudest and least manufactured; rather, it probably represents use of a convenient stone with a tabular fracture plane for a certain job or very restricted range of jobs. Whatever the actual range of uses of these function types, examination of associated wear types clearly documents use of most edged tool forms for a wide variety of tasks, not necessarily predictable from traditional functional labels. While there is a tendency for obvious (i.e., specialized) tool forms, particularly those with points, to have been used in a manner suggested by the functional label, it is clear that tools were used for a number of different jobs and not restricted to a single job. We have noted that the simple utilized flake was adapted to the widest range of tasks. Less obvious examples include projectile points, used for cutting and scraping as well as perforating, and compares, with hinged chipping wear more indicative of heavy cutting than scraping of soft hides.

SUGGESTED USE

Feathered chipping and feathered chipping-smoothing most likely represent light cutting operations on comparatively soft materials--hide, meat, tendon or soft plant parts. Hinged chipping and hinged chipping-smoothing indicate heavier, deeper cutting actions in which the tool comes into contact with bone, gristle or other hard but elastic material. Smoothing by itself may be more insterial dependent, with similar wear patterns produced by quite different uses. For example, smoothing along a unifacial or bifacial edge on a cryptocrystalline tool likely evidences light cutting or scraping use on a soft, elastic material. However, smoothing wear on an edge only on a quartzite tool, with its denser, less brittle and less sharp mass, may indicate cutting on hard, dense material which simply wears down the edge. Our cursory analysis does not permit us to investigate smoothing wear more

thoroughly (i.e., does the smoothing wear obliterate flake scars or other landmarks along the working edge, or does it obliterate the manufacture altogether, or are there strike within the smoothing wear? etc.). Crushing wear, either in combination with pecking or hinged or feathered chipping, indicates heavy tool use and repeated contact with hard surfaces like bone and/or stone working supports.

たいというというない。ここともない、「「「」」ということでなるなど、「「」」

「「「「「「」」」」

In general, then, we have four primary tool types described by attributes of wear: smoothing on edges and points, feathered chipping on edges and points, hinged chipping on edges and points, and crushing of edges, points and surfaces. Combinations thereof indicate variable functions, variable intensity of use, or persistent reuse of tool forms. The tabular knife provides a good example of the difficulty involved in assessing tool use within these broad attribute categories. Characterized by smoothing wear on edges only, tabular knives are ubiquitous. Because the smoothing wear does not extend onto any adjoining planar surface, we speculate that use was essentially vertical--the tabular knife was held upright in the hand and used to cut, scrape or saw through elastic material of some hardness, and perhaps came into contact with a stone working base. Certainly, the attrition of the edge, which obliterates flaking irregularities or other landmarks of manufacture, is not the result of cutting or scraping of soft, elastic material such as hide or meat, unless the hides or meat were worked over a solid, hard base which, rubbing against the knife, dulled the working edge over extended periods of use. Whatever their actual use, their wear patterns distinguish them from other flake tool forms on which smoothing consistently occurs on unifacial and bifacial adges and points, indicative of cutting, scraping and perforating uses, usually on relatively soft, tractable materials.

Another example of the difficulty of assessing tool function lies in the simple distinction between feathered and hinged chipping wear as distinct tool types. This distinction is the least pronounced of the four defined wear types--similar tool forms characteristically have both kinds of wear, although one or the other tends to predominate. We may explain this distinction on the basis of both cutting activity and worked medium--feathered chipping is produced by light cutting on relatively soft materials while hinged chipping reflects heavier, deeper cutting in which the tool comes into contact with harder, but still elastic materials. Or we may suggest that the distinction rests on the intensity and/or duration of use of the tool. Finally, we may submit that the difference, unless clearly correlated with distinctive tool forms, is inconsequential: both wear types indicate general butchering activity; any distinctions result from random use of like tool forms for light or heavy cutting, or variation in intensity or duration of use.

All of the formed tool types recovered show feathered and hinged chipping wear. Those with the least manufacture (e.g., simple utilized flakes and linear flakes) show the highest occurrence of feathered chipping wear. More complex tool forms or those that show resharpening or retouch (e.g., scrapers, blfaces, burins, projectile points, resharpened and retouched flakes) have proportionately higher frequencies of hinged chipping wear. Drills are an exception--feathered chipping wear slightly exceeds hinged chipping wear. The

seeming correlation between feathere; chipping wear and hinged chipping wear and relatively unmodified and carefully shaned or maintained tools respectively, leads us to suspect that the wear types may be largely a function of the intensity or duration of use in comparable activities.

EDGE ANGLE DISTRIBUTIONS

Measurement of edge angles within these general functional classes gives us another, complementary method of evaluating the function of different tool forms and differences in the activities represented within the defined zones. Figure 3-4 illustrates edge angle distributions for functional types for three classes of tunctional types: utilized flakes, refouched and resharpened flakes, and all other functional types excluding pestles and hammerstones. It also presents edge angle distributions by the two largest possible classes: objects with wear only and objects with wear and manufacture. Edge angle distributions of tunctional types within these classes are listed in Appendix B, Table B-1 to facilitate comparison since many of these artifacts are present in numbers too low for meaningful histograms to be drawn.

Edge angle distributions generally support inferences drawn from consideration of attributes of wear. Simple utilized flakes show a distribution skewed toward an acute edge angle in the range 16-31 degrees, reflecting selection for a sharp cutting edge and little concern for durability. Retouched and resharpened flakes show a more regular distribution, with a peak in a less acute edge angle range (46-66 degrees), which is also repeated in the distribution drawn for other functional types. These distributions indicate a paramount concern with strength of the working edge or point, and, thus, greater application of force and durability. When these three functional type classes are grouped into two major groups of wear only and wear and manufacture, this fundamental pattern shows even more clearly. Tools with wear and manufacture show a more normal distribution centered in a broad range from 46-65 degrees. Certainly, there is considerable overlap between the two distributions, particularly in the wear only distribution where there are two small peaks in the range 36-45 degrees and 50-55 degrees; but the different characteristics of these edge angle distributions reflect care in selection of a sharp edge for jobs of the moment and creation of less acute edge angles for formed tools for which design and durability were salient concerns.

ECONOMIC PATTERNS

いたが、「おおおおおおがったないか」では、それないでは、「「おおおおからの」、「おおおおからの「「おおおおから」」で、「おおおおから」で、「おおおおからない」で、「おおおおおおおおおおおおおおおおおお

The overwhelming majority of stone tools recovered from 45-00-282 document cutting, plercing, scraping, and chopping uses in soft to hard elastic materials, characteristics commonly associated with huntingbutchering-processing of game (97\$, N=1,127). Many of the tool forms could have been used for other economic pursuits, notably the processing of plant parts or woods, but the character of the assemblage seems to argue for hunting. Feathered and hinged chipping wear, often associated with smoothing, and primarily on unifacial and bifacial edges of simple flake tools, bifaces,



Figure 3-4. Edge angle distribution of functional tool types, 45-00-282.

burins, drills, and projectile points, indicate tool use on soft and hard materials or consistent reuse and heavier use of some functional types. Smoothing on the edges of tabular knives and the recovery of a large number of scrapers may indicate an emphasis on hide processing. However, it is equally likely that these forms may have been used to manufacture non-lithic elements of the tool kit; for instance, to shape and smooth wood or bone foreshafts and handles. Heavy crushing wear on the unifacial edges of choppers and surfaces of the numerous hammerstones may evidence considerable attention to marrow extraction and bone tool manufacture, or the working of small wood parts. Other heavy tools include 28 hammerstones, which may be further evidence of the importance of bone reduction or the emphasis on lithic reduction. Recovery of two pestles clearly documents the processing of plant parts at the site as well.

TEMPORAL AND SPATIAL PATTERNS

Differences in artifact distribution among zones are more a matter of the presence or absence of particular functional types than significant changes in intensity of tool use or wear patterning (Table 3-25). Differences among excavated zones, in particular, seem insignificant. The most marked contrast occurs between the excavated zones (Zones 1 through 4) and the beach collection (Zone 25). For example, Zone 25 yielded a much lower proportion of simple tiake tools and a correspondingly higher proportion of formed tools than the excavated zones. Tabular knives were only recovered from the beach collection, and bifaces, scrapers and projectile points comprised a much higher proportion of that zonal assemblage than in the excavated zonal assemblages.

Na performed a chi-square test for two or more independent samples on the distribution of functional types (using the collapsed categories of Table 2-25) across all five derived zones. and for the distribution of tool types (kind of wear by location of wear) across excavated zones and the beach collection. In both instances the derived chi-square value easily exceeded the critical values at the .05 level of significance. We conclude that the distribution of functional object types and tool types does vary significantly among the defined analytic zones and between the excavated zones and beach collection, and that the beach collection represents a different set of activities. As the beach collection is comparable in age to the other zones, this may indicate activity patterning on the site, with a marked difference in site use between an area closer to the river and one turther upslope. However, the beach collection is not comparable to the assemblages from the buried zones, either in duration or in recovery techniques. It is possible that proportions of specific artifact types within Zone 25 are inflated simply because it is a remnant surface onto which artifacts from the other zones settled. The presence of tabular knives in the beach collection and their absence in the other zones is the most striking difference.

activities, which was located nearer the river, and that the rising waters of Rufus Woods Lake have all but obliterated evidence of this site activity.

All evidence points to a very con. 'stent pattern of site use over the postulated 3,000 year span of occupatic, from ca. 7000-4000 B.P. This period, defined as the Kartar Phase in the Rufus Woods Lake project area, saw use of 45-D0-282 as very short-term camps, perhaps not even overnight stops, associated with tool production and tool kit maintenance. We have some very inconclusive evidence that the part of the site nearest the river (preserved as the Zone 25 beach collection) may have seen more intensive and prolonged use, but still confined to camps.

STYLISTIC ANALYSIS

Projectile points are the only class of artifacts from site 45-D0-282 that permit the researcher to make assessment of temporal period and/or cultural affiliation. They supply us with a reasonable temporal scale when we compare stylistic attributes of specimens in this collection with those considered diagnostic of defined projectile point types, either within this project area or on the Columbia Plateau as a whole. At 45-D0-282 this is particularly important, since we do not have radiocarbon dates or distinctive, dated geologic deposits.

PROJECTILE POINT CLASSIFICATION

Two separate but conceptually related analyses are used to classify projectile points. A morphological classification is used to define descriptive types that do not directly correspond to recognized historical types. This is intended as an independent check on the temporal distribution of projectile point forms in the Rufus Woods Lake project area and as a means to measure the distribution of formal attributes as well as point styles. An historical classification correlates these projectile points with recognized types with discrete temporal distributions. A multivariate statistical program which compares line and angle measurements taken along the outlines of the points is used to classify the specimens. Together, these analyses allow us to (1) assess formal and temporal variation in our collection without first Imposing prior typological constructs, (2) correlate specimens recovered from our study area with those found elsewhere on the Columbia Plateau in a consistent, verifiable manner, (3) develop a typology that incorporates both qualitative and quantitative scales of measurement, and (4) examine the temporal significance of specific formal attributes as well as aggregates viewed as ideal types.

Eleven classificatory dimensions have been defined for morphological classification: BLADE/STEM JUNCTURE, OUTLINE, STEM EDGE ORIENTATION, SIZE, BASAL EDGE SHAPE, BLADE EDGE SHAPE, CROSS SECTION, SERRATION, EDGE GRINDING, BASAL EDGE THINNING, and FLAKE SCAR PATTERN (Table 3-26). Of these, the first four (DI-DIV) define eighteen morphological types (Figure 3-5). The other seven serve to describe these types more fully, and permit the identification of variants within the types.

Table 3-26. Dimensions of morphological projectile point classification.

なるのです。

DIMEN	SION II BLADE-STEN JUNCTURE	DIMEN	SION VII: CROSS SECTION
M	Not energia	N.	Not applicable
4	Side-ontched	1.	PLanoconvex
	Shout de ted	2.	Biconvex
	Shunand	3.	Dimond
3.	Squereo Saubad	.	Tracez di dei
	Terinte minete	9.	Indeterminete
	Tuôn sa un tun ca	••	
DINEN	SION II: OUTLINE	DINEN	SION VIII: SERRATION
N.	Not applicable	N.,	Not applicable
	Triangular	1.	Not servated
· •	lancaolate	5.	Serreted
4	Todeterminete	8.	Indeterminete
DINE	SION III: STEN EDGE ORIBITATION	DIMEN	SION IX: EDGE GRENDING
М.	Not anolicable	H.	Not applicable
1.	Stealoht	1.	Not ground
2	Contracting	2.	Slade edge
3.	Excending	3.	Stan edge
g.	Indeterninete	9.	Indetensinete
••			
DIME	ISION IV: SIZE	DIHEN	ISION X: BAGAL EDGE THUMING
N.	Not applicable	М.,	Not applicable
1.	Large	1.	Not thinned
2.	Smell	2.	Short flake scars
		3.	Long flake scars
DINE	ISION VI BAGAL EDGE SHAPE	9.	Indeterminete
H .	Not applicable	DIMEN	SION XI: FLAKE SCAR PATTERN
1.	Straight		
- 2	Convex	N.	Not applicable
3.	Concave	1.	Variable
4.	Point	2.	Uni farm
5.	1 or 2 and notched	З.	Hi xed
9.	Inde Lore i ne Lo	A.,	Colisteral
		5.	Transverse
DINE	ISTON VI: BL.DE EDGE SHAPE	6.	Other
		9.	Indeterminete
H_	Not applicable		
1	Streight		
2	Excurvete		
3	Incurveta		
4.	Reported		
4.	Reported Indetare the La		



· · · •



e,

By defining the margins of projectile points, we are able to place them within one of the eighteen morphological types. This is done by drawing straight lines from nodes where the outline of the specimen changes direction. Figure 3-6 illustrates the technique. For a corner-notched triangular point, the blade is defined as line segment a A. The shoulder is line segment A 1. The neck is node 1. The stem is line segment 1 2. The base is line segment 2-a'. Terms applied and the number of line segments drawn vary given the two basic subdivisions of form. Lanceolates are generally defined by four or less line segments (aA12a'). Stemmed triangular forms are defined by five or less line segments (aA123a'). Side-notched triangular forms are defined by five or more line segments (aA12345a'). Table 3-27 lists the eighteen morphological types with descriptions, classification codes, and line segment definitions.

Cross-tabulation of classificatory dimensions DV-DXI supplies detailed descriptions of the eighteen morphological types and allows us to assess the temporal distribution of formal attributes as well as that of point styles. We might subdivide any or all of the types in terms of their basal edge shape, serration, or flaking pattern. We can also assess the chronological significance of concave bases, serrated margins, or regular collateral flaking pattern independent of associated morphological type. Further, we can use this information to establish variants in the basic historical types.

We have defined historical types on the basis of line and angle measurements in order to have a consistent classification method which utilizes published illustrations of projectile points. Other measurements such as weight and thickness were taken on projectile points in our collection, but problems of cost and efficiency precluded handling of specimens from other study areas. These measurements can be included in analyses of cur points, and, hence, for definition of types and type variants that will correlate with acknowledged types, but they are not part of the initial typological exercise. Justification for this decision is found in prior research emphasizing the outline of projectile points as the basis of classification (Benfer 1967; Ahier 1970; Gunn and Prewitt 1975; Hoimer 1978).

Our desire for a statistically derived classification prompted selection of a multivariate statistical method termed discriminant analysis (Nie et al. 1975). In this analysis, individual specimens are sorted into selected groups on the basis of mathematical equations derived from analysis of cases with known memberships. First, we assembled representative specimens for each acknowledged historical type, and tested group autonomy through analysis of specified discriminating variables. Then, we used derived equations called discriminant functions to assign specimens in our collection to the statistically defined projectile point types. All cases are given a probability of group membership, calculated as the distance a given case score is away from a group score. Discriminating variables--those providing the most separation between groups-are ranked and serve as type definitions. The outcome is a statistically defensible projectile point typology based on traditional, intuitively derived classifications. The resulting classification is consistent, and produces mathematically defined ranges of variability. It enables the researcher to quickly categorize a large collection, and it offers a sound, rational basis for definition of new types



Figure 3-6. Definition of projectile point outline.

Туре	Description	Classification	Definition
1	Large Triangetar	N 1 N 1	-
2	Smell Triangular	N 1 N 2	•
3	Large Side-notched	1 8 8 1	aA123, aA1234, aA12345
4	Small Side-natched	1 N N 2	aA128, aA1234, aA12346
5	Lance ol a te	N 2 N N	Ae
8	Shouldered LanceoLate	2 2 N N	sA, sA1, sA12
7	Large, Shouldered Triangular, contracting stam	2121	aA, aA1
8	Smell, Shouldered Triangular, contracting stam	2122	eA, eAl
9	Large, Shouldered Triangular. non-contracting stam	2 1 (13) 1	u/12, a/123
10	Small, Shouldered Triangular, non-contracting stem	2 1 (13) 2	aA12, aA123
11	Larga, Squared Triangular, contracting stam	3121	aAl
12	Small, Squered Triangular, contracting stam	3122	eX1
13	Large, Squered Triangular, non-contracting stem	3 1 (13) 1	aA12, aA123
14	Small, Squared Triangular, non-contracting stam	3 1 (13) 2	aA12, aA123
15	Large, Barbod Triang-Lar, contracting stem	4121	the contract of the contract o
16	Small, Barbed Triangular, contracting stem	4122	the contract of the contract o
17	Large, Berbed Triangular, non-contracting stem	4 1 [13] 1	sA12, sA12).
18	Smelt, Barbed Triangular, non-contracting stem	4 1 (13) 2	al12, al123

`.

ş. .

Table 3-27. Line segment definition of morphological projectile point types.

82 as well as an explicit definition of accepted types. We can thereby correlate the Rufus Woods Lake projectile point sequence with other chronologies in both a quantitative and qualitative manner. For a detailed discussion of procedures and assumptions involved in discriminant analysis see Johnson We assembled a type collection for the Columbia Plateau of over 1,200

specimens that constituted originally defined type examples, labelled specimens of recognized types, or type variants that were reasonably welldated. By critically reviewing the archaeological literature, we identified 23 historical types which we arranged in six formal type series (Figure 3-7). We consistently applied distinctions based on the original type definitions, modified, where appropriate, by subsequent research. We routinely defined type variants, usually suggested by prior researchers, which segregate specimens according to diagnostic patterns in morphology. Historical types identified here represent a synthesis of projectile point types and cultural reconstructions postulated by researchers in different areas of the Columbia Plateau, and were not taken from any single typology or chronological sequence (e.g., Butler 1961, 1962; Nelson 1969; Leonhardy and Rice 1970). Names are usually those applied by the first researcher to define a specific type. We developed variant labels by using the accepted type name followed by a letter denoting diagnostic variation. For a complete discussion of procedures followed see Lohse (1984g).

THE PROJECTILE POINT ASSEMBLAGE

(1978) and Klecka (1980).

Examples of six different historica! types were recovered from 45-D0-282 (Table 3-28). Descriptions of individual specimens follow in an outline form xspecifying physical characteristics and correlations with established projectile point types. Listings of authors and comparable illustrated specimens are not exhaustive, but are meant to alert the reader to artifact assemblages recovered from nearby study areas. Three measurements are given for each specimen: length, taken along a perpendicular axis bisecting the blade and haft; width, taken along a horizontal axis passing across the broadest part of the blade or blade-haft juncture; and thickness, taken through the blade-haft juncture. Specimens are illustrated in Plate 3-3 and digitized outlines are shown in Appendix B, Figure B-1.

CASCADE A (21) N=6

Material:	Measurement:
Jasper	-/1.5/.6 cm
Jasper	-/2.3/.7 cm
Jasper	-/1.4/.7 cm
Jasper	3.6/1.5/.5 cm
Jasper	-/1.5/.5 cm
Chalcedony	3.6/1.5/.4 cm
	Material: Jasper Jasper Jasper Jasper Jasper Chalcedony

		HISTORICAL	TYPF CLASS	FICATION		
DIVISION	4	INCEOLATE		TRIANG	ULAR	
SERIES	Simple	SHOULDERED	SIDE-NOTCHED	CORNER-REMOVED	CORNER-NOTCHED	BASAL-NUTCHED
түрЕ	11 LARGE LANCEOLATE	12 LIND COULEE	41 COLD SPRINGS	51 NESPELEM BAR	61 COLUMBIA A Corner-notched	71 CUILOMENE A Basal noiched
	15 WINDUST C Contracting base	13 WINDUST A	42 PLATEAU Side-notched	52 RABBIT ISLAND A 53 RABBIT ISLAND B	62 QUILOMENE Corner noiched	72 OUILOMENE 8 Basal noicheó
	21 CASCADE A	31 MAHKIN SHUULDERED			63 COLUMBIA B Corner-notched	73 COLUMBIA STEM A
	23 CASCADE C			ראיניטען איזעראיזער איזעראיז. איזעראיזער איזעראיז	64 WALLULA Ractengular stermod	74 COLUMBIA STEM B 75 COLUMBIA STEM C
	Types are numbered consecutive Type names are those most comi	iry within formal serves a two-orga code interant north applied - Mahkin Shouldered and Nespe	and approximate terracie to a solution of the the the types delined for the	e Aulus Woods Lake project area		

.

Figure 3-7. Historical projectile point type classification.

「日本語」現在はなななない。 中心にないため、東というか

Comment: All specimens, with possible exception of the two from Zone 25, appear to have been made on large blades. Initial reduction seems to have entailed percussion flaking, followed by pressure flaking, to define and sharpen the edges. All specimens have been basally thinned, either through removal of series of small short flakes from the margin all around the convex base or by removal of large long flakes from the lateral margins only. All show uniform collateral flaking extending in from the lateral margins to the midlines of the points, which in at least two examples is the arris of a large blade.

Comparable Specimens: Collier et al. 1942: Plate 1, j-m. Cressman 1960: Figure 41a, A, B. Butler 1962: Figure 9tt. Swanson 1962: Figure 36g. Leonhardy 1968: Figure 7h-q. Nelson 1969: Figure 421, n; Figure 43k, I. Leonhardy and Rice 1970: Figure 3b; Figure 4a-d. Chance and Chance 1982: Figure 165d, g, j; Figure 166a, d; Figure 169b, c; Figure 170k; Figure 175a, d; Figure 180d.

			·····	البالا الكالي المحدود المتحد والمحد والمحد والمحد والمحد والمحد
Historical type	Horchological classification	Zone	Feature	Association
Mah ha a Chaut daard	MI 1/20204 MM			
Mankin Shouldered	MINE 2221 NMI	23		
Cold Springs Side-notched	1PRID221NNI	23		
Cold Springs Side-notched	1 NA13111 NM	23		
Cold Springs Sidemotched	1NN12121NN1	4		
Lot Springs Side-notched	1NN11929400	4		
Hank n Shoul Gered	NCNHOC21124	23		
Lascade A	N2 NN2 21 11 23	23		
Lascade L	NZNNE132123	1		Cuefnee fied
Lascade A	N2NN2211122	1	42	Sufface find
Cascade A	N2NN9222193	3	13	
Cascade A	N2NN2222124	3	13	LODDLE SUFFACE
Cascade A	N2NN2132124	3	14	LODDLE SUFFECE
Mahkin Shouldered	22 NN1211122	25		
Mahkin Shouldared	22 NM1 211122	25		
Mahkin Shouldered	22 NN2221123	25		
Nespelam Ber	21212111NN	1		Surface find
Nespetem Sar	21219121NM	25		
Nespelem Bar	21212122NN3	25		
Columbia B corner-notched	21321221 NN3	25		
Blade fragments				
	99919921993	25		
	92919231994	25		
agastar share	999999921994	2		
	92919221993	3		
Stem frequents				
	99315929NN9	2		<u></u>
	99292929 NNB	2		
	99291929 NN9	2		
	99211929 N33	3		
		-		
	99212939123	25		
	99222929429	20		
	00221020120	2		
	00242020420	2	14	Cobbie sucface
	40204020NND	3		
	12221258KM	-		

Table 3-28. Classification of projectile points and projectile point fragments, 45-D0-282.

いたなな問題というないので問題というというで見たのであるとなって目的なななななどの問題となるなかななない。ためにはないのでのなどのなどのなどであると思想してなってきた。

CASCADE C. N=1

Ŋ

いたかったのが思想ななななななながられたが、そのでのなどのなどのなどのなどであったからないないないないとのできたとうできたが、そうないないできたのです。そのためないでは、そのためなどのなどのなどのないで、

Provenience:

Material:

Measurement:

Zone 1 Jasper 4.1/1.5/.6 cm

Comment: This specimen is a classic Cascade type projectile point as defined by Butler (1962) and redefined by Nelson (1969). It has a very regular lanceolate shape, with the widest part of the blade in the lower one-third of the outline. The base has been carefully thinned, but the mid-part of the specimen retains the thick, diamond-shaped cross section held to be characteristic of manufacture on a blade. Lateral margin serrations are large, and extend up from about the blade haft juncture to the distal point. Flaking is collateral, with flake scars of irregular width but uniform carry to the midline.

Comparable Specimen: Leonhardy and Rice 1970: Figure 4.

MAHKIN SHOULDERED LANCEOLATE (31) N=4

Provenience:	Material:	Measurement:					
Zone 25	jasper	3.0/1.5/.5 cm					
Zone 25	Jasper	-/1.6/.7 cm					
Zone 25	Chalcedony	4.4/2.0/.4 cm					
Zone 25	Basal t	2.9/1.5/.5 cm					

Comment: These four specimens, although shouldered lanceolate forms, are quite distinct, both in style and manufacture; this is very characteristic of the broad range of forms defined as Mahkin shouldered (Lohse 1984g). The basalt specimen is a squat, leaf-shaped form, roughed out by percussion flaking on a broad, thick flake. The striking platform and bulb of percussion are still evident at the proximal end. The chalcedony specimen is an elongate leaf-shaped form, roughed out by pressure flaking on a long, thin blade. No striking platform or bulb of percussion is visible. Only the margins have been reduced, leaving a weil-defined arris along the midline of the point. One jasper specimen leaf-shaped with the thick, diamond-shaped cross section indicative of manufacture on a secondary flake. Both have been reduced uniformly on the dorsal and ventral surfaces. Flaking patterns on the four specimens are quite variable, ranging from irregular, with reduction of only one surface, to fine, even collateral flaking, and reduction of both surfaces.

This type has been referred to as "points with slight shoulders and rudimentary stems" (Nelson 1969:113) and as shouldered or stemmed leafshaped points (Swanson 1962). It is considered to be a form transitional from lanceolate to stemmed or triangular, spanning a long period from about 6500-2000 B.P.

እግሥላ የእንዲሆን የሆን እንዲሆን እንዲሆ እንዲሆን እንዲ Comparable Specimens: Cressman 1960: Figure 41a,C,D,E. Swanson 1962: Figure 20m,n. Leonhardy 1968: Figure 7f,g. Nelson 1969: Figure 37a-d. Leonhardy and Rice 1970: Figure 14e. Chance and Chance 1982: Figure 163a; Figure 164b,c,g-i; Figure 167e; Figure 169b.

COLD SPRINGS SIDE-NOTCHED (41) N=4

Provenlence:	Material:	Measurement:					
Zone 25	Basalt	4.7/2.2/.5 cm					
Zone 25	Jasper	-/-/- cm					
Zone 4	Opal	4.4/2.2/.9 cm					
Zone 4	Jasper	-/-/.5 cm					

Comment: The basalt specimen is a long, elegant form with excurvate sides made on a large, thin primary flake. The side notches are broad and shallow. The basal margin is lightly notched on either side of the midline. The dorsal surface has been uniformly reduced, but the flat ventral surface shows reduction only along the lateral margins. The opal specimen is bulkier and less finely finished, made on a large, thick primary flake. A large potlid on the ventral surface attests to heat treatment prior to reduction, although inclusions in the stone still resulted in an irregular flaking pattern and rough appearance. Both jasper specimens are large side-notched basal fragments on thick flakes. Flaking on both appears to have been regular.

The basally notched specimen is unusual in collections from Rufus Woods Lake, and has no real correlate in assemblages recorded for nearby study areas.

Comparable Specimens: Bryan 1955: Plate 11. Shiner 1961: Plate 356, 46b. Fryxell and Daugherty 1962: 46b. Nelson 1969: Figure 37p-q. Leonhardy and Rice 1970: Figure 4e,f.

NESPELEM BAR (51) N=3

Provenlence:	Material:	Measurement:				
Zone 1	Jasper	-/2.0/.8 cm				
Zone 25	Jasper	-/-/.5 cm				
Zone 25	Chalcedony	4.1/1.7/.4 cm				

Comment: The chalcedony specimen is a finely flaked, serrated form made on a large, thin flake or blade. The dorsal surface has been uniformly reduced. The vential surface has been flaked only along the lateral margins. Interestingly, the bulb of percussion appears to lie at the distal tip of the point. The jasper specimen from Zone 25 is an elongate, triangular form made on a thick flake or blade. Initial reduction appears to have been by percussion flaking. Later modification of the edge was by pressure flaking. A snap running the length of the projectile point, paralleling the midline, most probably occurred during manufacture. The jasper specimen from Zone 1 was crudely roughed out by percussion flaking. A lateral break through the blade below the tip probably occurred during manufacture.

The jasper specimen from Zone 25 appears to have been used as a cutting tool after a break terminated the manufacturing process. Heavy edge attrition on the intact lateral margin suggests its use as a backed knife, with the flat break employed as a convenient point of leverage.

Comparable Specimens: Swanson 1962: Figure 20m. Nelson 1969: Figure 37b; Figure 41t,u. Chance and Chance 1982: Figure 158q; Figure 164i; Figure 172g; Figure 174c.

COLUMBIA CORNER-NOTCHED B (63) N=1

Provenience:	Material:	Measurement:				
Zone 25	Chalcedony	2.5/1.2/.5 cm				

Comment: This specimen is a small triangular form made on a short, thick flake. The flaking pattern is mixed but tends toward collateral. The stem is expanding and the lateral margins mildly excurvate.

Comparable Specimens: Nelson 1969: Figure 41pp-rr. Leonnardy and Rice 1970: Figure 7f-1. Chance and Chance 1982: Figure 158g; Figure 164d.

UNNAMED TRIANGULAR PROJECTILE POINT (81) N=1

Zone 25 Jasper 3.3/1.8/.4 cm

Comment: Similar specimens often are called blanks or preforms; however, attrition of the edges may indicate that this form is a functional tool rather than an unfinished projectile point.

Authors refer to these forms as small triangular projectile points with the caveat that they may or may not be confined to that functional category (e.g., Collier et al. 1942; Nelson 1969).

Comparable Specimens: Nelson 1969: Figure 44r. Chance and Chance 1982: Figure 150, i.

DETACHED BLADES N=4

Provenience:	Material:	Measurement:			
Zone 2	Jasper	-/1.8/.8 cm			
Zone 2	Jasper	-/1.4/.7 cm			
Zone 25	Jasper	-/1.6/.6 cm			
Zone 25	Jasper	-/-/.7 cm			

Comment: All four specimens represent lanceolate projectile point liadetip fragments. Three have diamond shaped cross sections, the other, a thick biconvex cross section. The two specimens from Zone 25 show mixed flaking patterns, tending toward collateral. The two specimens from stratified contexts, Zones 2 and 3, have uniform collateral flaking patterns.

These four specimens are lanceolate forms, and, given associated projectile point types, are most probably Cascade or shouldered lanceolate types dating from about 6500 to 3500 B.P.

Comparable Specimens: None. Illustrated examples of lanceolate and shouldered lanceolate forms cited previously are appropriate.

DETACHED CTEMS N=4

Provenlence:	Materiai:	Measurement:				
Zone 2	Jasper	-/1.4/.5 cm				
Zone 2	Jasper	1.1/1.5/.5 cm				
Zone 3	Jasper	-/1.4/.7 cm				
Zone 2	Jasper	-/1.5/.5 cm				

Comments: All four specimens are classified as stems because of the presence of a blade-haft juncture or overall configuration. All have been roughly flaked, either by percussion or pressure techniques. One specimen from Zone 2 has a slight basal notch at the approximate midline of the stem. All probably represent completed projectile points broken during use.

The specimen from Zone 3 may represent a lanceolate form since it lacks a definable blade-haft juncture; however, it is included here as a stem because its lateral margins flare out, perhaps indicative of a bladehaft juncture immediately above the lateral snap.

All four detached stems are considered representative of sloping and square-shouldered triangular types. Their large size probably indicates earlier forms diagnostic of a period from about 5000-3000 B.P. (cf., Nelson 1969). The notched stem specimen from Zone 2 may represent a "Quilomene Bar Base-notched" type (Nelson 1969: Figure 38, 1-p), which dates in this project area from about 3000-500 B.P. (Lohse 1984g).

Comparable Specimen: None. Illustrated examples of large, sloping and square-shouldered triangular projectile points are appropriate (cf., Nelson 1969).

DETACHED BASES N=5

Provenience: Zone 4 Zone 3	Material:	Measurement:				
Zone 4	Chalcedony	-/-/.3 cm				
Zone 3	Jasper	-/-/.4 cm				
Zone 2	Jasper	-/-/.4 cm				
Zone 2	Argillite	-/-/.2 cm				
Zone 25	Jasper	-/-/.5 cm				

Comment: These specimens are classified as detached bases because they lack a definable blade-haft juncture. All have been reduced through pressure flaking. The jasper specimen from Zone 25 exhibits a long narrow flake or flute extending up from its basal margin through the lateral snap. The jasper specimen from Zone 2 has a large impact fracture on one surface, attesting to breakage after manufacture.

The chalcedony specimen from Zone 4 may represent a side-notched type or expanding stem type, given the constriction of its lateral basal margin. Since Zone 4 produced Cold Springs Side-notched projectile points, it is likely that this base represents a large side-notched form. The other four specimens are most likely lanceolate forms.

Comparable Specimens: None. Illustrated examples of broken lanceolate forms in Nelson (1969) and Chance and Chance (1982) are probable correlates.

Zone 4, the oldest at the site, yielded two Cold Springs Side-notched points and a fragment possibly of the same type. These points are not numerous in the project area, but when found in dated context, occur before ca. 5000 B.P. (cf. Jaehnig 1984a; Lohse 1984f). That points from this zone are limited to Cold Springs Side-notched, without Mahkin Shouldered or Nespelem Bar projectile points, firmly places this early occupation sometime prior to ca. 5000 B.P.

Zone 3 contained only Cascade A points, which is consistent with the later stratigraphic position of the zone, as these points are generally found in slightly later contexts. Zone 2 had no diagnostic point types.

Zone 1 contained Cascade A, Cascade C, and Naspelem Bar points. Again, this association is consistent with a younger stratigraphic position, although the Nespelem Bar point is a surface find and not necessarily a temporal indicator for this zone. It is evident, however, given the occurrence of Cascade A and C point types, that the uppermost zone must be late Kartar Fhase and not early Hudnut Phase, and thus dates prior to 4000 B.P. and probably not earlier than 5000 B.P.

Zone 25, the beach collection, contains point types found in other zones--Cold Springs Side-notched, Mahkin Shouldered, Cascade A, Nespelem Bar-as well as a Columbia Corner-notched B.

Projectile point data, therefore, indicate that all buried deposits at the site are assignable to the Kartar phase (ca. 7000-4000 B.P.) defined for the project area. Not only the classified points but the fragments support this temporal assignment: In all zones, plade, stem, and base fragments are characteristic of simple lanceolate and shouldered lanceolate or large shuldered and side-notched triangular projectile point forms. The vertical distribution of point types in the site suggests a rough temporal sequence of occupation in which Zone 4 corresponds to the early part of the Kartar phase (ca. 7000-6000 B.P.), Zone 3 is intermediate, and Zones 1 and 2 date to the mid- to latter part of the Kartar phase (ca. 5000-4000 B.P.). The beach collection spans the entire time period represented by the buried components, and a Columbia Corner-notched point is evidence of a possible occupation in the Hudnut Phase.









4. FAUNAL ANALYSIS

Zoological remains from archaeological sites provide a unique source of data on the ecology and historic biogeography of animal species living in the site area, and on utilization of faunal resources by human occupants of the site. This chapter describes the faunal assemblage recovered from 45-D0-282, and summarizes the implications of the assemblage for understanding the archaeology of the site.

FAUNAL ASSEMBLAGE

The distribution of invertebrate faunal remains is summarized by zono in Table 2-2. The counts and weights of bone given in Table 2-2 do not represent the entire amount of bone examined by the faural analysts. They were obtained during laboratory processing, after "noneconomic" bone had been removed. Both categories of bone were included in the faunal analysis but additional weights and counts were not taken. As the majority of the identified specimens from 45-D0-282 are rodents originally included in the noneconomic bone category, the counts of identified bone (Table 4-1) are higher than the total bone counts reported in Table 2-2. Of the 496 identified elements 459 (92%) are mammalian, one (<1%) is amphibian, 22 (4%) are reptilian, and 14 (3%) are fish. Taxonomic composition and distribution of the vertebrate remains for the site as a whole and by zone are shown in Table 4-1. The invertebrate assemblage consists of seven shells weighing 56 g. The shells have not been identified.

The following summary presents criteria used to identify elements where appropriate, and comments concerning the past and present distribution and cultural significance of the taxa represented. A summary of the elements of each taxon is provided in Appendix C.

SPECIES LIST

MAMMALS (NISP=459)

Sylvilagus cf. nuttallii (Nuttail's cottontall) -- 9 elements.

<u>Sylvilagus nuttaliji</u> is an abundant resident of rocky sagebrush areas in the project area. Cottontalls were exploited ethnographically for fur and meat (Post 1938; Ray 1932).

Теха	Rite	Site										
	Tote	Totel		1 2		3	3 4		25			
	NISP ¹	MNI ²	NISP	MIZ	NISP	HNI	NISP	HHII	NISP	MNI	NISP	HINI
HAIMULIA (NISP-450)												
Leporidae <u>Sviivilagua nuttallii</u>	10	1	2	? 1	1	1	ı –	-	-	-		
Sciuridae Marmata flaviventris	9	1	-	-	2	1	2	1	-	-	5	1
Geomyidae <u>Thomomys telpoides</u>	327	-	15	4	40	6	104	18	150	18	18	4
Heteromyidae Peroanethus pervue	55	-	7	2	34	8	9	4	5	2	-	-
Cricatidee Lagurus curtatus Microtus sp. Peromyscus seniculatus Nectoma cinars	26 12 1 3 2	- 6 1 1	4 6 	3	10 2 - - 1	2	8 2 1 1	- 2 1 1	31-22-	- 1 - 1	1 1 - -	- 1 - -
Mustelidee <u>Texidee texwe</u>	1	- 1	-	-	:	-	-	-	-	-	1	1
Cenidae <u>Cenia</u> ep.	2	2	-	-	-	-	-	-	-	-	2	2
Cervidee <u>Odocolieus</u> ep.	5 4	- 1	2	ī	1	1	-	-	-	Ξ	5 1	ī
Deer-Sized	1	-	1	-	-	-	-	-	-	-	-	-
Elk-Stred	1	-	-	-	-	-	-	-	-	-	1	-
ANPHIBIA (HISP-1) Rentdaa/Bufontdee	1	1	-	-	-	-	-	-	1	1	-	-
REPTILIA (NISP-22) Chelydridae <u>Chrysenne picte</u>	22	1	1	1	2	1	18	1	-	-	1	1
PISCES (NISP=14) Selecnidee	14	-	1	-	2	-	8	-	3	-	-	-
TOTAL	497		45		96		154	·	165		37	

Table 4-1. Taxonomic composition and distribution of vertebrate remains, 45-D0-282.

Number of identified specimens,
Ninisum number of individuals.

101

Repair C. C.
Marmota flaviventris (yellow-bellied marmot) -- 9 elements.

All marmot remains have been tentatively assigned to the species \underline{M}_{\bullet} <u>flaviventris</u> on the basis of present distribution. This species is the only marmot now living in the project area, and is a common resident of talus slopes. <u>M. monax</u> has been recorded in extreme northeastern Washington and <u>M. calagata</u> occurs in the Cascades to the west of the project area (ingles 1965; Dalquest 1948). The three species are indistinguishable on the basis of osteological morphology, and the size ranges of the three overlap extensively. Marmots were exploited as a small game resource by ethnographic inhabitants of eastern Washington (Ray 1932; Post 1938). Their presence in this faunal assemblage may indicate prehistoric exploitation. Potential changes in distribution or cultural transport of animals preclude dismissing the possible occurrence of one or both of the more montane species in this assemblage.

<u>Ihomomys talpoides</u> (northern pocket gopher) -- 327 elements.

<u>Thomomys taipoides</u> is the only geomyid rodent in the project area. Because pocket gophers are extremely fossorial and there is very little evidence that they were utilized prehistorically or ethnographically, their presence in this assemblage may be considered fortuitous.

Pérognathus parvus (Great Basin pocket mouse) -- 55 elements.

<u>Perognathus parvus</u> is the only heteromyld rodent known in the project area. Like the pocket gophers, <u>P. parvus</u> is most likely present as a result of natural agents of deposition.

Cricetidae (New Yorld rats and mice) -- 26 elements.

Lacurus curtatus (sagebrush vole) -- 12 elements.

Lagurus curtatus generally inhabits dry sagebrush habitat which is sparsely grassed (Maser and Storm 1970:142). Only cranial material of this genus is readily distinguished from <u>Microtus</u> on osteological bases. The occlusal surface of the M^3 of <u>Lagurus</u> has a distinctive morphology (Maser and Storm 1970), and the location of the mandibular foramen is quite different for the two genera (Grayson 1984). La curtatus is probably present in this assemblage as a result of natural processes.

Microtus sp. (meadow mouse) -- 1 element.

Three species of <u>Microtus</u> occur in the site area: <u>M. montanus</u>, <u>M. pennsylvrnicus</u> and <u>M. longicaudus</u>. All three species inhabit marshy areas or live near streams. <u>M. montanus</u> can also be found in more

xeric areas. None of the elements recovered could be assigned to species. There is no evidence that this genus is present because of cultural process.

Peromyscus maniculatus (deer mouse) -- 3 elements.

<u>Peromyscus maniculatus</u> is a resident of all habitat types in the project area.

Neotoma cinerea (bushy-tailed woodrat) -- 2 elements.

Woodrats live in a variety of habitats in eastern Washington (ingles 1965). Woodrats were not considered desirable food by ethnographic inhabitants of the project area (Ray 1932:90).

Mustelidae (weasels, minks and allies) -- 1 element.

Taxidea taxus (badger) -- 1 elements.

<u>Taxidea taxus</u> is a powerful burrower and is found throughout eastern Washington, though not in large numbers. Badgers were regularly trapped by the Sanpoil and Nespelem (Ray 1932:85).

<u>Canis</u> sp. (wolf, coyotes and dogs) -- 2 elements.

Both <u>Canis latrans</u> (coyote) and <u>C. famillarus</u> (domestic dog) are common in the project area today. <u>C. latrans</u> is an indigenous species, <u>C.</u> <u>famillaris</u> has great antiquity in the northwest (Lawerence 1968). <u>C.</u> <u>lupus</u> (wolf) is also known to have been a local resident in the past, but has been locally extinct since about 1920 (ingles 1965). It was not possible to determine the species of these elements. Dogs were used ethnographically for hunting deer, but were not eaten, except in emergencies (Post 1938). Coyotes, however, were considered good food (Ray 1932:90).

Cervidae (deer, elk) -- 5 elements.

Odocoileus spp. -- 4 elements.

Deer-Sized (deer, sheep and antelope) -- 1 element.

Elk-Sized (elk, cow and bison) -- 1 element.

The elements identified as <u>Odocoileus</u> sp. may represent one or both of the two species of deer known in the project area (<u>O. hemionus</u> and <u>O. virginianus</u>). The merapodial fragment identified as deer-sized lacked features that may be used to distinguish deer, sheep and antelope. It

could represent any one of these three taxa. None of the non-artifactual cervid elements displayed evidence of human use such as butchering marks or burning.

AMPHIBIA (NiSP=459)

Ranidae/Bufonidae (frogs, toads) -- 1 element.

Both frogs and toads inhabit the project area (Stebbins 1966). Inadequate comparative material precluded assigning these elements to family.

REPTILIA (NISP=22)

Chrysemys picta (painted turtle) -- 22 elements.

<u>C. picta</u> is the only turtle currently living in the project area. <u>Clemmys marmorata</u> (western pond turtle) has been reported in the eastern part of Washington in the ethnographic literature, but there is no way to ascertain if taxonomic identification is accurate. <u>C.</u> <u>marmorata</u> now occur only on the west side of the Cascades and in the southern part of the state. On the basis of present distribution, all turtle remains have been assigned tentatively to <u>C. picta</u>. The turtle shell in this assemblage is too fragmentary to determine whether it is carapace or plastron.

PISCES (NISP=14)

Salmonidae (salmon, trout, whitefish) -- 14 elements.

These vertebrae could belong to any one of at least eight species of salmonid fish known in the project area. All fish vertebrae with parallel-sided fenestrated centra were assigned to this family.

DISCUSSION

The vertebrate taxa identified from 45-D0-282 are representative of the fauna expected in the project area. All taxa identified currently live in the project area. The assemblage is dominated by rodents, undoubtedly reflecting a natural accumulation of bones in this site. Most elements appear recent and intrusive, consequently, shifting abundances across zones probably reflect more about differences in burrowing behavior among species than environmental or cultural conditions in the past. The relative abundance of <u>Thomomys</u> talpoides increases with depth as the abundances of <u>Perognathus parvus</u> and <u>Lagurus curtatus</u> decrease. Because <u>L. talpoides</u> generally prefers more mesic conditions than either <u>P. parvus</u> or <u>L. curtatus</u>, greater abundances of <u>L. talpoides</u> in the past. However, <u>L. talpoides</u> can burrow to 1.5 m (Szuter 1983:3), while <u>P. parvus</u> and

L. curtatus make burrows that are from 10-45 cm deep (ingles 1965:291; see Szuter 1982). The uniform, recent appearance of most rodent elements in this assemblage supports the suggestion that shifting relative abundances of rodent taxa in this site are a function of differential burrowing ability among different species of rodents taxa.

Only nine identified elements appear to have been burned: one fragment of turtle shell from Zone 1, two salmonid vertebrae from Zone 3, and five antier fragments from Zone 25. The antier artifacts are discussed in Chapter 3. No evidence of butchering was observed on any of the identified elements. The small number of elements that may have been culturally deposited preclude making inferences from the fauna about subsistence or faunal utilization at this site.

5. FEATURES ANALYSIS

Twelve of the thirteen cultural features recorded at 45-D0-282 are arbitrary excavation levels which were designated in their entirety as features because they contained concentrations of basalt, and occasionally granite, rocks in association with high counts of lithic artifacts and debitage. The thirteenth cultural feature consists of two pesties, one of basalt and one of granite, which were found together.

The rock concentration features found in Zones 2, 3, and 4 represent use of natural basalt cobble concentrations for stone tool manufacture and repair. However, there is considerable variety of distribution of basalt among the features. Feature 4, a tight cluster of a few rocks (Plate 5-1), is a firepit. Feature 7 (Plate 5-3) is an amorphous scatter of basalt which contained a chipping station. Feature 5 (93N260E, Level 120) contained no cultural material, and is evidence of the natural occurrence of these scatters (Plate 5-2). Although the cultural materials appear to be primary deposits, the source and depositional history of the basalt is unclear. The basalt, which is largely angular, may have been transported from the outcrops to the south during the formation of the alluvial fan. In no case are the concentrations very dense; in some units (which were not featured), less than six basalt rocks were recorded.

The thirteen cultural features are described briefly below by analytic zone. They occur in three of the fire zones. No cultural features or basait concentrations were found in Zone 1 and Features were not recorded in the beach collection (Zone 25). Figures 5-1 and 5-2 show the distribution of features in each area of the site. Table 5-1 lists the provenience of the features, while Table 5-2 shows the lithics associated with each. The material types of the lithic objects are shown in Table 5-3. There is no separate table for bone or shell; aside from three salmon vertebrae (one in Feature 11, two in Feature 13), fish teeth (in Feature 11), and two pieces of shell (in Feature 12), only rodent bone was recovered from the feature levels.

ZONE 4

Features 3 and 4 in Area A are assigned to Zone 4, the lowest at the site. Zone 4 encompasses the strata of DU II, primarily coarse, gravelbearing strata which are part of the alluvial fan. Feature 3 is a scatter of basait, granite and river cobbles near the present-day beach. Cultural material is limited compared to the features of Zone 3; the feature designation is based primarily on the occurrence of basait. Feature 4 is probably a tirepit: even though the associated basait rocks shows no signs of



Plate 5-1. Feature 4, Zone 4, Area A, 45-D0-282.

国家のなから知識

τ,



Plate 5-2. Feature 5 (noncultural), Zone 4, Area A, 45-DG-282.



Plate 5-3. Feature 7, Zone 3, Area B, 45-D0-282.

100

. .





101

.*

÷

2001

0100000

できょうか たたい スケイ 読

50

いたいと言語でなったからう

こうないたちちちちちちちちちちちちちちち





.....

Area	Zone	Feature	Provenience	Unit size	Level
A	3	1	93 N23 0 E	1x2	70-100
		2	93 N2 40 E	1x2	80
	4	3	33 N2 30 E	1x2	130-140
		4	83 N250 E	1x2	110-150
8	2	6	2555W	1x1	40
_		9	6N6W	2x2	40-60
		10	4526W	2x2	50
	3	7	44 96 W	2x2	6090
		8	3454E	2x2	80
		11	54 S4 E	2x2	80-100
		12	64S14E	2x2	100-110
		13	64 S34 E	2x2	100
		14	54924E	2x2	100-110
			54524E	1x1	120

Table 5-1. Provenience of features, 45-D0-282.

STELL BUTCH

Table 5-2. Provenience of artifacts by feature, 45-D0-282.

	T													
Artifact type				_			Fe	sture						
	1	2	3	4	6	7	8	9	10	11	12	13	14	Total
Bifaca	-	-	_	-	-	2		2	-	4	-	4		0
Projectile point	-	-	-	-	-	-	-	-	_	<u>.</u>	-	-	3	9
Point fragment	-	-	-	-	-	-	-	_	-	_	_	-	-	
Scraper	-	-	-	-	-	_	1	1	-	_	_	_	1	1
Graver	-	-	-	-	-	-	_	<u> </u>	-	-	4	-	_	2
Core	-	-	-	-	-	-	-	1		2	4	4	_	1
3Lada	-	-	-	-	-	-	-	_	-	1	<u>_</u>	<u>'</u>	_	3
Hicrobiade	-	-	-	-	-	7	-	3	-	é	0		-	20
Burin	-	-	-	-	-	i	-	Ĕ	_	-	-	E	2	22
Unifacially retouched flake	-	-	-	-	-	1	-	1	-	2	-	1	1	6
Bifacially retouched flake	1	-	-	-	-	1	-	1	-	1	-		-	4
Utilized flake	-	-	1	1	-	7	2	7	1	18	3	-	я	49
Chopper	-	-	-	-	-	-	-		_	-	-	-	õ	-0
Hammerstone	-	-	-	-		1	-	-	-	1	-	-	-	5
Pestla	-	-	-	-	2	-	-	-	-	÷.	-	-	_	5
Debitage	27	11	28	4	-	652	87	185	33	677	286	200	374	2,562
TOTAL	28	11	27	5	2	672	90	e . 2	34	709	293	206	351	2,669

Table 5-3. Lithic artifacts and debitage by material type, 45-D0-282.

恣

١

Material								Feat	ture					
	-	~	0	4	g	~	ω	8	10	11	12	13	14	Totel
Jasper	11	-	14	3	ı	384	2	115	14	420	5	8	646	
Chal cedory	13	60	12	Q	I	282	36	8	:	12	168	35	105	
Petrified wood	1	1	1	1	•	1	•	; •	: '	, I 1	3	3	3	
Obsidian	1	I	1	ı	1	1	1	- 1	•	0			1	1 10
Opai	-	1	**	ł	1	1	1	1	- 1	U I	-	-	ı	٥
Fine-grained	· I	I	• 1	I	I	1	1	٩	1	. 1	1	1	i	N 0
quertzite								1			I	I	ı	V
Basalt	"	I	1	ł	•	8	۴	ı	•	10	I	G	c	0
Fi ne-grai ned	ł	1	I	I	•	• •	• 1	F	1	2 1	•	וכ	9.	d G
baselt								•			-		-	n
Argillite	i	-	ı	ł	ı	,	I	1	4	Ŧ	I	I	4	c
Granita	ı	ł	ı	ł		ı	•	ı	۰ı	• 1	1	1	1	• -
TOTAL	28	:	ล	9	Q	672	8	201	34	708	283	206	381	2.668

a Vini

fire-modification, they are tightly clustered and contain a few pieces of charcoal and charcoal flecks. Very little material occurred with the firepit.

ZONE 3

Zone 3 contains eight cultural features; again, all eight occur with basalt rock/river cobble scatters. Features 1 and 2, in Area A, may be secondary cultural deposits; river cobbles, basalt erratic fragments and spalls, and a matrix of silty sand may indicate erosional activity in this area of the site. Material counts for Features 1 and 2 are very low compared to Zone 3 features in Area B (an average of 7 artifacts per unit level in Feature 1 versus, e.g., 224 per unit level in Feature 7).

Features 7, 8, 11, 12, 13 and 14 all occur in the south half of Area B (Figure 5-2). Each is marked by a dense scatter of rounded and angular basalt, and river cobbles. Ocher and some charcoal occurs through the isature levels, indicating that some of the featured levels may have contained firepits. A large number of tools and associated debitage in these features suggests they result from primary cultural activity on a basalt-strewn surface. The overbank deposits which characterize this zone have apparently caused minimal disturbance of the features in Area B.

Stone tool manufacture is indicated in the field notes by a "pile of flakes" of the same material (4455W, Level 70, Feature 7), and a possible "knapper's station" in Feature 11 (Level 100). As Table 5-4 shows, twenty percent of the analyzed lithics from Feature 11 have partial cortex. This is very high compared to the number of specimens with partial cortex in other features. The features from Zone 3, Area B, offer the best possibility for the study of lithic technology as they appear to represent manufacturing activity on a single surface.

ZONE 2

Three features are recorded in Zone 2; all are in Area B. Feature 6 is a content feature consisting of two pestles (illustrated in Plate 3-6). The basalt pestle is 21 cm long and 6.5 cm thick; it has smooth sides and is battered on one end. The granite pestle, 22×7.5 cm, has smooth sides and both erds battered. The two pestles were not associated with basalt rock concentrations.

Features 9 and 10 are basalt strata exposed in two excavation units in the north half of Area B. They were vertically separated from the basalt concentrations in the south half by 20-30 cm in which no basalt features occurred. Feature 9 was defined after excavation because of its similarity with Features 7 and 8 (Zone 3, Area B). Feature 10 consists of two distinct layers of basalt (43-45 cm b.u.d., and 50-55 cm b.u.d.) in a sandy silt matrix. This is the only case of directly observable superposition of basalt features at 45-D0-282. From the number of artifacts recovered, especially the large number of rools in Feature 9, we believe these features to be primary cultural deposits.

					
Feature	Material	No cortex	Partial cortex	Full cortex	Unknown ²
1	Jasper	6	-	-	5
	Chal cado ny	5	1	-	6
[Opal	1	-	-	-
	Basalt	1	2	-	-
2	Jasper	1	-	-	-
	Chal cedony	5	-	~	4
-	Argillite	1	-	-	-
3	Jaspar	6	-	-	3
	Chalcedony	5	-	-	7
	Opel	1	-	-	-
4	Jasper	-	-	~	2
-	Unal Cedony	-	-	-	2
	Jasper	81	3	-	164
	Unal cedony	88	-	~	133
	5888L L	1	-	~	5
0	Dasper Chelondenu	10	4	-	22
	Basel +	14	_	-	22
9		23		-	
5	Chai cadoav	43	4	-	88
	Patrificd wood	-	<u>.</u>	-	3/
	Obsidian	-		-	
	Finangrained	-	-	~	2
	quartzite				
	Fine-grained	1	-	-	-
10	Jaanar	2	-	-	40
	Chai cadony	2	-	-	14
	Obsidian	-	-	-	2
	Araillite	-	-	-	1
11	Jasper	64	33	-	118
	Chai cedony	78	3	~	89
	Obsidian	1	-		
	Basalt	4	-	-	2
	Argillite	1	-	-	-
12	Jasper	24	1	~	75
	Chal cedony	28	2	~	65
	Fine-grained	-	-	-	1
	Obsidian	-	-	-	•
13	Jasper	18	-		70
	Chal cedonv	30	-	-	78
	Basalt	2	1	-	5
	Petrified wood	-	<u> </u>	-	1 .
14	Jasper	56	3	-	86
	Chail cedony	58	-	-	74
	Fine-grained	1	-	-	-
	na sa (r				

Table 5-4. Dorsal typography of lithic materials by feature, 45-D0-282¹.

1< 1/4 in flakes excluded.

²Only a sample of the lithic material from 45-DO-282 received full technological analysis. Therefore, for much of this material, the dorsal topography is "unknown".

CONCLUSIONS

The features at 45-D0-282 supplement our knowledge of the well-documented stone tool production activities at the site, but do not give any indication of other types of activities. The reasons for the co-occurrence of rock and artifacts documented by these features are unclear, especially since, in some instances, basalt scatters occurred without cultural material (e.g., Feature 5) and cultural material occurred without rock scatters (e.g., Feature 6). Since ilthic chipping stations have been identified in at least two features, we might speculate that the basalt strata also contained raw material, perhaps jasper nodules, needed for stone tool production. Alternatively, the basalt itself may have been used in firepits in some stages of that production. Given the tenuous evidence for both possibilities, either may be correct.

It is important to remember that these features were arbitrarily bounded by excavation units. Distinct horizontal boundaries occurred in only a few cases (eg. Feature 4). These features are probably but small portions of larger activity areas on deflated basalt-strewn surfaces.

6. SYNTHESIS

45-D0-282 was a frequently used short-term activity site throughout much of the 3,000 year span of the defined Kartar Phase (ca. 7000-4000 B.P.) A large collection of knapping debris and stone tools indicate that a primary activity over that period was the reduction of imported jasper and chalcedony into a broad range of hunting-butchering-processing tool forms. Other site activities are preserved in the recovery of two stone pestles, indicative of some limited plant processing, and a very meager assemblage of ungulate bone, documenting the hunting of deer and elk. We base our dating ot site deposits solely on the distribution of diagnostic projectile point types. The earliest occupation is marked by Cold Springs Side-notched poirts; later occupations by Cascade and Mahkin Shouldered points; and the final occupation by two Cascade varieties, Mahkin Shouldered and Nespelem Bar points. This succession of point styles in the Rufus Woods Lake project area characterizes the early to late Kartar Phase, and the association of typical Cascade projectile points with Mahkin Shouldered and Nespelem Bar points in the latest cultural level definitely places the and of site use prior to 4000 B.P., at the end of the Kartar Phase, and before the Hudnut Phase.

Excavation did not reveal any definable activity surfaces, nor any intact, bounded cultural features. Therefore, we have considered artifactual relationships primarily at the level of analytic zone. These stratigraphic divisions roughly match the identified geologic units of deposition, and appear to be temporal!" reliable, since diagnostic artifact distributions and associations of artifacts within the zones seem to retain integrity. Erosion in the active depositional environment of the alluvial fan led to deflation of surfaces, removing lighter artifacts such as bone and charcoal, if not actually redepositing heavier lithics. However, clusters of artifacts associated with exposed cobble surfaces within the zones do represent associations which we can assume to be in primary context. The lack of patterning in these areas suggests that cultural deposits are the result of frequent, short-term activities over time, and not the result of a few longterm occupations associated with the establishment of a base camp or living site. It is doubtful that 45-D0-282 ever represented anything more than a recurrent stopover, where hunting parties or task groups refurbished tool kits and cocked meals in preparation for the next day's travel.

Site 45-D0-282 is located on an extensive alluvial fan. The lowest cultural stratum lies on a coarse matrix of decomposing granite laid down either from bedrock weathering in situ or as alluvium from bedrock weathering upslope. During these initial occupations, this alluvial wash appears to have been deposited by slowly moving water. At the same time that the western portion of the site was being covered by slope wash from the bedrock behind the site, that part of the site nearer the river was receiving overbank deposition. Although the energy level for these geologic processes may have been relatively low, the earliest cultural deposits exhibit marked erosional disturbance (Zones 4 and 3). The uppermost deposits (Zones 2 and 1) are largely aeolian, and greater stability of the surface is indicated by the formation of A and B soil horizons. These uppermost zones were, however, subject to other forms of disturbance, such as seasonal flooding and historic plowing. Indeed, some site activity is best preserved in the beach collection, which constitutes secondary cultural associations on a broad erosional surface. As a consequence of repeated disturbances of site deposits over time, few primary artifact associations have been preserved, although the varying rates of alluvial deposition did provide marked temporal boundaries for constituent artifact assemblages.

Thirteen cultural teatures were defined during excavation of this site. All but one are arbitrary excavation levels identified as concentrations of basalt and granite cobbles in association with high counts of lithic artifacts and debris. The single exception is a content feature consisting of the two pestles. Artifacts in direct association with these cobble lenses appear to be primary cultural associations since the assemblages do contain a high proportion of formed tools and high densities of associated worn objects and debris. The cobbles may have acted as buffers to erosion, sealing the artifacts in loose association. The cobble surfaces themselves appear to be natural accumulations, although one may be the eroded remains of a fire hearth.

ZONE 4

The two features in Zone 4 are concentrations of basalt and granite river cobbles defined in excavation units near the present beach. Associated artifact counts were not high compared to distributions elsewhere in the zone; and, as a whole, Zone 4 produced the lowest artifact totals at the site. Functional object types consisted principally of utilized flakes and linear flakes, although a broad range of functional tool types were also present. The bulk of the assemblage consisted of a debitage of primary and secondary flakes. The recovery of four Cold Springs Side-notched projectile points, without any other diagnostic point types, suggests this zone dates to the early to mid-Kartar Phase (pre-6000 B.P.).

ZONE 3

など、このなどの意見などのなくなる。

STATES SEA

The six cultural teatures defined in Zone 3 together constitute a single basalt and granite cobble layer which contained a high proportion of formed tools and the highest density of artifacts in this zone. Among the tools associated with this layer were bifaces, microblades, retouched and utilized flakes, and choppers. Here as well excavators uncovered a "knapper's station" and a "pile of flakes." These two features (Feature 7 and Feature 11) produced 53% of the debitage and functional object types recovered from the cobble layer. Their association seems to indicate a primary cultural context preserved within the cobbies. Two lithic industries are in evidence: a generalized flake tool industry based upon the production of lamellar flakes from roughly prepared cores and a microblade industry, in which small blades were removed from delicate, carefully prepared wedge-shaped cores. Two Cascade A projectile points were recovered from this lens in direct association with two microbiades. Their occurrence in Zone 3, above the Cold Springs Side-notched points recovered from Zon 4, place these activities in the mid- to late part of the Kartar Phase, perhaps 6000-5000 B.P.

ZONE 2

Zone 2 contained two features within a cobble stratum much like those identified in Zones 3 and 4, and a single content feature of two pestles. This cobble layer did not yield high counts of artifacts, though it did contain a variety of functional object types and utilized flakes. The pestles were found in an isolated context, with no other associated artifacts. Among the projectile points recovered from Zone 2 were several stem fragments probably of shouldered lanceolate forms, and several basal fragments of unstemmed lanceolate forms. These probably correspond to Mahkin Shouldered and Cascade A projectile point types, and indicate a date in the latter part of the Kartar Phase (ca. 5000-4000 B.P.)

ZONE 1

Zone 1, the upper portion of the site deposit, contained no recognizable cultural features. Artifact counts were comparable to those of Zone 2, although no core was recovered from this zone (the only one that lacked a core). In general, historic activites--especially plowing--had greatly disturbed this layer. Recovered projectile points include Cascade A, Cascade C, Mahkin Shouldered, and Nespelem Bar, which in association, indicate activities in the late Kartar Phase (ca. 5000-4000 B.P.).

THE BEACH COLLECTION (ZONE 25)

The dense artifact scatter located just north of Area A between two large basalt erratics was thought to represent an <u>in situ</u> occupation surface only recently exposed by wave action. The area was surface collected within 1 x 1-m grid units. Artifact densities were higher than in any other area of

the site, but analysis revealed no apparent patterning in artifact locations. The lack of patterning and the occurrence of projectile point types characteristic of the entire Kartar Phase (ca. 7000-4000 B.P.), and a single point characteristic of the Coyote Creek Phase (ca. 2000-200 B.P.), indicate that this area is an eroded remnant surface comparable in age to all four excavated zones. Artifact counts were compared with those from excavated site deposits, since proportions of object and tool types could reveal site activities not represented in the zones further away from the river. The collection was found to include important additions to the total artifact inventory, such as tabular knives, projectile points, bifaces, scrapers and fire-modified rocks. The collection adds to our knowledge of site activities: hunting and butchering were done at the site, perhaps more frequently nearer the river margin; firepits were present suggesting short-term camps; and shellfish fragments (not collected) evidence use of the nearby river mussel beds tor tood. We conclude that a major portion of the site was nearer the river, and here many of the everyday living activities may have taken place. This segment of the prehistoric record, however, was ercded away by the rising water of Rufus Woods Lake.

DISCUSSION

The distributions of artifact types and the nature of artifact associations appear remarkably uniform across the four excavated zones. The manufacture of stone tools, usually from jasper and chalcedony, is the prevalent activity. These represent a range of tool forms from simple utilized flakes to carefully finished projectile points and drills, and range in size from microblades to large, blocky cobble choppers. Wear patterns on tool forms are also comparable in all zones, and indicate considerable tool use on the site as well as manufacture. Most wear is consonant with use on soft, pliable materials such as hides, meat, or woody plant parts. All site occupations appear to have occurred during the Kartar Phase (ca. 7000-4000 B.P.), with diagnostic projectile point types indicative of that entire span of time. The only indication of later occupation is the recovery of the single Columbia Corner-notched B projectile point from the beach.

Technological analysis of the artifact collection reveals a multifaceted, complex reductive strategy focused on the production of a wide range of tool forms primarily from cryptocrystalline stones. At least three industries are present: a generalized flake tool technology, in which lamellar flakes were removed from unprepared and/or prepared cores; a Levallois-like blade industry based on the removal of large blades by percussion from carefully prepared cores; and a microblade industry, in which tiny blades were removed from carefully prepared, small wedge-shaped cores. All three industries are in evidence in the earliest levels of the site and continue on into the latest site occupation.

The occurrence of the generalized flake tool industry coupled with a blade tool industry in the Kartar Phase (ca. 7000-4000 B.P.) in this project area has correlates in sites 45-D0-273 (Jaehnig 1984a) and 45-OK-11 (Lohse 1984f). The two are generally considered complementary facets of stone tool

technology during the Cascade Phase (8000-4000 B.P.) on the Columbia Plateau (cf. Leonhardy and Rice 1970: Rice 1972: Bense 1972). Their occurrence at 45-D0-282 in cultural contexts dated at ca. 7000-4000 B.P. is, therefore, not surprising, but two points are worthy of note. First, the overwhelming evidence indicates a far greater emphasis on the generalized flake tool technology. The only evidence for blade production is two large, Levalloislike blades, and a number of projectile points and other tool forms that appear to have been made on blades. No blade cores were recovered, nor were any halimarks of blade core preparation identified (cf. Muto 1976). All nonmicroblade cores are flake cores, most showing little or no preparation, and, in fact, many had been used as choppers or hammers before they were exhausted. The great majority of tool forms were made on conchoidal flakes. Second, a microblade industry is also associated with the generalized flake tool and blade industries. While we lack radiocarbon assays at 45-D0-282, diagnostic projectile points clearly document the microblade industry at ca. 7000-4000 B.P. Thus the assemblage from 45-D0-282 documents the regular use of three industries to produce flake tools, blake tools, and microblade tools during the period from 7000 to 4000 B.P. in the northern Columbia Plateau. This is coeval with the radiocarbon-dated assemblage from Ryegrass Coulee (Munsell 1968), and slightly later than the earliest dated occurrence of microblades on the Canadian Plateau, Drynoch Slide at ca. 7500 B.P. (Sanger 1968, 1970).

The microblade technology at 45-D0-282 in so early a period, may well represent a variant of the defined microblade industry characteristic of the northern Columbia Pateau. By Sangers' (1968, 1970) criteria, the linear flakes recovered from this site are microblades. However, after Sanger examined these blades he declared that they are distinct from those characterized within the "Plateau Microblade Tradition" (Sanger, personal communication 1982). For Sanger, who focuses primarily on identification of a cultural tradition and the tracing of cultural relationships, formal characteristics determine an artifact's designation as a microbiade. However, the microblades taken from this site consistently exhibit the required morphological (length, width, parallelism) and technological (core preparation, successive blade removal) characteristics. Well over 50% of the microblade specimens are non-triangular in cross section, exhibiting two or more arrises. The associated cores range from an ideal type specimen collected from the Zone 25 beach collection to several problematic examples recovered in excavated contexts, which, although wedge-shaped with a pronounced keel, exhibit less uniform blade scars.

Although these microblades and microblade cores may not fail readily within the idealized types considered characteristic of the Plateau Microblade Tradition, they are undoubtedly products of that specialized technique or a closely related technique. Small linear flakes or microblades have been recovered in Cascade Phase contexts elsewhere on the Columbia Plateau (cf. Butler 1961; Dumond 1962; Leonhardy and Rice 1970; Bense 1972), but without the characteristic microblade cores and in much smaller numbers than those recovered from 45-D0-282. An Important characteristic of this industry, which may distinguisn it from the more uniform Plateau Microblade Tradition, is the presence of multiple striking platforms on the microblade cores, a

characteristic which is rare in most microblade collections on the Columbia Plateau (Sanger 1970). This might account for the lack of uniformity in the cores and the more variable blade morphology. Thus, while they may not represent ideal specimens, they nevertheless must be considered microblades, and might be considered a variant of the idealized reduction sequence, a variant which is unusually well-represented in this site and in others in the Rufus Woods Lake project area.

Flake and blade tool forms show comparable patterns of wear--usually feathered and hinged chipping confined to a unifacial edge--, which indicates tool use on soft, pliable, and hard to elastic materials such as hides, meat and bone. The range of tool types recovered includes a preponderance of simple utilized flakes, a number of resharpened and retouched flakes. scrapers, bifaces, and a small assortment of burins, gravers, and drills, as well as numerous hammerstones and choppers. Although the use of the site was oriented toward stone tool manufacture, hunting, butchering, and attendant processing of game were also significant activities. Instances of multiple wear areas on single tool forms, and heavy attrition of working edges often entailing overlapping but distinct patterns of wear, as well as the high number of tools recovered, indicate intense activity. We cannot specify the size of prehistoric task groups, nor the duration and frequency of their visits, but we can speculate that activity was probably very short-term. ₩e base this inference on the lack of recognizable patterning in the artifact assemblage and absence of bounded activity surfaces or cultural features. Thus, the high number of tools and loose artifact associations are likely the product of many visits over the indicated 3,000 years of site use.

Economic activity probably focused on hunting of large and small game; although the faunal assemblage is sparse, identified ungulate bons evidences the taking of deer and elk. Shellfish fragments in the beach collection attest to use of nearby shellfish beds as well. The only evidence for plant collection is the two pestles recovered in direct association in Zone 2. The cultural remains are probably the product of recurrent campsites or stopovers, most likely involving small task groups, where jasper and chalcedony nodules or blanks collected elsewhere were routinely reduced into tools, and where meals of game, shellfish, and seeds or roots were processed and consumed. The virtual lack of firepits might indicate little overnight camping; however, the disturbance of cultural deposits throughout the site may have removed evidence of fires.

Interpretation of the nature of site activities and the organization of these across the site surface is difficult given active erosion of the site surface throughout much of the span of site occupation. It seems that the site deposits were repeatedly scoured by slope wash from the bedrock formations behind the site, and ephemeral stream channels which swept across the coarse, sandy surface deposits. Roden's have also disturbed the site, but this cannot account for the lack of patierning exhibited by the site deposits. It may be that most of the cultural deposits are secondary accumulations of material, and that only in those areas of distinct cobble layers, perhaps remnant natural surfaces, are cultural remains preserved in something like a primary context. If this is true, it is especially important that a form of site stratigraphy is intact--the stratigraphic-temporal distribution of diagnostic artifact types and occasional loose associations of artifacts within the defined depositional units and analytic zones. Association appears to be sound at a gross level: cultural context is preserved within the broad boundaries of the analytic zones. However, more detailed interpretations are difficult. A lack of distinct patterning in artifacts, and, thus, of definable cultural features, largely precludes assessments of group size or duration of activity. The lack of bone or other faunal remains means we cannot determine seasonality or the intensity of use of specific resources.

In terms of the cultural record of the Rufus Woods Lake project area, and perhaps for that of the northern Columbia Plateau as a whole, this site's importance lies in its large and varied artifact assemblage. This assemblage is indicative of at least three distinct processes of stone tool manufacture: a generalized flake tool technology, a Levallois-like blade technology, and a microblade technology. All three industries were evidently complementary facets of a single stone tool technology.

REFERENCES

Ahler, S.A.

- 1970 Projectile point form and function at Rodgers Shelter, Missouri. Missouri Archaeological Society, Research Series 8.
- Benfer, R.A.
 - 1967 A design for the study of archaeological characteristics. American Anthropologist 69:719-730.
- Bense, J.A.
 - 1972 The Cascade phase: a study in the effect of the altithermal on a cultural system. Unpublished Ph.D. dissertation, Department of Anthropology, Washington State University.
- Bryan, A.L.
 - 1955 An intensive archaeological reconnaissance in the northern Puget Sound region. Unpublished M.A. thesis, Department of Anthropology, University of Washington.
- Butler, B.R.
 - 1961 The Old Cordilleran culture in the Pacific Northwest. Idaho State College Museum, Occasional Papers 9.
 - 1962 Contributions to the prehistory of the Columbia Plateau. Idaho State College Museum, Occasional Papers 9.
- Campbell, S.K.
 - 1984a Archaeological Investigations at nonhabitation sites, Chief Joseph Dam Project, Washington. Office of Public Archaeology, University of Washington, Seattle.
 - 1984b Archaeological investigations at Sites 45-CK-2 and 45-CK-2A, Chief Joseph Dam Project, Washington. Office of Public Archaeology, University of Washington, Seattle.

Campbell, S.K., editor

1984c Report of burial relocation projects, Chief Joseph Dam Project, Washington. Office of Public Archaeology, University of Washington, Seattle.

Campbell, S.K., editor

1984d Research Design for the Chief Joseph Dam Cultural Resources Project. Office of Public Archaeology, University of Washington, Seattle.

Chance, D.H., and J.V. Chance

1982 Kettle Falls: 1971/1974. University of Idaho, Anthropologica: Research Manuscripts Series 69.

- Collier, D., A.E. Hudson, and A. Ford
 - 1942 Archaeology of the upper Columbia region. University of Washington, Publications in Anthropology 9(1).

Crabtree, D.E.

- 1967a Notes on experiments in flintknapping: 4. tools used for making flaked stone artifacts. Tebiwa 10:60-73.
- 1967b The flintknapper's raw materials. Tebiwa 10:8-24
- 1972 An introduction to flintworking. Idaho State University Museum, Occasional Papers 28.
- 1973 The obtuse angle as a functional edge. Tebiwa 16:46-53.

Cressman, L.S., D.L. Cole, W.A. Davis, T.M. Newman, and D.J. Scheans

- 1960 Cultural sequence at The Dalles, Oregon: a contribution to Pacific Northwest prehistory. American Philosophical Society, Transactions 50(10).
- Dalquest, W.W.
 - 1948 Mammals of Washington. University of Kansas Museum of Natural History, Publications 2.
- Daubenmire, R.F.
 - 1970 Steppe vegetation of Washington. Washington State University, Washington Agricultural Experiment Station, Technical Bulletin 62.
- Dumond, D.E.

1962 Blades and cores in Oregon. American Antiquity 27:419-424.

Dunnell, R.C.

- 1971 Systematics in prehistory. Free Press, New York.
- 1979 Archaeological potential of anthropological and scientific models of function. In Essays in honor of Irving B. Rouse, edited by R.C. Dunneli and E.S. Hall, Jr., pp. 41-73. Mouton, The Hague.
- Flenniken, J., and E. Garrison
 - 1975 Thermaily altered novaculite and stone tool manufacturing techniques. Journal of Field Archaeology 2:125-131.

- Franklin, J.F., and C.T. Dyrness
 - 1973 Natural vegetation of Oregon and Washington. U.S. Department of Agriculture, Forest Service, General Technical Report PNW-8.

Frison, G.C.

1968 A functional analysis of certain chipped stone tools. American Antiquity 33:149-155.

- Fryxell, R. and R.D. Daugherty
 - 1962 Interim report: archaeological salvage in the Lower Monumental Reservoir, Washington. Washington State University, Laboratory of Anthropology, Reports of Investigations 21.
- Grayson, D.K.
 - 1984 The paleontology of Gatecliff Shelter small mammals. In The archaeology of Monitor Valley, by D.H. Thomas. American Museum of Natural History Anthropological Papers, Vol 59 (part 1) 99-126.
- Gunn, J. and E.R. Prewitt
 - 1975 Automatic classification: projectile points from west Texas. Plains Anthropologist 20:139-149.
- Hayden, B. (editor)
 - 1979 Lithic use-wear analysis. Aczdemic Press, New York.
- Hibbert, D.M.
 - 1984 Quaternary geology and the history of the landscape along the Columbia between Chief Joseph and Grand Coulee Dams. In Summary of results, Chief Joseph Dam Cultural Resources Project, edited by M.E.W. Jaehnig and S.K. Campbell. Office of Public Archaeology, University of Washington, Seattle.
- Holmer, R.N.
 - 1978 A mathematical typology for archaic projectile points of the Eastern Great Basin. Unpublished Ph.D. dissertation, Department of Anthropology, University of Utah.
- Ingles, L.G.
 - 1965 Mammals of the Pacific states. Stanford University Press, Stanford.
- Jaehnig, M.E.W.

01

- 1983a Archaeological investigations at Site 45-OK-258, Chief Joseph Dam Reservoir, Washington. Ms. on file, U.S. Army Corps of Engineers, Seattle District.
- 1983b Chief Joseph Dam Cultural Resources Project: preliminary report of field investigations, 1978-1980. Office of Public Archaeology, University of Washington, Seattle.
- 1984a Archaeological investigations at Site 45-D0-273, Chief Joseph Dam Project. Office of Public Archaeology, University of Washington, Seattle.
- 1984b Archaeclogical investigations at Site 45-OK-18, Chief Joseph Dam Project, Washington. Office of Public Archaeology University of Washington, Seattle.
- Jaehnig, M.E.W. and S.K. Campbell (editors)

1984 Summary of results, Chief Joseph Dam Cultural Resources Project. Office of Public Archaeology, University of Washington, Seattle.

- Jermann, J.V., W.S. Dancey, R.C. Dunnell, and B. Thomas
 - 1978 Chief Joseph Dam cultural resources management plan. Ms. on file, U.S. Army Corps of Engineers, Seattle District.
- Johnson, R.J.
 - 1978 Multivariate statistical analysis in geography. London: Longwan Group Limited.
- Keeley, L.
 - 1974 Technique and methodology in microwear studies: a critical review. World Archaeology 5:323-336.
- Keeley, L.

1978 Microwear polishes on flint: some experimental results. In Lithics and subsistence: the analysis of stone tool use in prehistoric economies, edited by D. Davis. Vanderbilt University Publications in Anthropology 20.

Klecka, W.R.

- 1980 Discriminant analysis. Sage Publications, Inc. Beverly Hills.
- Lawrence, B.
 - 1968 Antiquity of large dogs in North America. Tebiwa 11(2):43-49.

Leeds, L.L., W.S. Dancey, J.V. Jermann and R.L. Lyman

- 1981 Archaeological testing at 79 prehistoric habitation sites: subsistence strategy and site distribution. Ms. on file, University of Washington, Office of Public Archaeology.
- Leonhardy, F.C.
 - 1968 An opinion on archaeological Interpretation in the Plateau. In Archaic prehistory in the Western United States, edited by C. Irwin-Williams. Eastern New Mexico University Contributions in Anthropology 1(3):27-31.
 - Leonhardy, F.C., and D.G. Rice
 - 1970 A proposed culture typology for the lower Snake River region, southeastern Washington. Northwest Anthropological Research Notes 4(1):1-29.
- Leonhardy, F.C., G. Schroedl, J.Bense, and S. Beckman
 - 1971 Wexptsnime (45GA61): preliminary report. Report to the National Park Service. Washington State University, Laboratory of Anthropology, Reports of Investigations No. 49.
- Leonhardy, F.C. and G.R. Muto
 - 1972 The Levallois technique on the Lower Snake River. Paper presented at the 1972 Northwest Anthropological Conference, Portland.
- Lohse, E.S.
 - 1984a Archaeological investigations at site 45-DO-204, Chief Joseph Dam Reservoir, Washington. Office of Public Archaeology, University of Washington, Seattle.

Lohse, E.S.

1984b Archaeological investigations at Site 45-D0-211, Chief Joseph Dam Reservoir, Washington. Office of Public Archaeology, University of Washington, Seattle.

Lohse, E.S.

- 1984c Archaeological investigations at Site 45-D0-242/243, Chief Joseph Dam Reservoir, Washington. Office of Public Archaeology, University of Washington, Seattle.
- Lohse, E.S.
 - 1984d Archaeological investigations at Site 45-D0-282, Chief Joseph Dam Reservoir, Washington. Office of Public Archaeology, University of Washington, Seattle.
- Lohse, E.S.
 - 1984e Archaeological investigations at Site 45-DO-326, Chief Joseph Dam Reservoir, Washington. Office of Public Archaeology, University of Washington, Seattle.
- Lohse, E.S.
 - 1984f Archaeological investigations at Site 45-0K-11, Chief Joseph Dam Reservoir, Washington. Office of Public Archaeology, University of Washington, Seattle.
- Lohse, E.S.
 - 1984g Rufus Woods Lake Projectile Point Chronology. In Summary of results, Chief Joseph Dam Cultural Resources Project, edited by M.E.W. Jaehnig and S.K. Campbell. Office of Public Archaeology, University of Washington, Seattle.
- Lyman, R.L.
 - 1976 Exploratory archaeological research along Rufus Woods Lake, upper Columbia River region, north central Washington, 1975. Washington State University, Washington Archaeological Research Center, Project Reports 29.
- Maser, C., and R.M. Storm
 - 1970 A Key to the Microtinae of the Pacific Northwest. Corvallis, Oregon State University Book Stores.
- Miss, C.J.
 - 1984a Archaeological investigations at Site 45-D0-214, Chief Joseph Dam Reservoir, Washington. Office of Public Archaeology, University of Washington, Seattle.
 - 1984b Archaeological investigations at Site 45-D0-285, Chief Joseph Dam Reservoir, Washington. Office of Public Archaeology, University of Washington, Seattle.

- 1984c Archaeological investigations at Site 45-OK-250/4, Chief Joseph Dam Reservoir, Washington. Office of Public Archaeology, University of Washington, Seattle.
- 1984d Archaeological investigations at Site 45-0K-287/288, Chief Joseph Dam Reservoir, Washington. Office of Public Archaeology, University of Washington, Seattle.
- Munsell, D.A.
 - 1968 The Ryegrass Coulee site. Unpublished M.A. thesis, University of Washington, Department of Anthropology.
- Muto, G.R.
 - 1971 A technological analysis of the early stages in the manufacture of iithic artifacts. Unpublished M.A. thesis, Department of Anthropology, Idaho State University.
 - 1976 The Cascade technique: an examination of a Levallois-like reduction system in early Snake River prehistory. Unpublished Ph.D. dissertation, Department of Anthropology, Washington State University.
- Nelson, C.M.
 - 1969 The Sunset Creek site (45-KT-28) and its place in Plateau prehistory. Washington State University, Laboratory of Anthropology, Reports of Investigations 47.
- Nie, N.H., C.H. Hull, J.G. Jenkins, K. Steinbrenner, and D.H. Bent 1975 SPSS: statistical package for the social sciences (second ed.). Mcgraw-Hill, New York.

Odell, G.H.

- 1977 The application of micro-wear analysis to the lithic component of an entire prehistoric settlement: methods, problems and functional reconstructions. Unpublished Ph.D. dissertation, Department of Anthropology, Harvard University.
- Piper, C.
 - 1906 Flora of the state of Washington. In Contributions to the U.S. National Herbarium (Vol. 11). U.S. Government Printing Office, Washington, D.C.
- Post, R.H.
 - 1938 The subsistence quest. In The Sinkaletk or Southern Okanogan of Washington, edited by L. Spier, pp. 9-34. George Banta, General Series in Anthropology 6.
- Ray, V.F.
 - 1932 The Sampoll and Nespelem: Sailsham peoples of northeast Washington. University of Washington, Publications in Anthropology 5.

Rice, D.G.

1972 The Windust phase in lower Snake River prehistory. Washington State University, Laboratory of Anthropology, Reports of Investigations 50.

Sanger, D.

- 1968 Prepared core and blade traditions in the Pacific Northwest. Arctic Anthropology 5:92-120.
- 1970 Midlatitude core and blade traditions. Arctic Anthropology 7:106-114.
- Semenov, S.A.
 - 1964 Prehistoric technology: an experimental study of the oldest tools and artifacts from traces of manufacture and wear (translated by M.M. Thompson). Cory, Adams & Mackay, London.
- Shiner, J.L.
 - 1961 The McNary Reservoir: a study in Plateau archaeology. River Basin Surveys Papers 23. Bureau of American Ethnology, Bulletin 179: 149-266.
- Smith, L., and A. Goodyear
 - 1976 The statistical discrimination of hard and soft percussors. Ms. in hands of authors.
- Speth, J.

- 1972 Mechanical basis of percussion flaking. American Antiquity 37:34-60.
- Stafford, B.D.
 - 1977 Burin manufacture and utilization; an experimental study. Journal of Field Archaeology 4:235-246.
- Stafford, B.D., and C.R. Stafford
 - 1979 A comparison of macro- and microwear functional analyses of lithics. Paper presented at the 44th Annual Meeting of the Society for American Archaeology, Vancouver, B.C.
- Stebbins, R.C.
 - 1966 **A field guide to western reptiles and amphibians.** Houghton Mifflin, Boston.
- Swanson, E.H., Jr.
 - 1962 The emergence of Plateau Culture. Idaho State College Museum, Occasional Papers 8.

Swanson, E.H., Jr. (editor)

- 1975 Lithic technology, making and using stone tools. Mouton, The Haque.
- Szuter, C.R.
 - 1982 The interpretation of rodent bones from Huhokam sites. Paper presented at the 47th Annual Meeting of the Society for American Archaeology, Minneapolls.

- Taylor, W.E. (Jr.)
 - 1962 A distinction between blades and microblades in the American Arctic. American Antiquity 27:425-426.
- Thomas, B., L.L. Larson, and M.G. Hawkes
 - 1984 Archaeological investigations at 30 historic sites, Chief Joseph Dam Reservoir, Washington. Office of Public Archaeology, University of Washington, Seattle.
- Wilmsen, E.N.
 - 1968 Functional analysis of flaked stone artifacts. American Antiquity 33:156-161.

......

1970 Lithic analysis and cultural inference: a paleo-indian case. University of Arizona, Anthropological Papers 16. APPENDIX A

RESULTS OF SOIL ANALYSES, 45-DO-282

.

Table A-1. Results of physical and chemical coll analysis, Column 1, 45-D0-282.

				Phy at	cal An	lyses					5	cal Anelyses	
Seept e	8		Particle Size			Const	ltuents						
5	Surface	(dry)	Sand/Sil t/Cl ey [X]	Chercoel (X)	4ªY (S)	Bone (X)	S)	Organic Matter (X)	Mi nerel e (X)	ł	Organic Matter [X]	Exchangeable Cal cium (ppm)	Phosphate (ppm)
-	\$9 0	10YR[4/2]	72/23/5	trace		trace	•	-	92+	8-15	15	3920	, R
CN	6-10	1078[5/1]- 40/8[5/3]	70/25/5	tr aca	•	1	ł	сч	+28	8.65	2	4060	80.5
m	14-24	10YR[42]	72/23/5	trece	ı	ı	•	traca	+88	8.43		3605	74.0
-	27-33	10YR[5/2]	5/12/J1	trace	•	ı	ı	trace	+68	8.25	trace	3710	80.8
10	37-47	10YR[6/2]	72/25/3	trace	•	•	1	trace	•68	0.10	tr ece	3890	63.0
•	47-58	10YR(6/2)	77/20/3	trace	ı	ı	1	trace	+68	8.05	trace	3920	57.4
~	58-68	10YR(6/2)	77/10/5	trace	1	4	ı	-	+96	7.70	ł	3820	58.7
-	74-84	10YH[5/2]	75/20/5	trace	•	ı	,	trace	+88	7,55	trace	3395	57.4
	84-94	10YR[5/2]	72/20/8	ı	1	•	•	trace	+68	7.75	ı	3385	49.7
	84-100	10YR[5/2]	70/25/5	•	۰,	ı	۱	trace	+86	7.80	Lrace	3500	44.1
-	103-113	10YR[7/2]	60/30/10	ı	36. 3	trace	trace	trece	64+	7.70	trace	4270	1
ũ	113-123	10YE(7/2)	67/23/10	۱	501	۱	ı	trace	+8-	7.60		4270	ı
13	130-140	10YR[6/3]	PK/10/5	ı	m	ı	i	trace	+98	7.75	۱	4375	39.2
-	140-148	10YR[8/3]	87/10/3	•	٠	ŀ	ł	trace	+68	7.80	•	4375	63.0

Results of physical and chemical soil analysis, Column 2, 45-D0-282. Table A-2.

41

				Phy st	al Andly						8	ical Analyses	
8 m.pl.	5		Particle Size			Const	tuente						
i.		Col or (dry)	Sand/Silt/Clay (X)	Charcoal (X)	Ash (S)	Bone (X)	Shell (X)	Organic Natter (X)	Hinerel = [X]	Æ	Organic Nattar (X)	Exchangeable Calcium (ppm)	Phosphate [ppm]
-	0-8+	107F[5/3]	67/10/3	trace		1	1	-	• 86	°×¥ ₩	M	1750	74.2
a	+-4	10YR[4/2]	B2/15/3	trace	i	1	I	-	•86	AN	AN	845	74.2
•	19-29	10YR[5/2]	82/13/5	trace	Lrece	ŀ	1	-	+88	N A	K N	2730	81.2
4	29-38	10YR[5/2]	60/15/5	·	trace	1	ı	trace	+98	AN	¥ N	2875	77.0
ĸ	46-65	10YA[6/2]	BQ/20/-	١	·	ł	ı	traca	•88	AA	A N	3045	76.3
•	55-85	10YA[6/2]	62/15/3	trrce	ł	ı	•	trace	•88	Ă	AN	3165	68.6
~	5 5-72	10YR(6/2)	82/15/3	ı	ı	trace	,	trace	+68	A N	A N	3080	67.8
•	78-88	10YR[E/2]	65/12/3	ı	•	ı	٠	trace	+86	AN	, AN	3080	68.6
•	88-88	10YR[6/2]	86/12/3	ı	1	ı	ı	trace	+68	A M	MA	3090	69.6
5	88-10 8	107R[6/2]	86/12/3	,	۱	trace	;	trace	+88	N A	A M	3080	73.5
:	108-116	10YR(6/2)	86/12/3		ł	ı	ı	trace	•88	A M	K N	3010	72.1
4	118-128	10YR[6/2]	80/15/5	ı	١	trace	ł	trace	+68	AA	N A	2800	74.2
13	133-143	10YR (8/2)	86/12/3	trace	1	ı	Erece	Prece	+88	N A	N A	3010	72.1
7	145-152	10YR[6/2]	66/12/3	ı	ı	,	·	trace	•88	AN	N A	3115	70.0
15	158-168	sel t/pepper	85/12/3	ł	ı	trace	trace	trace	+68	K N	N A	3185	66.5
=	169-179	Bel L/pepper	87/13/-	,	ı	trace	trace	trace	188	AN	N A	3165	57.4
1	179-189	sal L/pepper	90/10/-	•	•	١	ı	trace	+88	A M	N A	3185	63.0
=	100-199	sel L/pepper	82/18/-	1	trace	ı	•	trace	+86	M	N N	2835	66.5

Possibly Maxama seh. Pot messured.

 $\label{eq:relation} I = 0.25 \mpose \mpos \mpose \mpose \mpose \mpose \mpose \mpose \mpose \mpose \mpose \mpos \mpose \mpose \mpose \mpose \mpose \mpose \mpose \$

					Physic.	of Analys						cal And vere	
Sept.	8		Particle Size			e S	ti tuenta						
	Burteca	[ary]	Send/Sit L/Ci ey [3]	Chercoet (S)	s.	Bone (X)	8) (X) (X)	Organic Natter (S)	Minaral a (X)	£.	Organic Matter [3]	Exchangeable Cat cium form 1	Phosphata [ppm]
- 0	+13-+10	10YR[4/3]	87/10/3	trace		trece	1	-	8	5.20	61	364	
			66/12/3	I.	ł	ł	ı	C 4	86	6.50	tr nce		
•		(c/a)41/14		trace	trace	trace	ı	trace	•88	8.80	, ,	245	72.4
-	2-2				•	ı	trace	trace	+68	6.95	٠	1	102
	3	10/9 June	6/33///	TT DC D	•	ı	ŀ	traca	+65	6.65	۱	480	2
~	40-52	10/01/01	6/C1/30	trace.		ı	ı	trace	•88	6.85	4	284	74.9
-	57-67	10YB[5/3]			1	•	•	trace	+88	6.90	ł	567	74.2
. 4	18-12	10/2/2/01		1	•		ı	trace	• 88	6.95	•	294	73.5
2	59-16	10YR[5/2]	-/22/11			·	ł	trace	+ 88	0.0	ı	518	9.47
=	101-111	10YB[5/2]	60/35/K			ı	1	trace	+ 68	6.30	:	1330	74.9
12	111-116	10YB15/2	65 /30 /3			ì	ı	trace	+88	8.65	ı	1750	77.0
1	121-128	10YB'5/2]	2/10/12			1	1	trace	+68	6.70	ı	1925	1.11
1	134-144	10YR[5/2]	M0/15/5			ı	ı	trace	+88	6,90	ı	1750	8.62
	161-160	anute (n)				•	•	۱	58•	7.20	ı	2450	78.1
2	noi - 101	[>/c]H101	B/1/3	•	ï	•	ı	ı	100	7.30	,	756	1.11

¹Possibly Nazma ash.

<u>د د</u>

٠, 1.1.1.1.1.1

1

۰,

Table A-5. Resuits of physical and chemical soil analysis,Column 5,45-D0-282.

				Phy et c	al Anal	y (5	cel Anelysne	
Smple	8 3		Particle Size			Con	ti tuente						
i		Color (dry]	Bend/Silt/Ct ay (S)	Charcoal (X)	4 X	Bone [%]	8hei i (3)	Organic Matter (X)	M ner el e (X)	Ŧ	Organic Matter (X)	Exchangesbl s Cal cius (pps)	Phosphete [ppm]
-	+10-0	10YR[3/2]	76/20/8	trace	troco	1	1	2	87+	6.80	trace	1750	78.4
04	I	10YR[4/3]	65/30/5	trece	1	trace	ı	3	96+	7.00	Lr ace	23 60	78.4
-	10-20	10YR[3/3]	65/32/3	trace	trace	trace	1	~	97.5	7.10	trece.	2730	0.11
4	9 - Q	10YR(3/3)	65/30/5	tr ace	ı	trace	ı	•	- 86	04.7	trace	2415	60.5
2	28-38	10YR(4/2)	65/32/3	۱	•	trace	ı	trace	+66	7.50	1	2520	78.1
•	3-3	10YR[5/3]	65/32/3	•	ł	trace	trace	trace	+68	7.65	traca	2870	80.5
~	9 23	10YR(5/3)	67/28/5	•	1	trace	ı	trace	+68	2.85	ı	2695	1.62
•	8 3-7 3	10YR[5/3]	67/30/3	٠	ı	-	trace	trace	9 8•	7.70	trace	2800	79.1
æ	23-62	10YA(5/3)	72/25/3	trece.	ı	trace	trace	trace	•66	8.00	trace	30.60	81.8
10	8 9-5 9	10YR[5/3]	67/28/5	trace	1	trace	trace	trace	+68	7.90	trace	3045	78.4
:	86-105	10YR[5/2]	75/20/5	ı	•	trace	trace	trace	+ 69	8 .00	ı	3185	80.5
42	112-122	10YR[7/1]	77/20/3	F	ı	trace	•	trace	+66	8.25	1	3150	76.3
13	122-130	10YR[7/1]	5/02/11	i	•	i	ı	trace	+66	8.30	ı	2905	77.0

APPENDIX B

4

/

ARTIFACT ASSEMBLAGE, 45-DO-282

از از مربع بر مربع از مربع از بر از مربع از مر از مربع از مربع

Table B-1. Object edge angle by artifact type and zone, 45-00-282¹.

1

۲

े? © 1

×,

1.1

	Haret-Hore	burface Miec.	· · ·	•		•					• • • •	Ne er () 1	•••	•		•	1 1 1 1	•	•
		23 23 23	•	•	· ·	•			e •		, , , , , , , ,	••••••••••••••••••••••••••••••••••••••	•••		 	•	• • •	•	• • •
		58 23	-	•		•			· ·	 . . .	 	· · · · ·	 	•		, ,	 	•	· ·
		61 58 70 7		04		1	, e		-	 . .	• • •			•		-	, , ,		
	tdge angt e (degr	22	· · · · ·	-	•••	~	•••			• • ·	. .	 	1 1 1 05	•	· · · · · · · · · · · · · · · · · · ·	, ~u	•••	•	•••
		159 59		°s ••	• •	.	•	-	• •	•••	·	. -	. . .	•	s i Mii i Serer i Ca	-	 	•	- 1 1
	•	83 88 88 83 83	· · ·	•	· •	•	••••	•••	• •	••••	· · ·	· · · · ·	, , , ,			•	<i>.</i> <i>.</i>	•	· ·
• • • • • • • • • • • • • • • • • • •		11 11 11 11 11 11 11 11 11 11 11 11 11		•	 	1 1	 		, , , ,	1 86 1	 	• • • • • • • • • •	• •	1 1 1	• • • • • •	i 1	•••	1 1	

;

.

Table B-1. Cont'd.

Ľ,

	Fraction			ļ						Edge	e jõue	ldegre	(:										
		• 2	::	28	22	88	58	28	22	*8	58	38	52			28	58		58	1×	Burface	M1.6C.	Total
	terpening flake							1			•	1		1	4	-	-	-		4]		
		1	ı	•	٠	•	•	•	٠	-	-	,	-		•	,				•			•
		ı	•	•	•	ı	•	•	•	•			• •	-	•			,	,				~ 0
		1	•	•	•	•		•	·	•	•			• •	• •							•	- 15
		•	ı	•	•	•	ŀ	•	٠	•	•	•	•	• •	•								
		۱	•	ı	•	•	•	•	•	•	•	•	-	•		•		•	,	٠	ł	• •	
	ciatiy mehad fiata																						,
		ı	•	ı	•	•	•	0	•	a	ı												
		•	٠	٠	•	•	• •	• •		• •	a		•			• •					•	,	•
		٠	١	-	•	•	•	9	•	• •	• •	-				-					1	ı	•
		•	•	• •	-	•				, ,	• •	• •	•	0	•				•	•	•	•	2
	ect of Ly						-		•			•	•	•	-	,			•	•	•	•	•
	ouched flake																						
		1	•	••	•	ı	•	•		-		•	•	•		-					,		•
		١	•	-	-	-	-	٠	~	- 04	-	-	-	- 64		• •	Ŧ						• 8
		1	۱	1	•	-	-	•	Q,	-		24	0	-	• •	-	• .						::
		•	ł	•	•	-	-	-	•	-	•	1	•	-	•	•				•	ı		1
	irettes solu	1	ı	1	•	-	-	•	-	-	84	•	•	•	-	•	-	·		•	•	,	3
		•	;	2	8	8		:	•			•	4										
		•	•		2	29	•		• •	= *	• •	•	~ •	6 4	- 4	ı			-		٠		2
		•	2	3	8	2	1	6	• =	1	: 2	h a	• •	9 e	u a	• •		• •	•		1	•	2
		-	~	•	-	-	~	-	·~	-		0	•		• •	• •	• •	. .				•	
		•	#	ŧ	7	=	2	24	25	=	2	-	•	~	- 10	-	•	,		•		. 1	
	otal a	1	1	;	;																		2
			5.	2	ຊ	8	•	2:	2		•	:	•	•	•	4	•	•	-	•			178
			•	2 3	58	= 1	5	23	- 1	<u>ب</u>	.	2:	2	•	•	-	~		•	•	-	~	ž
		•	ě .	5;	5 -	9 •	2	5,	6 3		2.	21	2	2	-	04	~	-	,	-	:	-	800
		•		: 3	° %	6	8	• 2	28		• 1	~ g	~ ;	-:	• ;	•					• :	•	3
LG 28 64 154 111 101 79 64 60 74 51 72 78 28 28 19 10 1 2 6 28 3 1.41			!	;	:	:	:	:	;	:	,	 	3	-		•	٥		-	-	10	،	2
	5	8	3	1	111	101		1	8	74	ĩ	25		8	2	;	2	-	•		28	-	1

¹Hor-Lithis meterials deleted.
Table B-2. Distribution of tool types by zone, 45-D0-282.

	·········			•								Tat		<u> </u>
Busch type	Location of mean					1	•						i	
		1.8	11.42	l II	11/45		TVA	3	TVAS	1 1 1	TVAS	r.	TL/AL	
Utilized floke			• • •	814	•	• 221	•			12.04	•	- 78K		—
Canal China	Edge enty	1		-	0.0.5			, '		ľ	•	-	0.1/0.1	
I	unifactal adda	-	2.2/1.7	4	4.2/2.7	•	2.8/1.8	,		-	0.4/0.2	15	1.8/1.3	
	Bifactel edge	١	•	·	;	•	1.8/1.3	1		•	1.2/0.7	~	1.0/0.1	
	Point andy	. •				-	0.4/0.3			-	0.4/0.2	•	0.2/0.2	
	Point only			••						• -	0.4/0.2		1.0/1.0	
Feethered chipping	Unifacted edge	2'	5.32.5	£	62. b/ 40.8	165	50 . 3/ 30 . P	1	71.0/50.0	105	6. 96/6. 19	528	66.4.46.0	
		`;	/	e 8	4.2/2.4	23	5.1/4.2		1.1/3.1		0.7/5.2	4	6.0/4.0	
	Bifacted edge	-	0.7/0.6	. .	1.2/2.1	; ¬		<u>.</u> -	4.6/8.4	4	1.6/1.0	20	2.2/1.5	
	Point anly	•	4	•	,	•				-	0.40.2	-	0.1/0.1	
Unifectatly retouched	r l aire	ł		24		CI 4				# 22		84 7 8		
215 24	Unifacial adge	•	,	-	4.8/0.5	04	15.4/0.6	•	•	' :	•		3.0/0.2	
	Bifacial adga	•		•		-	B.0/0.3	•	1	• •		-	1.3/0.1	
	Voint anly Voint anly					•				- •	4.5/0.2	- {	1.3/0.1	
	Difactal adam	- ,		•				<u>,</u>	A.17.4.11.4	0 •		s °	0.5/6.62	
	Point andy	•		~	9.5/10.9	- ,		•		• .		• •	2.0/0.2	
Ninged chipping	Unifactet adge	~	8,5/2,1	= .	52.4/6.0	•	81.5/2.6	•	20.6/4.5	2	63.63.5	\$. 5. 8	
	Point only		,				, ,			• -	4.5/0.2		1.0/2.1	
Bifacial tr ratewohad	f Late	1		4		4 12		1		3				
Prosting	Edge enty		•	-	18.7/2.5			• •		• •		5-	2.8/0.3	
	Unifactet edge				16.7/0.5	-	E.0/E.8	,		'	•	~	5.0/0.2	
	Bifectel edge		12.5/0.5	- 1	16.7/0.5			•		~	2.00.5	•	11.8/0.3	
Feethered chipping	Unifacted adge	• •	50.02.2	4	ı		8. 67. 0		• •	•	25.0/0.5	- 04	28.5/0.0	
	Bifacial adge	ı				~	16.7/0.6	ı	,	•	•	~	5.8/0.2	
	Unifacted edge Bifactel edge	• •	25.0/1.1			n n	25.0/1.0 25.0/1.0	· •		• •	50.0/1.0	ð u	29.4/0.8	<u> </u>
Reference field		<u>/</u> #		*		ī		ž						
Berooth I ng	Unifacted edge		•	-	50.0/0.5	• •	ł	:		2.			6.7/0.1	
Fouthered chipping Ningod chipping	Unifacial adge Unifacial adge	• ••	57.1/2.2	. 🕶	50.0/0.5	• 🕶	100.0/0.3	e 1	100.0/1.1	• •	100.0/1.0	1 1 G	33,3/0,4 60,0/0.0	
M creix ede		4		°. ≢		ī		1						
Factored chipping Minged chipping	Unifacial adge Unifacial adge	cu i	1.1/0.01.1	Cu i	1.00.001	•	100.0/0.3	• •		~	100.0/0.5		87.5/0.5 14.3/0.1	
	,													_

•

Table B-2. Conted.

.

33

						- 1 4 1 1	•.	:	1				
												Ē	otal
	Lecation of wear		-		~			-			8		
		1	N/L	8	11/41	ĩ	11/45		N/N	3	TVAS	Fr B	11/4
Blass Feathered chipping	Unifectat adge	1 .		1	• • •	∾- ±	50.00.3			0 i		1	1.0.0.00
Tebeter Inire	unitaciel edge	, o ‡	•	· •	•	- c	50.0/0.3	' 9	1		•		1.0/0.09
	Edge enty	•			•) }	,	2		•	100.0/2.2	[-	100.0/0.8
Scraw Booking	Edge entry	1.		. '		1	٠	*	,	20- #	2.7/0.2	e/+	1.0/0.1
	Unifacial adge Bifacial adge	- 1	12.5/0.3	•	25.0/1.1	- 1	5.5/0.3		۰.	•	8.7/0.2		4.1/0.2 2.7/0.2
Total in patrix	Faint acty Faint acty Unificcial ada	•••		- : •-	12.5/0.5 50.0/2.2 12.5/0.5	۰ . ب -		• • • •	1.10.06	• - g ·	21.6/2.0 2.7/0.2 70.3/6.5	•-31	1.24.0.1
	•	I						1				•	
	Eage muly Matterial adapt	-	16.7/0.0			•	•	 t		Ż	•	-	1.7/0.1
	Bifactal ada				9.0/0.31	• ••	20.0/0.0		100.0/1.1	• •	12.1/1.0		1.7/0.1
Festioned thipping	Point only Unifectal odge	F	16.7/0.8	e 1	12.5/0.5		• •	• •	1 1	• • •			1.7/2.1
Minged an iselan	Bifacial adge Unifacial adge	n -	50.0/1.7	•	2.0/1.1	• •		, ,		•4;	12.1/1.0	•~;	12.1/0.6
	Bifactet edge	•	-	• •		•	5			-	12.1/1.0	.	19.00.7
Burta		i		1		ī		24		o ±		4	
	Unifacial adda	• -	100.0/0.6	. •	25.0/0.5	• •	100.0/0.3	• •		• •	a 1	- 0	12, L/U.1
Hinged Chipping	Unifected adge Bifacial adge	• •	• •	~	80.2/1.1 25.0/0.5	• 1	1 1	CH I	100.0/2.3		• •		0.00.9
Drill		ł		1		1				1			
	Unifected ada Point aniv		۰.			• •			1.1.1.1.1	9 1 2	٠		1.0/0.1
factored things	Unifectal ador				20.0/0.5	• • •	25.0/0.3	- 1	1.1/10.00	• 🖛	16.7/0.2	~ 4	11.60.2
	Paint any	ł	1		20.0/0.5	- 1	r.n.n. e	• .	• 1) Ca	.0/0.5	01 FI	11.8/0.2
	Petre only	• • • • 1	• •		20.0/0.5	1 🕶	29.0/0.1	, ,		m +	50.00.7	∢ №	20.5/0.2
Poot In.	Point andy	;	ı		100.0/4.5	n -	20,0/0.3	ł	,		,	•• • •	
Fee thered chipping Minand chipping	Cifacial adge Unifacial adge		• •	• •			20.00.3	•		'	í		1.0/2.01
	Bifacial adga Point anty	• •		1.1	۰.								16.7/0.1
						•	5. JU. V.	ı	•			-	18.7/0.1

dins

.

2

• ł

I 4

Table B-2. Contid.

ر مردی هد. مدر مردی

د. در در در

THE REAL PROPERTY.

9

. м

)

恋

01

Direct yee 1 2 3 4 26 Direct yee free 17/13 free 17/13 free 17/13 free 17/13 Object yee free 17/13 free 17/13 free 17/13 free 17/13 Object yee free 17/13 free 17/13 free 17/13 free 17/13 Constrait was Intervie Mark was Intervie Mark was Mark was 11/13 Mark was							70	2					Ŧ	1
Number of the sector of the	04-05 Systems	Location of wear				~	•		-			£		
Value (1) Main			Fra	TUAL	Ĩ	TL/AL	Freq	N.V.	Free	TA/AS	Fr B	TA/AT	Fred	TL/M
Paratring Paratring <t< td=""><td>rajectite point</td><td></td><td>ł</td><td></td><td>*</td><td></td><td>Ŧ</td><td></td><td>ž</td><td></td><td>-</td><td></td><td>1</td><td></td></t<>	rajectite point		ł		*		Ŧ		ž		-		1	
Function and failing Function	Be ad thing	Bifaciat adga	'	,	•	•	•	•		•	1	20.0/0.5	• •	WC 66
Hinde (a) (with (a)) Minit (a)) <	Farmered Birging	Unifacial adja	•		'	•	•	•	•	•	-	12.5/0.2		
Miner Alger Unificatina dela I <thi< th=""> <thi< td="" th<=""><td></td><td>Paint anty</td><td>•</td><td>ł</td><td>1</td><td>,</td><td>-</td><td>100.0/0.3</td><td>'</td><td></td><td>•</td><td></td><td>•</td><td></td></thi<></thi<>		Paint anty	•	ł	1	,	-	100.0/0.3	'		•		•	
Nitreating Anti-active Anti-act	Electo vers	Unifacial edge	,	•	،	•	•	•	•	•	04	25.0/0.5	• •	22.2/0
Print and above ti where a strain where a s		Bifacial adge	,	•	•	•	,	•	•	•	~	22.0/0.5	0	22.20
Addression Model		Paint any	,	ł	•	1	•	ł	٠	•	•	12.5/0.2	-	11.1/0
	relection and at the		1		1		9		5		3			
				,	-	311/05	2 '	1	2	1	È		Ĩ	
Markan biling	- Horad alian	Unifacial ador	•	•							. 6		- 0	
Hinger chiere Unification Image chiere Image chiere <th< td=""><td></td><td>Bifactal adom</td><td>•</td><td>•</td><td>•</td><td>33.370.5</td><td>J</td><td></td><td></td><td></td><td>• •</td><td></td><td>~ ~</td><td></td></th<>		Bifactal adom	•	•	•	33.370.5	J				• •		~ ~	
Interint with Interinterint with Interint Interint	Minced chipping	Unifacial ador	•	,	• •		•		•	. 1		0 0/0 00	••	
Operation bestation built enty towarise figs enty built enty figs enty built		Bifacial adaa	-	100.0/0.6	-	33.3/0.5	•	,	ı	ı	• •	28.6/0.5	- 4	36.4/0.
Marking Effective weak T <tht< th=""> <tht< th=""> T</tht<></tht<>			1		3		:		4 :		!			
Maint only Conside 1 <		Attack of a second	2		2		ľ				11			
Creating Educating Educat			•		•		•		ı	•	N 1	18.7/0.5	â	14.2/0.
Marken being wirfericity Eige with File.70.5 E 15.70.5 E 17.70.5 E 17.70.5 E 15.70.2 E					ı		•	1	ı	1		0.3/0.5	-	2.2
Marries Eigenity Ferritis F. 200.0/0.6 F F F. 200.2 T 7.1/0.2 T <td< td=""><td></td><td>the fract of adda</td><td>•</td><td></td><td>ı</td><td>•</td><td>ı</td><td>•</td><td>ı</td><td>•</td><td>Cu -</td><td>16.7/0.5</td><td>Q</td><td>14.3/0.</td></td<>		the fract of adda	•		ı	•	ı	•	ı	•	Cu -	16.7/0.5	Q	14.3/0.
Maintening Maintening <td></td> <td></td> <td></td> <td>ı</td> <td>•</td> <td></td> <td></td> <td></td> <td>1</td> <td>ı</td> <td>-</td> <td>8.3/0.2</td> <td>-</td> <td>2.2</td>				ı	•				1	ı	-	8.3/0.2	-	2.2
Cutation Edge only H=0 H=12 L=1.0.13 H=4 H=4 H=12 L=1.0.12 H=13 L=1.0.12 L=1.00.0.0.0.12 L=1.0.12 L=1.0.12					•	•		100.0/0.0	1		0	41.7/1.2	~	0/0.00
Manual Lange Hend		Laur auth	•	•	•	•	•	ı	I	•	-	8.30.2	-	7.16.
Grunding Eigenity Eigenitient Eigenity Eigenitient Eigenity Eigenity Eigenity Eigenity Cumbing Tenning unfected wdp Tenning unfected wdp </td <td></td> <td></td> <td>ł</td> <td></td> <td>ŧ</td> <td></td> <td>H-12</td> <td></td> <td>4</td> <td></td> <td>112</td> <td></td> <td>Me 20</td> <td></td>			ł		ŧ		H-12		4		112		Me 20	
Mitacial edge 1<	Gruthing	Edge enty	•	•	•	ı	-	6.0/E.8		1	; I	•	-	3.6/0.
Point anty - - - - - 1 0.3012 1 0.5012 <th1< th=""> 1 <th1< th=""></th1<></th1<>		Bifacial adge	•	1	•	•	٠	•	,	•	-	8.3/0.2	-	3.6/0.
Burfless Terminest surfless Terminest surfle		Point enly	•	•	٠	•	·	•	,	•	-	8.3/0.2	-	5.6
10 minute ourface - - 11 81.772.6 24 86.772.2 24 86.772.2 0.416 100.074.6 100.074.6 100.074.6 100.074.6 100.074.6 100.070.1 0.416 100.071.6 - 3 100.071.6 - 3 100.070.1 0.401 100.071.6 - 3 100.071.6 - - 3 100.070.1 0.401 100.071.6 - - 3 100.070.5 - - 3 3 0.401 11 100.070.6 - - 3 100.070.2 2 3 3 0.401 11 100.070.5 - - - - - 3 3 11 100.070.5 - - - - - - 3 3 3 11 100.070.5 - - - - - - 3 3 3 11 100.070.5 - - - - - - 3 3 3 11 100.070.5 - - - - - - 1 3 3 3 11 <td></td> <td>Burface</td> <td>ı</td> <td>1</td> <td>•</td> <td>•</td> <td>•</td> <td>•</td> <td>ł</td> <td>۱</td> <td>-</td> <td>6.3/0.2</td> <td>-</td> <td>3.6/0.</td>		Burface	ı	1	•	•	•	•	ł	۱	-	6.3/0.2	-	3.6/0.
Hile Helle Taminak burfaca Helle H		Terstrat Burface	•	ſ	•		=	BI.7/3.5	4	100.0/4.5	•	N. U.2.2	54	8.7/2.
Cumbing Tensinal purfece 3 100.0/1.6 3 100.0/1.6	wile		ł		ž		0		04		140			
ere Franking Perintenty Het	C-ushing	Termined ourface	•	ı	•	100.0/1.6	•	•		i	••	٠		100. e/e.
Greeking Portstenty	e . e		Ŧ		ī		D ž		0		Ĩ		5.4	
01AL 179 184 309 68 43 44	Crushing Kinged chipping	Point enly Unifectal edge		100.0/0.6	• •	100.0/0.5	4 1	۰.1		۰ı	•	100.0/0.2	- 0	33.3/0.
	7 A.		178		181		906		89		cu.		1 164	

The Court of tools with operified user stributes. Stra - Court of tools with operified user stributes. St - Percontage of the zonel memollage.



Figure 8-1. Projectile point outlines from digitized measurements, 45-D0-282. Upper number is the historic type (see Figure 3-7 for key). Lower number is master number.

cm.

A CANADA AND A CANADA

APPENDIX C

FAUNAL ASSEMBLAGE, 45-DO-282

Family Laporidae

Sylvilacus of nuttailii

Zone 1: 2 metapodials, 5 phalanges, 1 tibla fragment.

Zone 2: 1 calcaneus.

Family Sciuridae

Marmota flaviventris

Zone 2: 1 ulna fragment, 1 calcaneus.

Zone 3: 1 molar, 1 Incisor fragment.

Zone 25: 1 mandible, 1 mandible fragment, 1 humerus fragment, 1 radius fragment, 1 ulna fragment.

Family Geomyidae

Thomomys talpoides

- Zone 1: 2 mandibles, 3 mandible fragments, 1 humerus fragment, 2 innominate fragments, 4 maxilla fragments, 1 skuli fragment, 1 scapula, 1 tibia.
- Zone 2: 2 skulls, 7 skull fragments 6 mandibles, 7 mandible fragments, 4 scapulae, 2 humeri, 3 humerus fragments, 1 ulna fragment, 1 pelvis, 1 innominate, 1 saccrum, 2 femurs, 2 femur fragments, 1 tibla.
- Zone 3: 4 skulls, 11 skull fragments, 15 mandibles, 21 mandible fragments, 9 humeri, 3 humerus fragments, 3 ulnas, 1 ulna fragment, 2 radius, 7 scapulae, 7 innominates, 2 innominate fragments, 2 pelvis, 9 femurs, 5 femur fragments, 2 tibias, 1 tibia fragment.
- Zone 4: 7 skulls, 18 skull fragments, 4 incisors, 25 mandibles, 20 mandible fragments, 9 scapulae, 14 humeri, 5 humerus fragments, 7 ulnas, 2 ulna fragments, 5 radil, 5 innominate fragments, 1 calcaneus, 9 femurs, 5 femur fragments, 1 pelvis, 11 tibias, 2 tibia fragments.

Zone 25: 2 skulls, 7 skull fragment, 4 mandibles, 1 mandible fragment, 1 humerus, 1 humerus fragment, 1 radius, 1 tibia.

Family Heteromyidae

Ferognathus parvus

Zone 1: 1 mandible, 1 mandible fragment, 3 skuli fragments, 2 femurs.

Zone 2: 2 skulls, 5 skull fragments, 11 mandibles, 2 mandible fragments, 4 femurs, 1 pelvis, 5 innominates, 2 tibias, 1 tibia fragment, 1 scapula.

Zone 3: 1 mandible, 6 mandible fragments, 1 skull fragment, 1 femur.

Zone 4: 2 mandibles, 1 mandible fragment, 2 innominate fragments.

Family Cricetaidae

Zone 1: 2 mandible fragments, 1 skull fragment, 1 tibia.

Zone 2: 4 mandible fragments, 3 skull fragments, 1 femur, 1 scapula, 1 ulna fragment.

Zone 3: 4 mandible fragments, 2 skull fragments, 1 humerus fragment, 1 tibia fragment.

Zone 4: 1 skull fragment, 1 innominate, i uina fragment.

Zone 25: 1 mandible fragment.

Lagurus curtatus

Zone 1: 4 mandibles, 1 mandible fragment, 1 skull fragment.

Zone 2: 2 mandibles.

Zone 3: 2 mandibles.

Zone 4: 1 mandible fragment.

Zone 25: 1 mandible.

Microtus sp.

Zone 3: 1 mandible fragment.

Peromyscus maniculatus

Zone 3: 1 femur.

141

Zone 4: 1 mandible, 1 mandible fragment.

Neotoma cinerea

Zone 2: 1 molar.

Zone 3: 1 molar.

Family Mustelidae

Zone 25: 1 metapodiai.

Taxidea taxus

Zone 25: 1 humerus fragment.

Family Canidae

Canis sp.

Zone 25: 1 mandible with teeth, 1 molar.

Family Cervidae

Zone 25: 5 antler fragments.

Odocoileus sp.

Zone 1: 2 premolars.

Zone 2: 1 metatarsal fragment.

Zone 25: 1 molar.

Family Cervidae/Bovidae

Deer-Sized

Zone 1: 1 metapodial fragment.

Elk Size

Zone 25: 1 cervical vertebrae fragment.

ኤስ ግራት የኤስ የኤስ ምራት የኤስ የራሳት የአማሪ የትድብ እንዲ

Family Ranidae/Bufonidae

Zone 4: 1 radiouina

Family Chelydridge

22 Denomer

ዀቘዀጚቘጟኯፚጜፙዀኯኯጜጜዀዀዀዀዀዀዀ

Chrysemys picta

Zone 1: 1 shell fragment.

Zone 2: 2 shell fragments.

Zone 3: 18 shell fragments.

Zone 25: 1 shell fragment.

Family Salmonidae

Zone 1: 1 vertebra.

Zone 2: 1 vertebra, 1 vertebra fragment.

Zone 3: 4 vertebrae, 4 vertebra fragments.

Zone 4: 3 vertebrae.

12 A 2020 MOUNTAGE RECEIPTING

APPENDIX D:

いたが、中国などなな問題についたときならななななながです。東京などなどの意味がなくなる

DESCRIPTION OF CONTENTS OF UNCIRCULATED APPENDICES

Detailed data from two different analyses are available in the form of hard copies of computer files with accompanying coding keys.

<u>Functional analysis</u> data include provenience (site, analytic zone, excavation unit and level, and feature number and level (if applicable)); object master number; abbreviated functional object type; and coding that describes each tool on a given object. Data normally are displayed in alphanumeric order by site, analytic zone, functional object type, and master number. Different formats may be available upon request depending upon research focus.

<u>Faunal analysis</u> data include provenience (site, analytic zone, excavation unit and level, feature number, and level (if applicable)); taxonomy (family; genus, species); skeletal element; portion; side; sex; burning/butchering code; quantity; and age. Data normally are displayed in alphanumeric order by site, analytic zone, provenience, taxonomy, etc.

To obtain copies of the uncirculated appendices contact U.S. Army Corps of Engineers, Seattle District, Post Office Box C-3755, Seattle, Washington, 98124. Copies also are being sent to regional archives and libraries.